# WHITE ROAD AT PEAVY ROAD INTERSECTION IMPROVEMENTS

# CONTRACT DOCUMENTS AND SPECIFICATIONS

PREPARED FOR:



PREPARED BY:



511 Gloucester St. Brunswick, GA 31520 Phone: (912) 227-1969 www.gwesllc.com

PROJECT NO. 051.18.2.22

July 2024

#### WHITE ROAD AT PEAVY ROAD INTERSECTION IMPROVEMENTS

#### **JULY 2024**

#### PROJECT NO. 051.18.2.22

# **TABLE OF CONTENTS**

#### <u>DIVISION 00 – PROCUREMENT AND CONTRACTING SPECIFICATIONS</u>

000100	Invitation to Bid
000200	Instruction to Bidder
000400	Bid Form
000410	Bid Bond
000420	Bidder's Qualifications
000422	Corporate Certificate
000480	Non-Collusion Affidavit
000483	Contractor E-Verify
000484	Sub-Contractor E-Verify
000500	Contract Agreement
000550	Pre-Award Oath
000610	Performance Bond
000620	Payment Bond
000621	Notice of Award
000622	Notice to Proceed
000700	General Conditions
000800	<b>Supplementary Conditions</b>
	• •

# **DIVISION 01 – GENERAL SPECIFICATIONS**

011000	Summary of Work
012000	Price and Payment Procedures
013000	Administrative Requirements
013216	Construction Progress Schedule
013300	Submittal Procedures
014000	Quality Requirements
014529	Testing Laboratory Services
015000	Temporary Facilities and Controls
015633	Job Site Security
017000	<b>Execution and Closeout Requirements</b>
017123.13	Construction Layout

#### **DIVISION 31 – EARTHWORK**

311000	Site Clearing
312333	Trenching and Backfilling
312500	Erosion and Sedimentation Control

#### **DIVISION 32-EXTERIOR IMPROVEMENTS**

321123	Aggregate Base Courses
321216	Asphalt Paving
321313	Concrete Paving
321723	Pavement Marking

# <u>DIVISION 33 – UTILITIES</u>

330110	Disinfection of Water Utility Piping
330505	Hydrostatic Testing
330509	Thrust Restraint for Utility Piping
330561	Concrete Manholes
331413	Public Water Utility Distribution Piping
331419	Valves and Hydrants for Water Utility Services

# SECTION 000100 INVITATION TO BID

#### PART 1: GENERAL

#### 1.1 Competitive Cost Sealed Bids

Separate sealed Bids for construction of **White Road at Peavy Road Intersection Improvements for the City of Byron** will be received until <u>September 4, 2024 at 3:00 P.M. EST</u> in the City Municipal Building, 401 Main Street, Byron, GA 31008. Bids will then be publicly opened and read aloud at said office. Any bid received after <u>September 4, 2024</u> at 3:00 <u>P.M. EST</u> will not be considered by the Owner. Following the closing time for the receipt of bids no bid may be withdrawn for a period of sixty (60) days.

#### 1.2 Scope of Work

The work to be performed and provided under this Contract shall consist of furnishing all labor, materials, tools, equipment, safety equipment, permitting, erosion and sedimentation control measures, taxes, profit, overhead, and other costs and effort required to complete the following:

Construction of Roundabout at the Intersection of White Road and Peavy Road. The time allotted for substantial completion of the work is 210 consecutive calendar days from the Notice to Proceed, and the time allotted for final completion of work is **240 consecutive calendar days** from the Notice to Proceed.

The contractor will be required to coordinate all work with the City of Byron and other utility owners who have facilities within the project area. The project will be managed by GWES, LLC. located at 511 Gloucester Street, Brunswick, GA 31520, telephone number (912) 227-1969.

All Work shall be completed in accordance with the plans and specifications. The Work will be awarded in one (1) Contract.

#### 1.3 Plans, Specifications and Contract Documents

The Instructions to Bidders, Bid Forms, Contract Agreement, Drawings, Specifications and forms of Bid Bond, Performance Bond, Payment Bond and other Contract Documents may be obtained upon payment (non-refundable) of \$200 from:

GWES, LLC. 733 Carroll Street Perry, GA 31069 (706) 765-8767

051.18.2.22

Questions should be directed to: William Jones (706) 765-8767

William.jones@gwesllc.com

# 1.4 Contractor License-Not Required

#### 1.5 Bonds

Each Bid must be accompanied by a Bid Bond or certified cashier's check, in the amount of 10% of the base bid, prepared on the form of Bid Bond, section 00410, attached to the Contract Documents or a Surety Company's Standard Bid Bond, duly executed by the Bidder as principal and having as surety thereon a surety company licensed to do business in the State of Georgia and listed in the latest issue of U.S. Treasury Circular 570 with an "A" minimum rating of performance and a financial strength of at least five (5) times the contract price as listed in the most current publication of "Best's Key Rating Guide Property Liability". Each Bond shall be accompanied by a "Power of Attorney" empowering the attorney-in-fact to bind the surety and certified to include the date of the bond.

#### 1.6 Permits

- A. GDOT Encroachment and Access permit supplied by the Owner
- B. Contractor to furnish necessary permits for Construction of Work including the following:
  - 1. Notice of Intent (NOI) and Land Disturbance Permit

#### 1.7 Easements & Rights-of-Way

Owner has or will acquire all necessary easements and rights-of-way for permanent installations preceding the issuance of the Notice to Proceed.

#### 1.8 **Reservation of Rights**

The Owner reserves the right to reject any or all Bids, to waive informalities and to readvertise.

#### 1.9 **Contractor E-Verify**

Contractor shall supply E-Verify documentation with the bid to be deemed responsive.

#### **END OF SECTION 000100**

# SECTION 00200 INSTRUCTION TO BIDDERS

#### 1.01 Contract Documents

- A. The Contract Documents include the Contract Agreement, Invitation to Bid, Instructions to Bidders, Contractor's Bid (including all documentation accompanying the Bid and any post-Bid documentation required by the Owner prior to the Notice of Award), Bonds, all Special Conditions, General Conditions, Supplementary Conditions, Specifications, Drawings, Alternates, and addenda, together with written amendments, change orders, field orders and the Engineer's written interpretations and clarifications issued in accordance with the General Conditions on or after the date of the Contract Agreement.
- B. Shop drawing submittals reviewed in accordance with the General Conditions, geotechnical investigations and soils reports, and drawings of physical conditions in or relating to existing surface or subsurface structures at or contiguous to the site, are not Contract Documents.
- C. The Contract Documents shall define and describe the complete work to which they relate.

#### 1.02 Definitions

- A. Where the following words or the pronouns used in their stead occur herein, they shall have the following meaning:
  - 1. "Owner" shall mean the City of Byron, party of the first part to the Contract Agreement, or its authorized and legal representatives.
  - 2. "Engineer" shall mean CITY OF BYRON's staff engineer(s) or CITY OF BYRON's consultant engineers.
  - 3. "Contractor" shall mean the party of the second part to the Contract Agreement or the authorized and legal representative of such party.
  - 4. "Work" and "Project" shall mean the entire completed construction required to be furnished under the Contract Documents.
  - 5. "Contract Time" shall mean the number of consecutive calendar days as provided in the Contract Agreement for completion of the Project, to be computed from the date of the Notice to Proceed.
  - 6. "Liquidated Damages" shall mean the sum of \$500 per day which the Bidder agrees to pay for each consecutive calendar day beyond the Contract Time required to complete the Project. Liquidated Damages will end upon written notification from the Owner of final acceptance of the Project.
  - 7. "Products" shall mean materials or equipment permanently incorporated into the Project.

- 8. "Provide" shall mean to furnish and install.
- 9. "Balanced Bid" shall mean a Bid in which each of the unit prices and total amount bid for each of the listed items reasonably reflects the value of that item with regard to the entire job considering the prevailing cost of labor, material and equipment in the relevant market. A Bid is unbalanced when, in the opinion of the Owner, any unit prices or total amounts bid on any of the listed items do not reasonably reflect such values.
- 10. "Substantial completion of the work", solely for the purposes of Official Code of Georgia Annotated (O.C.G.A.) §13-10-81(c), shall be defined as occurring on the date of the written notification from the Engineer that the Project is ready for final inspection, as specified in Section 00700, 14.04.

#### 1.03 Preparation and Execution of Bid

- A. Each Bid must be prepared to represent that it is based solely upon the materials and equipment specified in the Contract Documents. Complete sets of the Bidding Documents are available from GWES, LLC 733 Carroll Street, Perry, GA 31069 upon payment of \$200 for each full sized set.
- B. Each Bid must be submitted on the Bid forms which are attached to the Contract Documents. All blank spaces for Bid prices, both words and figures, must be filled in, in ink. In case of discrepancy, the amount shown in words will govern. All required enclosed certifications must be fully completed and executed when submitted.
- C. Each Bid must be submitted in a sealed envelope, addressed to the Owner. Each sealed envelope containing a Bid must be plainly marked on the outside with the **Project Name and Bid Number.**
- D. The Bidder shall provide on the outside of the sealed envelope the following information, otherwise the Bid will not be opened and will be returned to the Bidder:

#### 1. Bidder's Name (Business)

- E. If forwarded by mail, the sealed envelope containing the Bid must be enclosed in another envelope addressed to the Owner at the address given in the Invitation to Bid.
- F. Any and all Bids not meeting the aforementioned criteria for Bid submittal, may be declared non-responsive, and subsequently returned to the Bidder.
- G. The Contractor, in signing a Bid on the whole or any portion of the Project, shall conform to the following requirements:

051.18.2.22

1. Bids which are not signed by individuals making them shall have attached thereto a power of attorney evidencing authority to sign the Bid in the name of the person for whom it is signed.

- 2. Bids which are signed for a partnership shall be signed by all of the partners or by an attorney-in-fact. If a Bid is signed by an attorney-in-fact, there should be attached to the Bid a power of attorney executed by the partners evidencing authority to sign the Bid.
- 3. Bids which are signed for a corporation shall have the correct corporate name thereof and the signature of the president or other authorized officer of the corporation manually written below the corporate name following the wording "By \_\_\_\_\_\_\_". Corporation seal shall also be affixed to the Bid.
- 4. Bids which are signed for a limited liability company shall be executed in the name of the firm by a member and accompanied by evidence of authority to sign. The state of formation of the firm and the official address of the firm shall be shown.
- 5. The Bid shall contain evidence of Bidder's authority and qualification to do business in the state of Georgia, or Bidder shall covenant in writing to obtain such authority and qualification prior to award of the Contract and attach such covenant to the Bid. Bidder's state contractor license number, if any, shall also be shown on the Bid Form.

#### 1.04 Method of Bidding

- A. The unit or lump sum price for each of the several items in the Bid of each Bidder shall include its pro rata share of overhead and profit so that the sum of the products, obtained by multiplying the quantity shown for each item by the unit price, represents the total Bid. Any Bid not conforming to this requirement may be rejected. Additionally, Unbalanced Bids will be subject to rejection. Conditional Bids will not be accepted. The special attention of all Bidders is called to this provision, for should conditions make it necessary to revise the quantities, no limit will be fixed for such increased or decreased quantities nor extra compensation allowed.
- B. The total of all estimated prices will be the sum of the products of the estimated quantity of each item and the corresponding unit price. The final quantities and Contract Price will be determined in accordance with paragraph 11.03 of the General Conditions.
- C. Discrepancies between the multiplication of units of Work and unit prices will be resolved in favor of the unit prices. Discrepancies between the indicated sum and any column of figures and the correct sum thereof will be resolved in the favor of the correct sum.

#### 1.05 Submittal of Bid

- A. With each copy of the Bidding Documents, a Bidder is furnished one copy of the Bid Form, and the Bid Bond Form. The Bid Form is to be completed and submitted with the Bid security and the documents listed in Article 6 of the Bid Form (00 04 00).
- B. A Bid shall be submitted no later than the date and time prescribed and at the place indicated in the advertisement or Invitation to Bid and shall be enclosed in an opaque sealed envelope plainly marked with the Project title, the name and address of Bidder and shall be accompanied by the Bid security and other required documents. If a Bid is sent by mail or other delivery system, the sealed envelope containing the Bid shall be enclosed in a separate envelope plainly

marked on the outside with "Bid Enclosed" and "Project Name". A mailed bid shall be addressed to City of Byron, 401 Main Street, Byron, GA 31008.

C. Bids will only be accepted from entities that have received bidding documents from the Engineer.

#### 1.06 Modification and Withdrawal of Bid

- A. A Bid may be modified or withdrawn without revealing the amount of the bid by submitting an appropriate document duly executed in the same manner that a Bid must be executed and delivered to the place where Bids are to be submitted prior to the date and time for the opening of Bids.
- B. In accordance with O.C.G.A 36-91-52, if within forty-eight (48) hours after Bids are opened any Bidder files a duly signed written notice with Owner and promptly thereafter demonstrates to the reasonable satisfaction of Owner that there was a material and substantial mistake in the preparation of its Bid, that Bidder may withdraw its Bid, and the Bid security will be returned.
  - 1. Such error in the calculation of the bid can be documented by clear and convincing written evidence;
  - 2. Such error can be clearly shown by objective evidence drawn from inspection of the original work papers, documents or materials used in the preparation of the bid sought to be withdrawn.
  - 3. The bidder serves written notice upon the Owner either prior to the award of the contract or not later than forty-eight (48) hours after the opening of bids, excluding Saturdays, Sundays and legal holidays.
  - 4. The bid was submitted in good faith and the mistake was due to a calculation or clerical error, an inadvertent omission or typographical error as opposed to an error in judgment; and
  - 5. The withdrawal of the bid will not result in undue prejudice to the Owner or other bidders by placing them in a materially worse position than they would have occupied if the bid had never been submitted.

Thereafter, if the Work is rebid, that Bidder will be disqualified from further bidding on the Work. Furthermore, no Bidder who is permitted to withdraw a Bid shall for compensation be allowed to supply any material or labor, or perform any subcontract or other work agreement for, the person or firm to whom the contract is awarded or otherwise benefit, directly or indirectly, from the performance of the project for which the withdrawn Bid was submitted.

#### 1.07 Addenda and Interpretations

A. No interpretation of the meaning of the Drawings, Specifications or other pre-bid documents will be made to any Bidder orally.

- B. Every request for such interpretation should be made in writing and addressed to Claire Swaim GWES, LLC, 733 Carroll Street, Perry, Ga. 31069 or email at claire.swaim@gwesllc.com, and to be given consideration must be received at least five (5) business days prior to the date fixed for opening Bids.
- C. Any and all such interpretations and any supplemental instructions will be in the form of written Addenda to the Contract Documents which, if issued, will be mailed, shipped or faxed to all prospective Bidders (at the respective addresses furnished) prior to the date fixed for the opening of Bids.
- D. Failure of Bidders to receive or acknowledge any Addendum shall not relieve them of any obligation under the Bid. All Addenda shall become part of the Contract Documents.

#### 1.08 Bid Security

- A. Each Bid must be accompanied by a Bid Bond, prepared on the form of Bid Bond included herein or a Surety Company's Standard Bid Bond, duly executed by the Bidder as principal and having as surety thereon a surety company authorized to do business in the State of Georgia, listed in the latest issue of U.S. Treasury Circular 570 and with an "A" minimum rating of performance and a financial strength of at least five times the contract price as listed in the most current publication of "Best's Key Ratings Guide Property Liability", in the amount of ten (10) percent of the maximum amount bid. Attorneys-in-fact who sign Bonds must file with each Bond a currently dated copy of their power of attorney.
- B. If for any reason whatsoever the successful Bidder withdraws from the competition after opening of the Bids, or if Bidder refuses to execute and deliver the Contract and Bonds required within fifteen (15) consecutive calendar days after receipt of notice of the acceptance of Bid, the Owner may proceed to enforce the provisions of the Bid Bond.

#### 1.09 Receipt and Opening of Bids

- A. The Owner may consider a minor irregularity any Bid not prepared and submitted in accordance with the provisions hereof and may waive any minor irregularities or reject any and all Bids. Any Bid may be withdrawn prior to the scheduled time for the opening of Bids or authorized postponement thereof. Any Bid received after the time and date specified shall not be opened.
- B. Bids will be opened at the time and place indicated in the Advertisement or Invitation to Bid and unless obviously non-responsive, read aloud publicly. An abstract of the amounts of the base bids and major alternates, if any, will be made available to Bidders after the opening of Bids.

#### 1.10 Subcontracts

A. The Bidder is specifically advised that any person, firm or other party to whom it is proposed to award a subcontract under this Contract must be acceptable to the Owner.

#### 1.11 Conditions of the Project

- A. Each Bidder must be informed fully of the conditions relating to the construction of the Project and the employment of labor thereon. Failure to do so will not relieve a successful Bidder of the obligation to furnish all material and labor necessary to carry out the provisions of the Contract. Insofar as possible, the Contractor, in carrying out the work, must employ such methods or means as will not cause any interruption of or interference with the work of any other Contractor.
- B. The Bidder is advised to examine the location of the Project and to be informed fully as to its conditions; the conformation of the ground; the character, quality and quantity of the products needed preliminary to and during the prosecution of the work; the general and local conditions and all other matters which can in any way affect the work to be done under the Contract. Failure to examine the site will not relieve the successful Bidder of an obligation to furnish all products and labor necessary to carry out the provisions of the Contract.
- C. The Bidder shall notify the Owner of the date and time Bidder proposes to examine the location of the Project. The Bidder shall confine examination to the specific areas designated for the proposed construction, including easements and public right-of-ways. If, due to some unforeseen reason, the Owner's proceedings for obtaining the proposed construction site (including easements), have not been completed, the Bidder may enter the site only with the express consent of the property owner. The Bidder is solely responsible for any damages caused by examination of the site.

#### 1.12 Notice of Special Conditions

If any special federal, state, county or city laws, municipal ordinances, and the rules and regulations of any authorities having jurisdiction over construction of the Project, enclosed, herein referred to, or applicable by law to the Project, conflict with requirements of the Contract Documents, then the most stringent requirement prevails.

#### 1.13 *Obligation of Bidder*

By submission of a Bid, each Bidder warrants that Bidder has inspected the site and has read and is thoroughly familiar with the Contract Documents (including all addenda). The failure or omission of any Bidder to examine any form, instrument or document shall in no way relieve any Bidder from any obligation in respect to the Bid.

# 1.14 Method of Award

- A. The Contract will be awarded to the responsive, responsible Bidder submitting the Bid which is in the best interest of the Owner as determined by the Owner.
- B. The Bidder to whom the award is made will be notified. The Owner reserves the right to reject any and all Bids and to waive any minor irregularities in Bids received whenever such rejection or waiver is in the Owner's interest.
- C. A responsive Bidder shall be one who submits a Bid in the proper form without qualification or

intent other than as called for in the Contract Documents, and who binds himself or herself on behalf of the Bid to the Owner with the proper Bid Bond completed and attached, and who properly completes all forms required to be completed and submitted at the time of the Bidding. The Bidder shall furnish all data required by these Contract Documents. Failure to do so may result in the Bid being declared non-responsive.

- D. A responsible Bidder shall be one who can fulfill the following requirements:
  - 1. The Bidder shall maintain a permanent place of business. This requirement applies to the Bidder where the Bidder is a division of a corporation, or where the Bidder is fifty (50) percent or more owned by a person, corporation or firm.
  - 2. The Bidder shall demonstrate adequate construction experience and sufficient equipment resources to properly perform the work under and in conformance with the Contract Documents. This evaluation will be based upon a list of completed or active projects and a list of construction equipment available to the Bidder to perform the work. The Owner may make such investigations as deemed necessary to determine the ability of the Bidder to perform the work, and the Bidder shall furnish to the Owner all such information and data for this purpose as the Owner may reasonably request. The Owner reserves the right to reject any Bid if the evidence submitted by, or investigation of, such Bidder fails to satisfy the Owner that such Bidder is properly qualified to carry out the obligations of the Contract and to complete the Project contemplated therein.
  - 3. The Bidder shall demonstrate financial resources of sufficient strength to meet the obligations incident to the performance of the work covered by these Contract Documents. The ability to obtain the required Performance and Payment Bonds will not alone demonstrate adequate financial capability.
- E. Acceptance of the Bidder's documentation and substantiation or Contract Award by the Owner does not relieve the Bidder of liability for non-performance as covered in the Contract Documents, nor will the Bidder be exempted from any other legal recourse the Owner may elect to pursue.

#### 1.15 Contract Security and Insurance

A. Article 5 of the General Conditions or as modified by the Supplementary Conditions, establishes the requirements for performance and payment bonds and insurance. When the successful Bidder delivers the executed agreement to Owner, it shall be accompanied by such bond documents and insurance certificates.

#### 1.16 Contractor and Subcontractor's Insurance

A. Contractor and all Subcontractor's shall not commence work under this contract until he has obtained all the insurance required by the Supplementary Conditions.

#### **END OF SECTION 000200**

# SECTION 000410 BID BOND

#### STATE OF GEORGIA

COUNTY OF PEACH
KNOW ALL MEN BY THESE PRESENTS, that we,
WHEREAS, the Principal has submitted to the City of Byron's White Road at Peavy Road Intersection Improvements project.
NOW THEREFORE, the conditions of this obligation are such that if the Bid be accepted, the Principal shall, within fifteen (15) days after receipt of conformed Contract Documents, execute a Contract in accordance with the Bid upon the terms, conditions and prices set forth therein, and in the form and manner required by the Contract Documents and execute sufficient and satisfactory separate Performance and Payment Bonds payable to the Owner, each in an amount of one hundred (100) percent of the total Contract Price, in form satisfactory to the Owner, then this obligation shall be void; otherwise, it shall be and remain in full force and effect in law; and the Surety shall, upon failure of the Principal to comply with any or all of the foregoing requirements within the time specified above, immediately pay to the aforesaid Owner, upon demand, the amount hereof in good and lawful money of the United States of America, not as a penalty, but as liquidated damages.
This bond is given pursuant to and in accordance with the provisions of O.C.G.A. Section 36-91-1 <i>et.seq.</i> and all the provisions of the law referring to this character of bond as set forth in said Sections or as may be hereinafter enacted and these are hereby made a part hereof to the same extent as if set out herein in full.

IN WITNESS WHEREOF, the said Principal has hereunder affixed its signature and seal, and said Surety has hereunto caused to be affixed its corporate signature and seal, by its duly authorized officers, on this \_\_\_\_\_

051.18.2.22

day of \_\_\_\_\_\_, 20\_\_\_\_.

CONTRACTOR - PRINCIPAL:	
By:(Signature on File)	
Title:	
Address:	-
Phone:	_
Attest:	_
Attest: (Signature on File)	(SEAL)
Title:	
Note: Attest for a corporation must be by the corporate secretary; for a p individual by a notary.	artnership by another partner; for an
SURETY:	
By:(Signature on File)	
(Signature on File)	
Title:	
Phone:	
Attest:	
(Signature on File)	(SEAL)
Title:	
Note: Surety companies executing bonds must appear on the Treast (Circular 570 as amended) and be authorized to transact business in the Resident agent in state in which Work is to be performed:	
Name:	-
Address:	
<u></u>	_
Phone:	

# END OF SECTION 00 04 10

# SECTION 00 04 00 BID FORM

TO:		City of Byron
FROM	1:	
FOR:		White Road at Peavy Road Intersection Improvements Project
DATE	ED:	
Articl	e 1 – Bi	dder's Acknowledgements
1.01	exami surrou	indersigned Bidder, in compliance with your Invitation to Bid for the construction of this Project having and the Contract Documents, the site of the proposed work, and being familiar with all of the conditions anding the construction of the proposed Project, including the availability of materials and labor, hereby ses to construct the Project in accordance with the Contract Documents.
1.02	Agree labor intend defini	sidder proposes and agrees, if this Bid is accepted, to contract with the City of Byron in the form of Contract ement specified, to furnish all necessary products, machinery, tools, apparatus, means of transportation and necessary to complete the construction of the Work in full and complete accordance with the reasonably led requirements of the Contract Documents to the full and entire satisfaction of the City of Byron with a te understanding that no money will be allowed for extra work except as set forth in the Contract ments.
1.03	fifteer bid bo liquid	ndersigned further agrees that in case of failure on his part to execute the said contract and the Bonds within (15) consecutive calendar days after written notice being given of the award of the contract, the check or and accompanying this bid, and the monies payable thereon shall be paid into the funds of the Owner as ated damages for such failure, otherwise, the check or bid bond accompanying this proposal shall be ed to the undersigned.
Articl	e 2 – Bi	dder's Representations
2.01	In sub	omitting this Bid, Bidder represents that:
	A.	Bidder has examined and carefully studied the Bidding Documents, other related data identified in the Bidding Documents, and the following Addenda, receipt of which is hereby acknowledged:
		Addendum No. Addendum Date
	В.	Bidder has visited and become familiar with the Site and is satisfied with the Site conditions that may affect cost, progress and performance of the Work.
	C.	Bidder is familiar with and satisfied as to all Laws and Regulations that may affect cost, progress, and performance of the Work.

D.

Bidder has carefully studied: (1) reports of explorations and tests of subsurface conditions at or

contiguous to the Site and all drawings of physical conditions relating to existing surface or subsurface structures at or contiguous to the Site that have been identified in Part 6 of the Supplemental Conditions and (2) reports and drawings of Hazardous Environmental Conditions, if any, at the Site that have been identified in Part 6 of the Supplemental Conditions.

- E. Bidder has considered the information known to Bidder; information commonly known to contractors doing business in the locality of the Site; information and observations obtained from visits to the Site; the Bidding Documents; and the Site-related reports and drawings identified in the Bidding Documents, with respect to the effect of such information, observations and documents on (1) the cost, progress and performance of the Work; (2) the means, methods, techniques, sequences and procedures of construction to be employed by Bidder, including applying the specific means, methods, techniques, sequences and procedures of construction expressly required by the Bidding Documents; and (3) Bidder's safety precautions and programs.
- F. Based on the information and observations referred to in Paragraph 2.01.E above, Bidder does not consider that further examinations, investigations, explorations, tests, studies or data are necessary for the determination of this Bid for performance of the Work at the price(s) bid and within the times required and in accordance with the other terms and conditions of the Bidding Documents.
- G. Bidder is aware of the general nature of work to be performed by Owner and others at the Site that relates to the Work as indicated in the Bidding Documents.
- H. Bidder has given written notification to Engineer of all conflicts, errors, ambiguities or discrepancies that Bidder has discovered in the Bidding Documents and Bidder finds Engineers written resolution thereof acceptable.
- I. The Bidding Documents are generally sufficient to indicate and convey understanding of all terms and conditions for the performance of the Work for which this Bid is submitted.
- J. Bidder will submit written evidence of its Authority to do business in the State of Georgia prior to the date of execution of the Agreement.

#### Article 3 – Bidder's Certification

#### 3.01 Bidder certifies that:

- A. This Bid is genuine and not made in the interest of or on behalf of any undisclosed individual or entity and is not submitted in conformity with any collusive agreement or rules of any group, association, organization, or corporation;
- B. Bidder has not directly or indirectly induced or solicited any other Bidder to submit a false or sham Bid;
- C. Bidder has not solicited or induced any individual or entity to refrain from bidding; and

- D. Bidder has not engaged in corrupt, fraudulent, collusive or coercive practices in competing for the Contract. For the purposes of this Paragraph 4.01.D:
  - 1. "corrupt practice" means the offering, giving, receiving or soliciting of anything of value likely to influence the action of a public official in the bidding process;

- 2. "fraudulent practice" means an intentional misrepresentation of facts made (a) to influence the bidding process to the detriment of the Owner, (b) to establish bid prices at artificial non-competitive levels or (c) to deprive Owner of the benefits of free and open competition;
- 3. "collusive practice" means a scheme or arrangement between two or more Bidders, with or without knowledge of Owner, with the purpose to establish bid prices at artificial, non-competitive levels; and
- 4. "coercive practice" means harming or threatening to harm, directly or indirectly, persons or their property to influence their participation in the bidding process or affect the execution of the Contract.

#### Article 4 - Basis of Bid

4.01 The Contractor shall review Section 012000 Price and Payment Procedures for methods and limits for payments to the Contractor for the following pay items.

Item No.	Description	Unit	Quantity	Unit Price	Total Price
1	Mobilization and Demobilization				
a.	Mobilization	1	LS		
b.	Traffic Control	1	LS		
2	Erosion and Sedimentation Control				
a.	Type "NS" Silt Fence (Sd1-NS)	2510	LF		
b.	Haybale Check Dam (Cd-Hb)	36	EA		
C.	Tree Protection (Tr)	21	EA		
d.	Concrete Washdown	1	EA		
e.	Outlet Protection (St)	38	SY		
f.	Slope Stabilization (Ss)	2600	SY		
g.	Temporary Grassing (Ds1)	2	AC		
h.	Temporary Mulching (Ds2)	2	AC		
i.	Permanent Grassing (Ds3)	2	AC		
j.	Construction Exit (Co)	5	EA		
3	Grading / Paving				
a.	Grading Complete	1	LS		
b.	Graded Aggregate Bases 10 inch incl. Matl.	4600	TN		
C.	Recycled 2 inch Asphalt Concrete 12.5 mm Superpave, GP 2 only, incl. Bitum, Matl, and H Lime (220 lbs / sy)	660	TN		
d.	Recycled 2 inch Asphalt Concrete 19 mm Superpave, GP 1 or 2, incl. Bitum, Matl, and H Lime (220 lbs / sy)	650	TN		
e.	Recycled 4 inch Asphalt Concrete 25 mm Superpave, GP 1 or 2, incl. Bitum, Matl, and H Lime (440 lbs / sy)	1300	TN		
f.	Rumble Strips 20 ft Long	12	EA		
g.	Concrete Integral Median, Type 7, 6 inch	600	SY		
<u> </u>	Concrete Truck Apron Stamped	350	SY		

i.	Concrete Flume	14	EA		
j.	Concrete Header Curb, Type 2, 6 inch	250	LF		
k.	Concrete Header Curb, Type 7, 6 inch	190	LF		
l.	Concrete Header Curb, Type 9, 4 inch	300	LF		
m.	Concrete Curb and Gutter, Type 2, 30 inch	1700	LF		
n.	Sawed Joints	100	LF		
4	Piping / Infrastructure				
a.	15 inch Storm Drain Pipe, H1-10	24	LF		
b.	18 inch Storm Drain Pipe, H1-10	40	LF		
C.	15 inch Flared End Section	2	EA		
d.	18 inch Flared End Section	2	EA		
e.	Controlled Low Strength Flowable Fill	30	CY		
f.	Highway Signs and Posts	1	LS		
g.	Pavement Striping and Markings	1	LS		
h.	Remove Sanitary Sewer Pipe - 8 inch	50	LF		
i.	Remove Sanitary Sewer Manhole	1	EA		
j.	Sanitary Sewer Manhole	1	EA		
k.	Adjust Water Valve Box to Grade	1	EA		
l.	Water Main Pipe - 6 inch	150	LF		
m.	Water Main 45 Deg. Bend - 6 inch	1	EA		
n.	Connect to Existing Water Main	2	EA		
0.	Water Main 22.5 Deg. Bend - 6 inch	1	EA		
p.	PVC Carrier Pipes -3 inch	200	LF		
q.	Water Valve - 6 inch	2	EA		
r.	Fire Hydrant - 6 inch	1	EA		
	Lighting and Electrical Instrumentation and				
S.	Controls Installation Complete	1	LS		
5	Allowances				
t.	Owners Allowance	1	LS	\$25,000.00	\$25,000.00
				Total Base Bid	\$0.00

\$		
Dollars & Cents		
Price in Words		

051.18.2.22

Total Base Bid (Total Items 1-5 in the amount of)

# **Article 5 – Time of Completion**

5.01 Bidder agrees that the Work will be substantially complete within Two Hundred Ten (210) consecutive calendar days and will be completed and ready for final payment in accordance with Paragraph 14.07 of the General Conditions within **Two Hundred Forty (240) consecutive calendar days** after the start day identified in the Notice to Proceed.

Bidder accepts the provision of the Agreement in regards to liquidated damages in the event the Bidder fails to complete the Work within the times specified in the Agreement.

#### Article 6 – Attachments to this Bid

- 6.01 Bidder shall submit the following documents with the Bid:
  - A. Required Bid security in the form of ten (10) percent of the Total Base Bid;
  - B. List of Proposed Subcontractors;
  - C. List of Proposed Suppliers;
  - D. List of Project References;
  - E. Evidence of authority to do business in the State of Georgia;
  - F. Utility Contractor License-Not Required for GC
  - G. Statement of Bidder's Qualifications with supporting data (Section 00420);
  - H. Corporate Certificate (Section 00422)
  - I. Non-collusion Affidavit of Prime Bidder (Section 00480);
  - J. Contractor Affidavit of Employment Eligibility (Section 00483);
  - K. Subcontractor Affidavit of Employment Eligibility (Section 00484); and
  - L. Pre-Award Oath (Section 00550)

# **Article 7 – Define Terms**

7.01 The terms used in this Bid with the initial letter capitalized shall be defined as stated in the Instructions to Bidders, General Conditions and Supplementary Conditions.

#### **Article 8 – Bid Communication**

8.01 All communications regarding this bid shall be addressed to:

William Jones

**JULY 2024** 

GWES, LLC. 733 Carroll Street Perry, Ga 31069 Phone: (706) 765-8767

william.jones@gwesllc.com

# **Article 9 – Bid Submittal**

This bid is submitted by:			
Bidder:			
By:			
Name:			
Title:			
Attest:			(SEAL)
Bidder's Business Addre	ss:		
Phone No.	Fax No		
E-mail		_	
Submitted on		_20	
Utility Contractor Licens	e No. of Contractor performing	g work:	

#### END OF SECTION 00 04 00

# SECTION 000420 STATEMENT OF BIDDER'S QUALIFICATIONS

To demonstrate Bidder's qualifications to perform the Work, each Bidder must be prepared to submit to the Owner the following:

2. Permanent main office address
3. When organized.
4. If a corporation, where incorporated.
5. How many years have you been engaged in the contracting business under your present firm or trade name?
6. Contract on hand: (Schedule these, showing amount of each contract and the appropriate anticipated dates of completion.)
7. General character of work performed by your company.
8. Have you ever failed to complete any work awarded to you? If so, where and why?
9. Have your ever defaulted on a contract? If so, where and why?
10. List the more important projects recently completed by your company, stating the approximate cost for each, and the month and year completed.
11. List your major equipment available for this contract.
12. Experience in construction work similar in importance to this project.
13. Background and experience of the principal members of your organization, including the officers.
14. Credit available: \$
15. Give Bank Reference
16. Will you, upon request, fill out a detailed financial statement and furnish any other information that may be required by the Owner.
17. The undersigned authorizes and requests any person, firm or corporation to furnish any information requested by the Owner in verification of the recitals comprising the Statement of Bidder's Qualifications.

1. Legal Name of Bidder.

All of the above questions must be answered and the date given must be clear and comprehensive. This

statement must be notarized. If necessary, questions may be answered on separate attached sheets.

(Name of Bidder)	
Ву:	
Title	
State of Georgia	
County of	
	, being duly sworn deposes and says that he
is of	and that the answers to Name of Organization)
	Name of Organization)
the foregoing questions and all statements thereis	n contained are true and correct.
Subscribed and sworn to before me this	day of, 20
	(Notary Public)
	My Commission expires:
	, 20

**END OF SECTION 000420** 

# SECTION 000422 CORPORATE CERTIFICATE

Ι,	, certify that I am the Secretary of the Corporation named as Contractor
in the foregoing Bid; that	, who signed said Bid on behalf of the Contractor
was then	of said Corporation; that said Bid was duly signed for and on behalf of said
Corporation by authority of its E	Board of Directors, and is within the scope of its corporate powers; that said
Corporation is organized under the	ne laws of the State of
This day of	20
(Corpo	orate Secretary)(Signature on File)
	(SEAL)

**END OF SECTION 000422** 

# SECTION 000480 NON-COLLUSION AFFIDAVIT OF PRIME BIDDER

STATE OF <u>GEORGIA</u> COUNTY OF <u>PEACH</u>	
, being first duly sworn	, deposes and says that:
He or she is of	, the Bidder that has submitted the
He or she is fully informed respecting the preparation circumstances respecting such Bid;	and contents of the attached Bid and of all pertinent
Such Bid is genuine and is not a collusive or sham Bid	;
in interest, including this Affiant, has in any way indirectly with any other Bidder, firm or person to so Contract for which the attached Bid has been submitted Contract, or has in any manner, directly or indirectly, conference with any other Bidder, firm or person to fix Bidder, or to fix any overhead, profit or cost element of to secure through any collusion, conspiracy, connivated City of Byron or any person interested in the proposed. The price or prices quoted in the attached Bid are to	Fair and proper and are not tainted by any collusion, part of the Bidder or any of its agents, representatives,
Signed: _	(Signature on File)
	(Signature on File)
Title:	
Subscribed and Sworn to before me this day of_	,
(Notary Public)	
My Commission Expires:	

# SECTION 000483 CONTRACTOR AFFIDAVIT OF EMPLOYMENT ELIGIBILITY

By executing this affidavit, the undersigned Contractor verifies its compliance with O.C.G.A. 13-10-91, stating affirmatively that the individual, firm, or corporation which is engaged in the physical performance of services on behalf of **City of Byron** has registered with, is authorized to use and uses the federal work authorization program commonly known as E-Verify, or any subsequent replacement program, in accordance with the applicable provisions and deadlines established in O.C.G.A. §13-10-91. Furthermore, the undersigned Contractor will continue to use the federal work authorization program throughout the contract period.

The undersigned further agrees that, should it employ or contract with any subcontractor(s) in connection with the physical performance of services pursuant to this contract with the **City of Byron**, Contractor will secure from such subcontractor(s) similar verification of compliance with O.C.G.A. 13-10-91 Contractor further agrees to maintain records of such compliance and provide a copy of each such verification to the **City of Byron** at the time the subcontractor(s) is retained to perform such service. Contractor hereby attests that its federal work authorization user identification number and date of authorization are as follows:

E-Verify Company ID Number			
Date of Authorization			
Name of Contractor			
WHITE ROAD AT PEAVY ROAD INTERSECT	ION IMPROVE	MENTS	
Name of Project			
I hereby declare under penalty of perjury that the f	foregoing is true a	and correct.	
Executed on the day of	_, in	_ (city),	_(state).
Signature of Authorized Officer or Agent	Printed Name ar	nd Title of Authorize	d Officer or Agent
SUBSCRIBED AND SWORN BEFORE ME ON	THIS THE	DAY OF	, 20
Notary Public			
My Commission Expires:			

# SECTION 00 04 84 SUB-CONTRACTOR AFFIDAVIT OF EMPLOYMENT ELIGIBILITY

•	ontractor verifies its compliance with O.C.G.A. §13-10-91,	
	orporation which is engage in the physical performance of	
sub-contractor has privity of contract) and	(name of subcontractor with whom such (name of contractor) on behalf of	
City of Byron has registered with, is authorized to use and uses the federal work authorization program		
	t replacement program, in accordance with the applicable	
	§ 13-10-91. Furthermore, the undersigned sub-contractor	
-	zation program throughout the contract period and the	
	physical performance of services in satisfaction of such	
_	an affidavit to the sub-contractor with the information	
-	ersigned sub-contractor shall submit, at the time of such	
* * * * * * * * * * * * * * * * * * * *	(name of sub-contractor with whom such sub-	
contractor has privity of contract) Additionally	the undersigned sub-contractor will forward notice of the	
	(name of sub-contractor with whom	
such subcontractor has privity of contract). Sub-c	contractor hereby attests that its federal work authorization	
user identification number and date of authorization		
assi identification number and date of damonization	in the tab follows.	
E-Verify Company ID Number		
Date of Authorization	Name of Sub-Contractor	
Name of Project	Name of Public Employer	
I hereby declare under penalty of perjury that the fo	oregoing is true and correct.	
Executed on, 20 in	(city), (state).	
	(State).	
Signature of Authorized Officer or Agent		
Signature of Fautionized Officer of Figure		
Subscribed and Sworn before me on this the	Day of , 20	
	_ ,	
Notary Public		
My Commission Expires:	<u> </u>	

# SECTION 000500 CONTRACT AGREEMENT

This Contract Agreement made and entered into on the			, by and
between the <b>City of Byron</b> , party of the first part (hereinafter, party of the second part (hereinafter called the	* * * * * * * * * * * * * * * * * * * *		
WITNESSETH:			
That the Contractor, for the consideration hereinafter fully set	out, hereby agrees with the Ov	vner as fo	llows:
That the Contractor will furnish all products, tools, considescription necessary to carry out and to complete in a good, of Byron's White Road at Peavy Road Intersection Imputs strict conformity with the Drawings and the Specifications, Contractor, the Invitation to Bid, Instructions to Bidde Performance and Payment Bonds and all Addenda hereto in parts of this Contract Agreement, as if fully contained herein.	firm, substantial workmanlike provements project, and will of together with the foregoing ers, General and Supplement incorporated (if applicable) whi	e manner t complete Bid made ntary Con	the City work in by the
That the Contractor shall commence the Work to be performed specified in a written Notice to Proceed and shall substant consecutive calendar days and shall fully complete all works. Time is of the essence and is an essential element of the Commencement of the Work within the time limit named before commencement of the Work or defaults in completion the Contractor shall be liable for such liquidated damage established as a penalty but are calculated and agreed upon into the uncertainty and impossibility of making a determination incurred by the Owner and the general public of Byron, Geo Contractor to complete the Work on time. Such liquidated data are cumulative and shall be in addition to every other remedy by statute, or under the Contract.	tially complete all work heretork hereunder within 240 consorts this Contract, and the Contractor of \$500 for each calendar herein. If the Contractor abandon of all the Work after commerces. These fixed liquidated in advance by the Owner and this ion as to the actual and consecuring as a result of the failure amages referred to herein are in	under with ecutive caper shall part day that dons the Cencement damages are Contract quential don the part tended to	hin 210 alendar  y to the there is Contract thereof, are not ctor due damages rt of the be and
The Owner hereby agrees to pay to the Contractor for the fasubject to additions and deductions as provided in the Specific States of America, the sum of	ications and Bid, in lawful mo  Dolla f the nature of the Work afore lifficulties encountered in the p f the Work, its suspension or di e thereof, as herein provided,	ney of the ars (\$	e United  Tom the n of the ance and

The Owner shall make monthly partial payments to the Contractor in accordance with the provisions of the Contract Documents. Retainage will be held at five (5) percent of the Work completed and stored materials The Contractor may request a reduction in the percentage of retainage. The City may or may not grant such request based upon the progression of work in accordance with the Contract Agreement. Regardless of the percentage, retainage shall be held until the Work is complete and the Contractor has fulfilled all

requirements of the Contract Agreement.

Final payment on account of this Contract Agreement shall be made within thirty (30) days after the completion by the Contractor of all work covered by this Contract Agreement and the acceptance of such work by the Owner, in accordance with the provisions of the Contract Documents.

It is further mutually agreed between the parties hereto that if, at any time after the execution of this Contract Agreement and the surety bonds hereto attached for its faithful performance, the Owner shall deem the surety or sureties upon such bond to be unsatisfactory, or if, for any reason, such bond ceases to be adequate to cover the performance of the Work, the Contractor shall, at no additional expense to Owner, within five (5) days after the receipt of notice from the Owner to do so, furnish an additional bond or bonds in such form and amount, and with such surety or sureties as shall be satisfactory to the Owner. In such event, no further payment to the Contractor shall be deemed to be due under this Contract Agreement until such new or additional security for the faithful performance of the Work shall be furnished in manner and form satisfactory to the Owner.

IN WITNESS WHEREOF, the parties hereto have executed this Contract Agreement under their respective seals on the day and date first above written in four counterparts each of which shall, without proof or accounting for the other counterparts, be deemed an original Contract.

# APPROVED AS TO FORM BEFORE EXECUTION Attorney for the Owner OWNER: City of Byron By:\_\_\_\_\_ Name:\_\_\_\_\_(Please Print) Title:\_\_\_\_\_ WITNESS: Name:\_\_\_\_\_(Please Print) (SEAL) Title:\_\_\_\_\_ CONTRACTOR: By:\_\_\_\_\_ Name:\_\_\_\_\_(Please Print) Title: ATTEST: Name: (Please Print) (SEAL) Title:\_\_\_\_

Note: If the Contractor is a corporation, the Contract Agreement shall be signed by the president or vice president, attested by the secretary and the corporate seal affixed. If the Contractor is a partnership, the Contract Agreement shall be signed in the partnership name by one of the partners, with indication that he or she is a general partner.

# SECTION 00 05 50 PRE-AWARD OATH

We have not directly or indirectly violated law and we have not:

- Prevented or attempted to prevent competition in such bidding or proposals by any means whatever,
- Prevented or endeavored to prevent anyone from making a bid or proposal thereof by any means whatever, nor
- Caused or induced another to withdraw a bid or proposal for the work.

We affirm that no other officers, agents or other persons acted for the Contractor in the bidding for and procurement of this Contract.

Signature	Printed Name	Title	Date
Subscribed and Sworn be	fore me on Day of	, 20	
Notary Public			
My Commission Expires:	, 20		

# SECTION 000610 PERFORMANCE BOND

STATE OF GEORGIA	BOND NO
COUNTY OF PEACH	
(hereinafter known as Contractor), and we,acknowledge ourselves indebted and firmly bound a those entitled thereto, in the sum of	e,, as Principal,, as Surety, do hereby and held unto the City of Byron for use and benefit of
BUT THE CONDITION OF THE FOREGOING OB	LIGATION OR BOND IS THIS:
Intersection Improvements" project as more fully date of	the City Of Byron's "White Road at Peavy Road" appears in a written Contract Agreement bearing the 20, a copy of which Contract Agreement is by ally and faithfully perform all the undertakings and inbefore referred to and shall fully indemnify and save whatsoever which it may suffer by reason of any failure or reimburse and repay the said Owner any and all outlay such default, and shall correct all defects in products and completion of all Work, then this obligation shall be null
addition to the terms of the said Contract Agreeme Specifications accompanying the same shall in any w	agreed that no change, extension of time, alteration or nt, or in the Work to be performed thereunder, or the ray affect the obligations under this Contract Agreement damage, extension of time, alteration or addition to the the Contract Documents.
	the provisions of O.C.G.A. Section 36-91-1 <i>et.seq</i> . and ter of Bond as set forth in said Sections or as may be hereof to the same extent as if set out herein in full.

IN WITNESS WHEREOF, the said Contra has hereunto caused to be affixed its corp day of, 20	orate signature and seal, by its	duly authorized officers, on this
CONTRACTOR - PRINCIPAL:		
By:		
Name:	(Please Print)	
Title:		
ATTEST:		-
Name:	(Please Print)	- (77.11)
Title:		(SEAL)
Note: Attestation for a corporation must be for an individual by a notary.	e by the corporate secretary; for	a partnership by another partner;
SURETY:		_
Ву:		_
Name:	(Please Print)	
Title:		
ATTEST:		
Name:	(Please Print)	- (SEAL)
Title:		_

Note: Surety companies executing Bonds must appear on the Treasury Department's most current list (Circular 570 as amended) and be authorized to transact business in the state of Georgia.

#### **END OF SECTION 000610**

# SECTION 000620 PAYMENT BOND

STATE OF	GEORGIA	BOND NO.	
COUNTY (	OF PEACH		
(hereinafter	known as Contractor), and we,	ereinafter called the Owner), in the penal su  Dollars (\$ ) lawful	, as Surety, um of
		h sum will and truly to be made, we bind as, jointly and severally, firmly by these pro	ourselves, our
Improveme	, 20 for the CITY C	certain Contract Agreement with said of BYRON's "White Road at Peavy Roa Contract), which Contract Agreement and reof as fully as if set out herein.	d Intersection
whom any p Contractor a labor, produ Agreement, payment of	portion of the Work provided for in said and of such subcontractors shall promp acts, services, or supplies for or in the or in any amendment or extension o	on is such, that if said Contractor and all such Contract Agreement is sublet and all assetly make payments to all persons supply prosecution of the Work provided for in for addition to said Contract Agreement y the claimant in suits on this Bond, the full force and effect.	signees of said ving them with such Contract nt, and for the
HOWEVER	R, this Bond is subject to the following co	onditions and limitations:	
(a)	prosecution of the Work provided for action against the Contractor and Suret a proceeding, instituted in the coun Agreement is to be performed or in a Such right of action shall be asserted claimants for its use and benefit against	has furnished labor, products, or supplied in said Contract Agreement shall have a sty on this Bond, which right of action shall try in which the Work provided for in any county in which Contractor or Surety in proceedings instituted in the name of the said Contractor and Surety or either party of said Contract Agreement) in which act ment rendered thereon.	direct right of l be asserted in said Contract does business. the claimant or y (but not later
(b)		accept service of process or other pleading and hereby consent that such service sh	
(c)		or a greater sum than the penalty of this B on that is instituted later than one (1) year	

	s may be hereinafter enacted, and these are out herein in full.	are hereby made a part hereof to the
has hereunto caused to be affiz	said Contractor has hereunder affixed it sed its corporate signature and seal, by	its duly authorized officers, on this
CONTRACTOR - PRINCIPA	L:	<u> </u>
	By:	<u> </u>
	Name:(Please Print)	
	Title:	<u> </u>
ATTE	ST:	_
	Name:(Please Print)	
	Title:	(SEAL)
Note: Attestation for a corpora for an individual by a notary.	tion must be by the corporate secretary;	for a partnership by another partner;
SURETY:		<u></u>
	Ву:	_
	Name:(Please Print)	_
	Title:	_
WITNESS	S:	_
	Name:(Please Print)	
	Title:	(SEAL)

This Bond is given pursuant to and in accordance with provisions of O.C.G.A. Section 36-91-1 *et.seq.* hereinafter, and all the provisions of law referring to this character of Bond as set forth in

**END OF SECTION 000620** 

Note: Surety companies executing Bonds must appear on the Treasury Department's most current list (Circular 570 as amended) and be authorized to transact business in the state where the Project is located.

(d)

# **SECTION 000621** NOTICE OF AWARD

	Date:	 
Project: White Road at Peavy Road Intersection Improvements		
Bidder:		
Bidder's Address:		
This document serves as notification that your Bid datedabove has been considered. You are the Successful Bidder and are award		
Included with this Notice of Award are:		

Four (4) copies of the proposed Contract Documents, excluding Drawings which will be delivered separately.

The following conditions must be completed within fifteen (15) days of the date you receive this Notice of Award:

- 1. Deliver four (4) fully executed counterparts of the Contract Documents to the Owner.
- 2. Deliver the Contract Performance and Payment Bonds as specified in the Instructions to Bidders (Section 000200) and General Conditions (Paragraph 5.01) with the executed Contract Documents.
- Deliver the Contract Certificate of Insurance as specified in the General Conditions (Paragraph 3. 5.03) and Supplementary Conditions (Part 8) with the executed Contract Documents.

Failure to fulfill these requirements within the identified time will entitle Owner to consider you in default, annul this Notice of Award and declare your Bid security forfeited.

Owner will return to you one (1) fully executed counterpart of the Contract Documents within ten (10) days after your compliance with the above conditions.

		City of Byron
		Authorized Signature
		Title
Acceptance of Notice		
Receipt of the above Notice of A	Award is hereby ackn	owledged by:
Contractor	this day	y of, 20
Ву:		
By:		

**END OF SECTION 000621** 

# SECTION 00 06 22 NOTICE TO PROCEED

	Date:
Project: White Road at Peavy Road Intersection In	provements
Bidder:	
Bidder's Address:	
This document serves to notify you that the Contract on You are to beg Documents on or before that date. In accordance vector of the completion and readiness for final payment is	in performing your requirements under the Contract with the Contract Agreement, the date of Substantial
Preceding the start of any Work at the Site, Paragraph Owner must each deliver to the other certificates of maintain in accordance with the Contract Documents.	
City of Byron	
Owner	
Authorized Signature	
Title	
Date	
Accepted By:	
Contractor	
Authorized Signature	
Title	
Date	

END OF SECTION 00 06 22

# **SECTION 00700 - GENERAL CONDITIONS**

# TABLE OF CONTENTS

Antio		Page No.
	le 1 – Definitions and Terminology	-
	Defined Terms	5
1.02	Terminology	9
Artic	le 2 – Preliminary Matters	
2.01	Delivery of Bonds and Evidence of Insurance	11
	Copies of Documents	11
	Commencement of Contract Times; Notice to Proceed	11
2.04	Starting the Work	11
2.05	Before Starting Construction	11
2.06	Preconstruction Conference; Designation of Authorized Representatives	11
2.07	Initial Acceptance of Schedules	12
Artic	le 3 – Contract Documents: Intent, Amending, Reuse	
3.01	Intent	12
3.02	Reference Standards	12
3.03	Reporting and Resolving Discrepancies	13
	Amending and Supplementing Contract Documents	13
3.05	Reuse of Documents	14
3.06	Electronic Data	14
Artic	le 4 – Availability of Lands; Subsurface and Physical Conditions; Hazardo	us Environmental
	le 4 – Availability of Lands; Subsurface and Physical Conditions; Hazardo litions; Reference Points	us Environmental
Cond		us Environmental
<b>Cond</b> 4.01	litions; Reference Points	_
Cond 4.01 4.02	litions; Reference Points Availability of Lands	15
4.01 4.02 4.03	Litions; Reference Points  Availability of Lands  Subsurface and Physical Conditions	15 15
Cond 4.01 4.02 4.03 4.04	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions	15 15 15
Cond 4.01 4.02 4.03 4.04 4.05	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions Underground Facilities	15 15 15 17
Cond 4.01 4.02 4.03 4.04 4.05 4.06	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions Underground Facilities Reference Points Hazardous Environmental Condition at Site	15 15 15 17 17
Cond 4.01 4.02 4.03 4.04 4.05 4.06	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions Underground Facilities Reference Points Hazardous Environmental Condition at Site	15 15 15 17 17
Cond 4.01 4.02 4.03 4.04 4.05 4.06 Artic 5.01	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions Underground Facilities Reference Points Hazardous Environmental Condition at Site	15 15 15 17 17 17
Cond 4.01 4.02 4.03 4.04 4.05 4.06 Artic 5.01 5.02	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions Underground Facilities Reference Points Hazardous Environmental Condition at Site  le 5 – Bonds and Insurance Performance, Payment, and Other Bonds	15 15 15 17 17 18
Cond 4.01 4.02 4.03 4.04 4.05 4.06 Artic 5.01 5.02	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions Underground Facilities Reference Points Hazardous Environmental Condition at Site  le 5 – Bonds and Insurance Performance, Payment, and Other Bonds Licensed Sureties and Insurers	15 15 15 17 17 18
Cond 4.01 4.02 4.03 4.04 4.05 4.06 Artic 5.01 5.02 5.03	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions Underground Facilities Reference Points Hazardous Environmental Condition at Site  le 5 – Bonds and Insurance Performance, Payment, and Other Bonds Licensed Sureties and Insurers Certificates of Insurance	15 15 15 17 17 18 19 19
Cond 4.01 4.02 4.03 4.04 4.05 4.06 Artic 5.01 5.02 5.03 5.04	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions Underground Facilities Reference Points Hazardous Environmental Condition at Site  le 5 – Bonds and Insurance Performance, Payment, and Other Bonds Licensed Sureties and Insurers Certificates of Insurance Contractor's Insurance	15 15 15 17 17 18 19 19 20 20
Cond 4.01 4.02 4.03 4.04 4.05 4.06 Artic 5.01 5.02 5.03 5.04 5.05	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions Underground Facilities Reference Points Hazardous Environmental Condition at Site  le 5 - Bonds and Insurance Performance, Payment, and Other Bonds Licensed Sureties and Insurers Certificates of Insurance Contractor's Insurance Owner's Liability Insurance	15 15 15 17 17 18 19 19 20 20 21
Cond 4.01 4.02 4.03 4.04 4.05 4.06 Artic 5.01 5.02 5.03 5.04 5.05 5.06	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions Underground Facilities Reference Points Hazardous Environmental Condition at Site  le 5 - Bonds and Insurance Performance, Payment, and Other Bonds Licensed Sureties and Insurers Certificates of Insurance Contractor's Insurance Owner's Liability Insurance Property Insurance	15 15 15 17 17 18 19 19 20 20 21 21
Cond 4.01 4.02 4.03 4.04 4.05 4.06 Artic 5.01 5.02 5.03 5.04 5.05 5.06 5.07	Availability of Lands Subsurface and Physical Conditions Differing Subsurface or Physical Conditions Underground Facilities Reference Points Hazardous Environmental Condition at Site  le 5 - Bonds and Insurance Performance, Payment, and Other Bonds Licensed Sureties and Insurers Certificates of Insurance Contractor's Insurance Owner's Liability Insurance Property Insurance Waiver of Rights	15 15 15 17 17 18 19 19 20 20 21 21 21

Artic	ele 6 – Contractor's Responsibilities		
6.01	Supervision and Superintendence		24
6.02	Labor; Working Hours		24
6.03	Services, Materials, and Equipment		25
6.04	Progress Schedule		25
6.05	Substitutes and "Or-Equals"		25
6.06	Concerning Subcontractors, Suppliers, and Others		28
6.07	Patent Fees and Royalties		29
6.08	Permits		29
6.09	Laws and Regulations		29
6.10	Taxes		30
6.11	Use of Site and Other Areas		30
6.12	Record Documents		30
6.13	Safety and Protection		30
6.14	Safety Representative		31
	Hazard Communication Programs		32
	Emergencies		32
6.17	Shop Drawings and Samples		32
	Continuing the Work		34
	Contractor's General Warranty and Guarantee	34	
	Indemnification		34
	Delegation of Professional Design Services		35
<u>Artic</u>	ele 7 – Other Work at the Site		
7.01	Related Work at Site		36
7.02	Coordination	36	
7.03	Legal Relationships	37	
Artic	ele 8 – Owner's Responsibilities		
	Communications to Contractor		37
	Furnish Data		37
	Pay When Due		37
	Lands and Easements; Reports and Tests		37
	Insurance		37
	Change Orders		38
8.07	Inspections, Tests, and Approvals		38
8.08	Limitations on Owner's Responsibilities		38
8.09	Undisclosed Hazardous Environmental Condition		38
	Evidence of Financial Arrangements		38
8.11	Compliance with Safety Program		38
0.11	Compilance with safety 1 rogram		50
<u>Artic</u>	ele 9 – Engineer's Status During Construction		
9.01	Owner's Representative		38
9.02	Visits to Site		38
9.03	Project Representative		39
9.04	Authorized Variations in Work		39

CITY OF BYRON

WHITE RD AT PEAVY RD INTERSECTION IMP.

9.05	Rejecting Defective Work	39
9.06	Shop Drawings, Change Orders and Payments	39
9.07	Determinations for Unit Price Work	40
9.08	Decisions on Requirements of Contract Documents and Acceptability of Work	40
9.09	Limitations on Engineer's Authority and Responsibilities	40
	le 10 – Changes in the Work; Claims	
	Authorized Changes in the Work	41
	Unauthorized Changes in the Work	41
	Execution of Change Orders	41
	Notification to Surety	41
10.05	Claims	42
	le 11 – Cost of the Work; Allowances; Unit Price Work	40
	Cost of the Work	42
	Allowances	45
11.03	Unit Price Work	45
	le 12 - Change of Contract Price; Change of Contract Times	
	Change of Contract Price	46
	Change of Contract Time	47
12.03	Delays	47
	le 13 – Tests and Inspections; Correction, Removal, or Acceptance of Defective Work	
	Notice of Defects	48
	Access to Work	48
	Tests and Inspections	48
	Uncovering Work	49
	Owner May Stop the Work	49
	Correction or Removal of Defective Work	50
	Correction Period	50
	Acceptance of Defective Work	51
13.09	Owner May Correct Defective Work	51
	le 14 – Payments to Contractor and Completion	
	Schedule of Values	52
	Progress Payments	52
	Contractor's Warranty of Title	54
	Substantial Completion	54
	Partial Utilization	55
	Final Inspection	55
	Final Payment	55
	Final Completion Delayed	56
14.09	Waiver of Claims	57
Artic	le 15 – Suspension of Work and Termination	

15.01 Owner May Suspend Work	57
15.02 Owner May Terminate for Cause	57
15.03 Owner May Terminate for Convenience	58
15.04 Contractor May Stop Work or Terminate	59
Article 16 – Dispute Resolution 16.01 Methods and Procedures	59
Article 17 – Miscellaneous	60
17.01 Giving Notice	60
17.02 Computation of Times	60
17.03 Cumulative Remedies	60
17.04 Survival of Obligations	60
17.05 Controlling Law	60
17.06 Heading	61

# <u>ARTICLE 1 – DEFINITIONS AND TERMINOLOGY</u>

#### 1.01 Defined Terms

- A. The following terms shall have the specified meanings wherever they are used with initial capital letters in the Bidding Requirements or Contract Documents. Terms with initial capital letters in the Contract Documents will also include references to identified articles and paragraphs and the titles of other documents or forms.
  - 1. Addenda: Written or graphic instruments issued prior to the opening of Bids which clarify, correct, or change the Bidding Requirements or the proposed Contract Documents.
  - 2. Agency: The Federal or State agency named as such in the Agreement.
  - 3. *Agreement:* The written instrument which is evidence of the agreement between Owner and Contractor covering the Work.
  - 4. Application for Payment: The form acceptable to Engineer which is to be used by the Contractor during the course of the Work in requesting progress or final payments and which is to be accompanied by such supporting documentation as is required by the Contract Documents.
  - 5. Asbestos: Any material that contains more than one percent asbestos and is friable or is releasing asbestos fibers into the air above current action levels established by the United States Occupational Safety and Health Administration.
  - 6. Balanced Bid: A Bid in which each of the unit prices and total amount bid for each of the listed items reasonably reflects the value of that item with regard to the entire job considering the prevailing cost of labor, material and equipment in the relevant market. A Bid is unbalanced when, in the opinion of the Owner, any unit prices or total amounts bid on any of the listed items do not reasonably reflect such values.
  - 7. *Bid:* The offer or proposal of a Bidder submitted on the prescribed form setting forth the prices for the Work to be performed.
  - 8. *Bidder:* The individual or entity who submits a Bid directly to Owner.
  - 9. *Bidding Documents:* The Bidding Requirements and the proposed Contract Documents (including all Addenda).
  - 10. *Bidding Requirements:* The advertisement or invitation to bid, Instructions to Bidders, Bid security of acceptable form, if any, and the Bid Form with any supplements.

- 11. *Change Order:* A document recommended by Engineer which is signed by Contractor and Owner and authorized an addition, deletion, or revision in the Work or an adjustment in the Contract Price or the Contract Times, issued on or after the Effective Date of the Agreement.
- 12. *Claim:* A demand or assertion by Owner or Contractor seeking an adjustment of Contract Price or Contract Times, or both, or other relief with respect to the terms of the Contract. A demand for money or services by a third party is not a Claim.
- 13. *Contract:* The entire and integrated written agreement between the Owner and Contractor concerning the Work. The Contract supersedes prior negotiations, representations, or agreements, whether written or oral.
- 14. *Contract Documents:* Those items so designated in the Agreement. Only printed or hard copies of the items listed in the Agreement are Contract Documents. Approved Shop Drawings, other Contractor submittals, and the reports and drawings of subsurface and physical conditions are not Contract Documents.
- 15. *Contract Price:* The moneys payable by Owner to Contractor for completion of the Work in accordance with the Contract Documents as stated in the Agreement (subject to the provisions of Paragraph 11.03 in the case of Unit Price Work).
- 16. *Contract Times:* The number of consecutive calendar days as provided in the Contract Agreement for completion of the Project, to be computed from the date of the Notice to Proceed.
- 17. *Contractor:* The party of the second part to the Contract Agreement or the authorized and legal representative of such party.
- 18. Cost of the Work: Defined in Paragraph 11.01.
- 19. *Drawings:* That part of the Contract Documents prepared or approved by Owner which graphically shows the scope, extent, and character of the Work to be performed by Contractor. Shop Drawings and other Contractor submittals are not Drawings as so defined.
- 20. Effective Date of the Agreement: The date indicated in the Agreement on which it becomes effective, but if no such date is indicated, it means the date on which the Agreement is signed and delivered by the last of the two parties to sign and deliver.
- 21. Engineer: The City of Byron's staff engineer(s) or consultant engineers.
- 22. *Field Order:* A written order issued by engineer which requires minor changes in the Work but which does not involve a change in the Contract Price or the Contract Times.
- 23. General Requirements: Sections of Division 1 of the Specifications.

- 24. *Hazardous Environmental Condition:* The presence at the Site of Asbestos, PCBs, Petroleum, Hazardous Waste, or Radioactive Material in such quantities or circumstances that may present a substantial danger to persons or property exposed thereto.
- 25. *Hazardous Waste*: The term Hazardous Waste shall have the meaning provided in Section 1004 of the Solid Waste Disposal Act (42 USC Section 6903) as amended from time to time.
- 26. Laws and Regulations; Laws or Regulations: Any and all applicable laws, rules, regulations, ordinances, codes, and orders of any and all governmental bodies, agencies, authorities, and courts having jurisdiction.
- 27. *Liens:* Charges, security interests, or encumbrances upon Project funds, real property, or personal property.
- 28. Liquidated Damages: The sum of \$500 per day which the Bidder agrees to pay for each consecutive day, beyond the Contract Time, required to complete the Project. Liquidated Damages will end upon written notification from the Owner of final acceptance of the Project.
- 29. *Milestone:* A principal event specified in the Contract Documents relating to an intermediate completion date or time prior to Substantial Completion of all the Work.
- 30. *Notice of Award:* The written notice by Owner to the Successful Bidder stating that upon timely compliance by the Successful Bidder with the conditions precedent listed therein, Owner will sign and deliver the Agreement.
- 31. *Notice to Proceed:* A written notice given by Owner to Contractor fixing the date on which the Contract Times will commence to run and on which Contractor shall start to perform the Work under the Contract Documents.
- 32. *Owner:* The **City of Byron**, party of the first part to the Contract Agreement, or its authorized and legal representatives.
- 33. *PCBs*: Polychlorinated biphenyls.
- 34. *Petroleum:* Petroleum, including crude oil or any fraction thereof which is liquid at standard conditions of temperature and pressure (60 degrees Fahrenheit and 14.7 pounds per square inch absolute), such as oil, petroleum, fuel oil, oil sludge, oil refuse, gasoline, kerosene, and oil mixed with other non-Hazardous Waste and crude oils.
- 35. *Products*: Materials or equipment permanently incorporated into the Project.
- 36. *Progress Schedule:* A schedule, prepared and maintained by Contractor, describing the sequence and duration of the activities comprising the Contractor's plan to accomplish the Work within the Contract Times.

- 37. *Project:* The total construction of which the Work to be performed under the Contract Documents may be the whole, or a part.
- 38. *Project Manual:* The bound documentary information prepared for bidding and constructing the Work. A listing of the contents of the Project Manual, which may be bound in one or more volumes, is contained in the table(s) of contents.
- 39. *Provide:* To furnish and install.
- 40. *Radioactive Material:* Source, special nuclear, or byproduct material as defined by the Atomic Energy Act of 1954 (42 USC Section 2011 et seq.) as amended from time to time.
- 41. Related Entity: An officer, director, partner, employee, agent, consultant, or subcontractor.
- 42. *Resident Project Representative:* The authorized representative of Engineer who may be assigned to the Site or any part thereof.
- 43. *Samples:* Physical examples of materials, equipment, or workmanship that are representative of some portion of the Work and which establish the standards by which such portion of the Work will be judged.
- 44. *Schedule of Submittals:* A schedule, prepared and maintained by Contractor, of required submittals and the time requirements to support scheduled performance of related construction activities.
- 45. Schedule of Values: A schedule, prepared and maintained by Contractor, allocating portions of the Contract Price to various portions of the Work and used as the basis for reviewing Contractor's Applications for Payment.
- 46. *Shop Drawings:* All drawings, diagrams, illustrations, schedules, and other data or information which are specifically prepared or assembled by or for Contractor and submitted by Contractor to illustrate some portion of the Work.
- 47. *Site:* Lands or areas indicated in the Contract Documents as being furnished by Owner upon which the Work is to be performed, including rights-of-way and easements for access thereto, and such other lands furnished by Owner which are designated for the use of Contractor.
- 48. *Specifications:* That part of the Contract Documents consisting of written requirements for materials, equipment, systems, standards and workmanship as applied to the Work, and certain administrative requirements and procedural matters applicable thereto.
- 49. *Subcontractor:* An individual or entity having a direct contract with Contractor or with any other Subcontractor for the performance of a part of the Work at the Site.

- 50. Substantial Completion: The time at which the Work (or a specified part thereof) has progressed to the point where, in the opinion of Engineer, the Work (or a specified part thereof) is sufficiently complete, in accordance with the Contract Documents, so that the Work (or a specified part thereof) is ready for final inspection and can be utilized for the purposes for which it is intended. The terms "substantially complete" and "substantially completed" as applied to all or part of the Work refer to Substantial Completion.
- 51. Successful Bidder: The Bidder submitting a responsive Bid to whom Owner makes an award.
- 52. *Supplementary Conditions:* That part of the Contract Documents which amends or supplements the General Conditions.
- 53. *Supplier:* A manufacturer, fabricator, supplier, distributor, materialman, or vendor having a direct contract with Contractor or with any Subcontractor to furnish materials or equipment to be incorporated in the Work by Contractor or Subcontractor.
- 54. *Underground Facilities:* All underground pipelines, conduits, ducts, cables, wires, manholes, vaults, tanks, tunnels, or other such facilities or attachments, and any encasements containing such facilities, including those that convey electricity, gases, steam, liquid petroleum products, telephone or other communications, cable television, water, wastewater, storm water, other liquids or chemicals, or traffic or other control systems.
- 55. Unit Price Work: Work to be paid for on the basis of unit prices.
- 56. Work: The entire construction or the various separately identifiable parts thereof required to be provided under the Contract Documents. Work includes and is the result of performing or providing all labor, services, and documentation necessary to produce such construction, and furnishing, installing, and incorporating all materials and equipment into such construction, all as required by the Contract Documents.
- 57. Work Change Directive: A written statement to Contractor issued on or after the Effective Date of the Agreement and signed by Owner and recommended by Engineer ordering an addition, deletion, or revision in the Work, or responding to differing or unforeseen subsurface or physical conditions under which the Work is to be performed or to emergencies. A Work Change Directive will not change the Contract Price or the Contract Times but is evidence that the parties expect that the change ordered or documented by a Work Change Directive will be incorporated in a subsequently issued Change Order following negotiations by the parties as to its effect, if any, on the Contract Price or Contract Times.

# 1.02 Terminology

A. Intent of Certain Terms or Adjectives:

1. The terms "as allowed," "as approved," "as ordered," "as directed," or other terms of like effect included in the Contract Documents are to authorize an implementation of professional judgment by the Engineer. The adjectives "reasonable", "suitable". "acceptable", "proper", "satisfactory" or other adjectives of comparable effect are utilized to describe an action or determination of Engineer in relation to the Work. intention of Engineer's exercise of professional judgment, action or determination will be exclusively for the evaluation of the Work for compliance with the information in the Contract Documents and with the design concept of the Project as a whole as specified in the Contract Documents. The use of any such term is not intended to be effective appoint Engineer any duty or authority to supervise or direct the execution of the Work or to assume any responsibility contrary to the provisions of Paragraph 9.09 any other provision within the Contract Documents.

# B. Day:

1. The word "day" means a 24 hour period measured from midnight to the following midnight.

# C. Defective:

- 1. The word "defective", when describing the word "Work", refers to Work that is either unsatisfactory, faulty, or deficient in that it:
  - a. does not conform to the Contract Documents; or
  - b. does not meet the requirements of any applicable inspection, reference standard, test or approval referred to in the Contract Documents; or
  - c. Has been damaged prior to Engineer's recommendation of final payment (unless responsibility for the protection thereof has been assumed by Owner at Substantial Completion in accordance with Paragraph 14.04 or 14.05).

# D. Furnish, Install, Perform, Provide:

- 1. The word "furnish," when used in connection with services, materials, or equipment, shall mean to supply and deliver said services, materials, or equipment to the Site (or some other specified location) ready for use or installation and in usable or operable condition.
- 2. The word "install," when used in connection with services, materials, or equipment, shall mean to put into use or place in final position said services, materials, or equipment complete and ready for intended use.
- 3. The words "perform" or "provide," when used in connection with services, materials, or equipment, shall mean to furnish and install said services, materials, or equipment complete and ready for intended use.

- 4. When "furnish," "install," "perform," or "provide" is not used in connection with services, materials, or equipment in a context clearly requiring an obligation of Contractor, "provide" is implied.
- F. Unless stated otherwise in the Contract Documents, words or phrases that have a well-known technical or construction industry or trade meaning are used in the Contract Documents in accordance with such recognized meaning.

# **ARTICLE 2 – PRELIMINARY MATTERS**

- 2.01 Delivery of Bonds and Evidence of Insurance
  - A. Contractor shall deliver required bonds to Owner when Contractor delivers the executed counterparts of the Agreement to Owner.
  - B. *Evidence of Insurance:* Prior to starting any Work at Site, Contractor and Owner shall provide to the other certificates of insurance (along with any other proof of insurance that either of them may request), as noted in the Supplementary Conditions, which Contractor and Owner, in accordance with Article 5, are required to purchase and maintain.

# 2.02 Copies of Documents

- A. Owner shall provide Contractor up to six (6) printed or hard copies of the Drawings and Project Manual with additional copies being available at the cost of reproduction.
- 2.03 Commencement of Contract Times; Notice to Proceed
  - A. Contract Times will start on the day identified in the Notice to Proceed. A Notice to Proceed may be given at any time thirty days after the Effective Date of the Agreement. The start of the Contract Times will never run later than the sixtieth day after the day of Bid opening or the thirtieth day after the Effective Date of the Agreement.
- 2.04 Starting the Work
  - A. No work will be allowed prior to the date on which the Contract Times start. Contractor shall begin Work at the site on the date that the Contract Times start.
- 2.05 Before Starting Construction
  - A. *Preliminary Schedules:* Contractor shall submit to Engineer, for review, within ten days after the Effective Date of the Agreement preliminary:

- 1. Progress Schedule that indicates the times (dates or numbers of days) for beginning and finishing various phases of Work. This preliminary Progress Schedule shall also include Milestones specified in the Contract Documents;
- 2. Schedule of Submittals; and

3. Schedule of Values for the entire Work that includes quantities and prices of items which when summed together are equal to the Contract Price and subdivides the Work to allow for progress payments during the Work. Submitted prices will include an appropriate amount of overhead and profit applicable to each item.

# 2.06 Preconstruction Conference; Designation of Authorized Representatives

- A. Prior to beginning any Work at the Site, a conference attended by Owner, Contractor, Engineer and others as applicable will be held to provide a working understanding between the parties of the Work and to discuss the preliminary Schedules in Paragraph 2.05.A, process for Shop Drawings and other submittals, Applications for Payment and maintaining required records.
- B. Owner and Contractor shall each designate an individual(s) to act as authorized representative in regards to the services and responsibilities under the Contract at this conference. The individuals shall possess the ability to transmit instructions, receive information, provide decisions related to the Contract and act on behalf of each party.

# 2.07 Initial Acceptance of Schedules

- A. At least ten days prior to submission of the initial Application for Payment a meeting shall be held and attended by Engineer, Contractor and others to review for acceptability to Engineer, as provided below and in Section 01310, the schedules referenced in Paragraph 2.05.A. If necessary, Contractor will make corrections to the schedules and resubmit within ten days. Owner will not provide any progress payment until Contractor submits satisfactory schedules.
  - 1. Engineer will accept the Progress Schedule if an orderly progression of Work is provided to be completed within the Contract Times. Such acceptance will not relieve Contractor of their full responsibility nor make Engineer liable for Progress Schedule, progress of Work, sequencing or scheduling.
  - 2. Engineer will accept Contractor's Schedule of Submittals if a practical arrangement for review and processing of the required submittals is provided.
  - 3. Engineer will accept Contractor's Schedule of Values if, in form and substance, it provides a reasonable allocation of the Contract Price for the component parts of the Work.

# ARTICLE 3 - CONTRACT DOCUMENT: INTENT, AMENDING, REUSE

## 3.01 Intent

- A. The Contract Documents are complementary; what is required by one is as binding as if required by all.
- B. The Contract Documents are intended to describe a functionally complete project to be constructed in accordance with the Contract Documents. Any labor, documentation, services,

materials, or equipment that is mandatory to produce the intended results will be provided at no additional cost to Owner.

C. Engineer will issue clarifications and interpretations of the Contract Documents.

# 3.02 Reference Standards

# A. Standards, Specifications, Codes, Laws, and Regulations

- 1. Reference to standards, specifications, manuals, codes of any technical society, organization, association or Laws or Regulations, shall mean the standard, specification, manual, code, or Laws or Regulations that are in effect at the time of opening of Bids unless stated differently in the Contract Documents.
- 2. No provisions of any standard, specification, manual, code or instruction of a Supplier shall change the obligations of Owner, Contractor, subcontractor, consultants, agents or employees from those described in the Contract Documents. No provision or instruction shall assign any duty or authority to supervise or direct performance of the Work or undertake any responsibility inconsistent with Contract Documents to Owner, or their officers, directors, members, partners, employees, agents, consultants or subcontractors.

# 3.03 Reporting and Resolving Discrepancies

# A. Reporting Discrepancies:

- 1. Contractor's Review of Contract Documents Before Starting Work: Contractor is responsible to study and compare Contract Documents and confirm relevant figures and field measurements before beginning each part of the Work. Contractor is responsible for providing Engineer with written reports of any conflict, error or discrepancy which Contractor discovers. Engineer will provide Contractor with a written interpretation after which Contractor may proceed with any affected Work.
- 2. Contractor's Review of Contract Documents During Performance of Work: If any conflict, error or discrepancy within the Contract Documents, or between Contractor Documents and (i) any applicable Law or Regulation, (ii) any standard, specification, manual, or code, or (iii) any instruction by any Supplier, occurs during the performance of the Work, the Contractor is responsible for providing Engineer with written reports of the conflict. Contractor will not proceed with any affected Work until a revision or supplement of the Contract Documents is issued.
- 3. Unless Contractor had actual knowledge of any conflict, error or discrepancy in the Contract Documents, neither Owner nor Engineer will hold Contractor liable for failure to report.

#### B. Resolving Discrepancies:

1. Unless stated in the Contract Documents, the provisions of the Contract Documents

take precedence in the resolution of any conflict, error or discrepancy between the provisions of the Contract Documents and:

- a. the provisions of any standard, specification, manual, code, or the instruction of any Supplier; or
- b. the provisions of any Laws or Regulations applicable to the performance of the Work, unless violation of such Law or Regulation would result from the interpretation of the Contract Documents.

# 3.04 Amending and Supplementing Contract Documents

- A. Amendments to the Contract Documents may be made to accommodate additions, deletions and revisions in the Work or to modify the terms and conditions by either a Change Order of Work Change Directive.
- B. The requirements of the Contract Documents may be supplemented and minor changes in the Work may be approved by one of the following:
  - 1. Field Order;
  - 2. Engineer's approval of a Shop Drawing; or
  - 3. Engineer's written interpretation.

# 3.05 Reuse of Documents

- A. Neither Contractor, nor any Subcontractor or Supplier may:
  - 1. acquire any title to or ownership rights of any Drawings, Specifications or other documents prepared by or bearing the seal of Engineer or its consultants; or
  - 2. Reuse Drawings, Specifications, other documents or copies of the Project without Owner's and Engineer's specific written consent.
- B. The prohibitions in Paragraph 3.05 will last beyond final payment or termination of the Contract. These prohibitions will not prevent Contractor from possessing copies of the Contract Documents for record purposes.

#### 3.06 Electronic Data

A. The data furnished by Owner/Engineer to Contractor or Contractor to Owner/Engineer that may be relied upon is limited to printed copies, unless otherwise noted in the Supplementary Conditions. Electronic files are provided for the convenience of the recipient and any conclusion resulting from electronic files will be the sole risk of the user. Hard copies govern any difference between electronic fills and hard copies.

- B. Due to the deterioration and inadvertent modification of electronic media, it is the responsibility of the receiving party to perform acceptance test within sixty days, after which the receiving party will be deemed to have accepted the data. The transferring party will correct any errors detected within the sixty days.
- C. The transferring party is not responsible for compatibility, usability or readability of documents due to differing software applications, operating systems or computer hardware of the recipient.

# <u>ARTICLE 4 – AVAILABILITY OF LANDS; SUBSURFACE AND PHYSICAL CONDITIONS; HAZARDOUS ENVIRONMENTAL CONDITIONS; REFERENCE POINTS</u>

# 4.01 Availability of Lands

- A. The Site of the Project shall be furnished by Owner. Owner shall inform Contractor of any restrictions or encumbrances that are specific to the Site that Contractor must comply with during performance of the Work. It is the responsibility of the Owner to obtain and pay for, in a timely manner, any easements needed for permanent structures or permanent changes to existing facilities. If Owner and Contractor are unable to agree on entitlement to or on the amount of any adjustment in the Contract Price or Contract Times that results due to delay caused by Owner's furnishing of the Site the Contractor may make a claim as provided by Paragraph 10.05.
- B. Contractor may request a current statement of record legal title and legal description of the lands upon which the Work will be performed as necessary to file a mechanic's or construction lien against such lands in accordance with applicable Laws and Regulations.
- C. Contractor is responsible for providing all additional lands and access to lands that may be required for temporary construction facilities or for the storage of equipment and materials.

# 4.02 Subsurface and Physical Conditions

- A. *Reports and Drawings:* The Supplementary Conditions identify:
  - 1. reports of explorations and tests of subsurface conditions at or contiguous to the Site that are known by the Owner.
  - 2. drawings of physical conditions relating to existing surface or subsurface structures at or contiguous to the Site that are known by the Owner (except Underground Facilities).
- B. Limited Reliance by Contractor on Technical Data Authorized: "Technical data" contained in such reports and drawings are not Contract Documents but the Contractor may rely on them regardless. The Supplementary Conditions identify such "technical data". Except for reliance on "technical data", Contractor may not rely upon or make claim against Owner or Engineer or any of their officers, directors, members, partners, employees, agents, consultants or subcontractors with respect to:
  - 1. the extensiveness of such reports and drawings for Contractor's purposes, including, but

- not limited to, means, methods, techniques, sequences and procedures of construction employed by Contractor; or
- 2. any other information, data or interpretations of such reports or drawings; or
- 3. any interpretation made by the Contractor of any "technical data" or any other data, interpretations or information.

# 4.03 Differing Subsurface or Physical Conditions

- A. *Notice:* If Contractor believes that any subsurface or physical condition that is uncovered or revealed either:
  - 1. differs in nature from any "technical data" Contractor is entitled to rely on as provided in Paragraph 4.02; or
  - 2. is of such a nature to mandate a change of the Contract Documents; or
  - 3. differs from what is shown in the Contract Documents; or
  - 4. is unusual in nature and is materially different from ordinary conditions that are recognized as inherent to the work as provided by the Contract Documents; it is the responsibility of the Contractor to inform the Owner and Engineer in writing promptly after becoming aware of the condition and to halt any other disturbance of the subsurface or physical conditions (except in the case of an emergency as required in Paragraph 6.16.A). No further Work shall be performed at the site until Contractor receives a written order to do so.
- B. *Engineer's Review:* Engineer will promptly review the relevant condition after receipt of written notice as required by Paragraph 4.03.A and determine if there is a need for additional testing in the area. Engineer will inform Contractor in writing of the findings and conclusions.
- C. Possible Price and Times Adjustments:
  - 1. The Contract Price or Contract Times will be increased or decreased based on the findings of such differing subsurface or physical condition subject to the following:
    - a. such condition must meet at least one of the categories described in Paragraph 4.03.A; and
    - b. any adjustment in Contract Price for Work that is unit price based will be subject to the provisions of Paragraphs 9.07 and 11.03.
  - 2. No adjustment in the Contract Price or Contract Times will be made if:

051.18.2.22

a. Contractor had knowledge of such conditions at the time Contractor submitted a Bid.

- b. discovery of such condition could have been reasonably revealed through any examination, investigation, exploration, test or study of the Site as required by the Bidding Requirements or Contract Documents that should be conducted by Contractor before making final commitment to Owner; or
- c. written notice, as required by Paragraph 4.03.A, was not executed by Contractor.
- 3. If an agreement is not able to be made between Owner and Contractor for an adjustment in the Contract Price or Contract Time, a Claim may be made as provided in Paragraph 10.05. Contractor shall not hold Owner or Engineer liable for any claims, costs, losses or damages sustained by Contractor in connection with any other project or anticipated project.

# 4.04 Underground Facilities

- A. Shown or Indicated: The Underground Facilities at or contiguous to the Site that are shown in the Contract Documents is based on information and data furnished to Owner or Engineer by the owners of such Underground Facilities. Unless otherwise provided in the Supplementary Conditions:
  - 1. Owner and Engineer are not responsible for accuracy of any information or date provided by others; and
  - 2. Contractor has full responsibility of the following and the cost of all will be included in the Contract Price:
    - a. reviewing and verifying all provided information and data;
    - b. locating all Underground Facilities shown in the Contract Documents;
    - c. coordinating Work with owners of all Underground Facilities at the Site during construction; and
    - d. safety and protection of all Underground Facilities and fixing any damage that is the result of the Work.

#### B. Not Shown or Indicated:

- Contractor must contact owner of any Underground Facility that is uncovered at or contiguous to the Site which was not shown or not shown with reasonable accuracy in the Contract Documents. The Contractor must also provide Owner and Engineer with written notice. Engineer will promptly review the Underground Facility and determine if a change in the Contract Documents is required due to the existence of such Underground Facility. It will be the responsibility of the Contractor to protect such Underground Facility.
- 2. If a change in the Contract Documents is deemed necessary by the Engineer, a Work

Change Directive or a Change Order will be issued to document such consequences. As necessary, an adjustment shall be made in the Contract Price or Contract Time due to the hindrance caused by such Underground Facility. If an agreement on the adjustment of the Contract Price or Contract Time between Owner and Contractor is not reached, a Claim may be made, as provided by Paragraph 10.05.

# 4.05 Reference Points

A. Reference points necessary for the Contractor will be provided by Owner through engineering surveys. It is the responsibility of the Contractor to lay out the Work and protect and preserve established reference points. The Contractor shall not relocate or make changes to established reference points without written approval from Owner. Contractor shall inform Engineer of any reference points that are lost, destroyed or require relocation. Contractor will bear responsibility for accurate replacement or relocation of said reference points by professionally qualified personnel.

#### 4.06 Hazardous Environmental Condition at Site

- A. *Reports and Drawings:* Any Hazardous Environmental Conditions at the Site that are known by the Owner will be identified by reports and drawings in the Supplementary Conditions.
- B. Limited Reliance by Contractor on Technical Data Authorized: "Technical data" will be identified in the Supplementary Conditions. Contractor may rely on the "technical data" in such reports and drawing but reports and drawings are not Contract Documents. Except for reliance upon "technical data", Contractor may not make any claim against Owner or Engineer with respect to:
  - 1. The comprehensiveness of "technical data" for use by the Contractor for any aspects of but not limited to safety precautions, techniques, means, methods and procedures of construction used by Contractor; or
  - 2. data, interpretations, opinions and information contained in the "technical data"; or
  - 3. any interpretation or conclusion developed from the "technical data" by Contractor or any other data, interpretation, opinion or information.
- C. Any Hazardous Environmental Condition uncovered by Contractor at the Site that was not shown in the Drawings, Specifications or identified in the Contract Documents will not be the responsibility of the Contractor. Any Hazardous Environmental Condition created by materials brought to the Site by anyone the Contractor is responsible for, shall be the responsibility of the Contractor.
- D. If a Hazardous Environmental Condition is encountered or created by Contractor or anyone Contractor is responsible for, Contractor must immediately: (i) isolate or secure said condition; (ii) stop all Work in any area affected by or in connection with said condition, except in an emergency as required by Paragraph 6.16.A; and (iii) notify Owner and Engineer and later confirm notification in writing. Engineer will promptly determine if a qualified expert is needed

- to evaluate said condition. Owner will promptly take action to obtain required permits and will provide Contractor with a written notice as required in Paragraph 4.06.E.
- E. Contractor will not be required to resume Work in affected area until Owner has obtained required permits and delivered written notice to Contractor specifying: (i) the affected site has been rendered safe for Work to resume; or (ii) any precautions needed to safely resume Work. If an agreement between Owner and Contractor is not met concerning Contract Price or Contract Items as a result of Work stoppage or any special conditions required to resume Work, a Claim may be made as provided in Paragraph 10.05.
- F. If Contractor does not reasonably believe that it is safe to resume Work or does not agree to resume Work under special conditions required after written notice from Owner, the Owner may order affected area to be deleted from the Work. If an agreement between Owner and Contractor is not met concerning Contract Price or Contract Items as a result of deleting said portion of the Work, a Claim may be made as provided in Paragraph 10.05. Owner may use Owner's forces or others, in accordance with Article 7, to perform the deleted portion of the Work.
- G. Owner shall indemnify and hold harmless Contractor and anyone under the responsibility of the Contractor against all claims, costs, damages and losses due to the Hazardous Environmental Condition, provided that the Hazardous Environmental Condition: (i) was not indicated in any Contract Documents, Drawings or Specifications within the scope of the Work; and (ii) was not caused by Contractor or anyone under the responsibility of the Contractor. Nothing in this Paragraph obligates Owner to indemnify any individual or entity from consequences caused by that individual or entity's own negligence.
- H. Contractor shall indemnify and hold harmless Owner and engineers, officers, directors, members, partners, employees, agents, consultants, and subcontractors of the Owner against all claims, costs, damages and losses due to any Hazardous Environmental Condition created by Contractor or anyone under the responsibility of the Contractor. Nothing in this Paragraph obligates Contractor to indemnify any individual or entity from consequences caused by that individual or entity's own negligence.
- I. Any Hazardous Environmental Condition uncovered or revealed at the Site does not fall under the provisions of Paragraph 4.02, 4.03 and 4.04.

## ARTICLE 5 – BONDS AND INSURANCE

- 5.01 *Performance, Payment and Other Bonds* 
  - A. Performance and payment bonds shall be furnished by Contractor. Each bond shall be in an amount at least equal to the Contract Price. The bonds must remain in effect until the end of the correction period, as specified in Paragraph 13.07, or one year after the date that the final payment is due, whichever is later, unless there is a conflict with Laws or Regulations or the Contract Documents. Any other bonds required by the Contract Documents must be furnished by the Contractor.

- B. All bonds shall be executed by sureties named in the list of "Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds and as Acceptable Reinsuring Companies" as published in Circular 570 by the U.S. Department of the Treasury. Said bonds shall be in the form required by the Contract Documents unless it conflicts with any Laws or Regulations. All bonds must be signed by an agent or attorney-in-fact and in order to bind the surety must have a certified copy of that individual's authority. The effective date shall be the date the agent or attorney-in-fact signs each bond and shall be shown with the evidence of authority.
- C. Contractor shall notify Owner and Engineer if the surety on any bond has its right to do business in the state of Georgia terminated, becomes insolvent, is declared bankrupt or no longer meets the requirements of Paragraph 5.01.B. The Contractor shall provide another bond and surety within 20 days of notification of Owner and Engineer.

#### 5.02 Licensed Sureties and Insurers

A. All surety or insurance companies used by Contractor must be duly licensed or authorized to issue bonds or insurance policies for the limits and coverages required within the state of Georgia. Any additional requirements in Sections 00610 or 00620 or the Supplementary Conditions must also be met by any surety or insurance company used by Contractor.

# 5.03 Certificates of Insurance

- A. Certificates of insurance that Contractor is required to purchase and maintain, must be delivered to Owner by Contractor with copies to each additional insured and loss payee shown in the Supplementary Conditions.
- B. Contractor is obligated to maintain such insurance regardless of failure of Owner to demand evidence or certificates of Contractor's compliance with insurance requirements or failure of Owner to find any deficiency in the compliance from the documentation provided by Contractor.
- C. Owner does not claim that Contractor will be adequately protected by the insurance coverage and limits required by this Contract.
- D. Contractors required insurance and insurance limits will not be interpreted as a limitation on Contractor's liability under the indemnities granted to Owner by the Contract Documents.

# 5.04 Contractor's Insurance

- A. Insurance shall be purchased and maintained by Contractor that is appropriate for Work being performed and provides protection from any claims that may arise as a result of Contractor's performance of the Work or any other obligations under the Contract Documents, regardless if it is performed by Contractor or any Subcontractor or Supplier, or anyone directly or indirectly employed by Contractor to perform any of the Work, or anyone whose acts they may be liable for:
  - 1. disability benefits and other similar employee benefit acts claimed under workers'

compensation;

- 2. damages claimed due to bodily injury, occupational sickness or disease or death of Contractor's employees or any person other than Contractor's employees;
- 3. damages claimed due to bodily injury, sickness or disease or death of any person other than Contractor's employees;
- 4. claims for damages that are insured by personal injury liability coverage that are sustained by:
  - a. any person as a result of an offense that is related either directly or indirectly to the employment of said person by Contractor; or
  - b. any other person for any reason.
- 5. damages claimed, other than to the work itself, due to injury to or destruction of tangible property regardless of location, which also includes loss of use as a result of such; and
- 6. damages claimed due to bodily injury or death of any person or any property damage arising from the ownership, maintenance or use of any motor vehicle.
- B. The policies of insurance required by this Paragraph 5.04 shall:
  - 1. with respect to insurance required by Paragraphs 5.04.A.3 through 5.04.A.6, be written on an occurrence basis, include Owner, Engineer and any other individuals or entities identified in Supplementary Conditions as additional insureds, all shall be listed as additional insureds, and coverage should be included for the respective officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of all such additional insureds and the insurance provided to the additional insureds must provide primary coverage for all claims covered by means of that;
  - 2. include at minimum the specific coverages and be written to equal or exceed the limits of liability required by the Supplementary Conditions or required by Laws or Regulations, whichever is greater;
  - 3. contain contractual liability insurance to cover indemnity obligations of the Contractor as described in Paragraphs 6.11 and 6.20;
  - 4. contain a provision stating the coverage will not be canceled, renewal refused or materially changed until at minimum thirty days after written notice has been given to Owner, Contractor and to both Owner and Contractor's additionally insured that are identified in the Supplementary Conditions to whom a certificate of insurance has been issued;
  - 5. be in effect until final payment and any time after final payment when Contractor may be correcting, removing or replacing defective work in accordance with Paragraph 13.07; and

- 6. include completed operations coverage:
  - a. insurance will remain in effect for two years after final payment.
  - b. Owner and each other additional insured shall be furnished with evidence of continued insurance at final payment and one year after by Contractor.

#### 5.05 Owner's Liability Insurance

A. Owner may choose to purchase and maintain at Owner's expense liability insurance to protect Owner against claims that may arise because of operations under the Contract Documents.

# 5.06 Property Insurance

- A. Property insurance will be purchased and maintained by Owner for the Site in the amount of full replacement cost unless otherwise provided in the Supplementary Conditions. This insurance shall:
  - 1. provide coverage for Owner, Engineer, Contractor, Subcontractors and any other individual or entities identified by the Supplementary conditions and anyone whom is deemed to have an insurable interest and shall be listed as a loss payee;
  - 2. provide coverage for physical loss or damage to the Work, falsework, materials and equipment in transit and temporary buildings. At minimum it shall insure against: fire, theft, water damage, debris removal, collapse, earthquake, vandalism or malicious mischief, lightning, extended coverage, demolition occasioned by enforcement of Laws and Regulations and any other cause of loss required by Supplementary Conditions;
  - 3. provide coverage for expenses incurred due to repair or replacement of any insured property;
  - 4. provide coverage for stored materials and equipment at the Site or at another location agreed upon by Owner in writing, provided that said materials and equipment have been included in an Application for Payment;
  - 5. allow Owner to partially utilize the Work;
  - 6. include startup and testing; and
  - 7. provide continued coverage until final payment is made unless an agreement is made in writing between Owner, Engineer and Contractor and thirty days written notice given to each other loss payee that has been issued a certificate of insurance.
- B. As required by the Supplementary Conditions or Laws and Regulations, Owner shall purchase and maintain insurance to cover equipment breakdown or additional property insurance that includes the interests of Owner, Engineer, Contractor, Subcontractor and any other individual or

entities identified by the Supplementary Conditions.

- C. A provision or endorsement is required for all purchased and maintained insurance policies stating the coverage will not be canceled, materially changed or renewal refused until thirty days prior notice has been given to Owner, Engineer, Contractor and any other loss payee who was issued a certificate of insurance.
- D. It will not be the responsibility of the Owner to purchase and maintain any property insurance to protect the interests of Contractor, Subcontractors or others in the Work to the extent of any deductible amounts identified in the Supplementary Conditions. The Contractor, Subcontractor or others may purchase and maintain property insurance coverage at their own expense due to each bearing a risk of loss of an identified deductible amount.
- E. Contractor may request in writing any other special insurance to be included in the property insurance policies provided in this Paragraph 5.06. If possible, Owner shall include said insurance with the cost being charged to Contractor through a Change Order. Owner will inform Contractor whether or not such insurance was purchased by Owner prior to the beginning of Work at the Site.

# 5.07 Waiver of Rights

- A. The intent of Owner and Contractor is for all policies purchased in accordance with Paragraph 5.06 to protect Owner, Engineer, Contractor, Subcontractors and all other individuals or entities identified in the Supplementary Conditions as loss payees with said policies. It is also their intent to provided primary coverage for all losses and damages through said policies. Provisions shall be contained in all policies to provide that insurers will have no rights of recovery against any of the insureds or loss payees should any payment of any loss or damage be made. Owner and Contractor waive: (i) all rights against one another, their officers, directors, members, partners, employees, agents, consultants and subcontractors for any and all losses and damages due to any of the perils or causes of loss that are covered by said policies or any of the property insurance applicable to the Work; (ii) all rights against Subcontractors and any loss payees, as identified by the Supplementary Conditions, under said policies for losses and damages they cause. None of the above waivers shall extend to the rights that any party making such waiver may have to the proceeds of insurance held by Owner as trustee or otherwise payable under any policy so issued.
- B. Owner waives all rights against Contractor, Subcontractor and their officers, directors, members, partners, employees, agents, consultants and subcontractor for:
  - loss caused by business interruption, loss of use, or other loss extending beyond direct
    physical loss or damage to Owner's property or the Work as a result of fire or other peril
    whether or not insured by Owner; and
  - 2. loss or damage to the finished Project or part of the Project due to fire or other insured peril or any other cause of loss covered by any property insurance that is maintained by the Owner during partial utilization in accordance with Paragraph 14.05, after Substantial Completion in accordance with Paragraph 14.04 or after final payment in accordance

with Paragraph 14.07.

C. Provisions shall be included in any insurance policy maintained by Owner to prevent insurer from any recovery against Engineer, Contractor, Subcontractors and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each in the event of payment by insurer for any loss, damage or consequential loss.

# 5.08 Receipt and Application of Insurance Proceeds

- A. Any insured loss will be adjusted with Owner and made payable to Owner as fiduciary for the loss payees. Any money received will be deposited by Owner in a separate account and distributed in accordance with said agreement to the parties of interest. Unless a special agreement is reached, the money received will be applied to the repair or replacement of the Work and the cost will be covered by a Change Order.
- B. Unless one of the parties of interest objects in writing within fifteen days of an occurrence of loss, the Owner will have the power to adjust and settle any loss with the insurers. If an objection is made, it will be the responsibility of the Owner to adjust and settle any loss with the insurers in accordance to the new agreement. If no agreement is reached between the parties of interest, the Owner will adjust and settle the loss with insurers and if required in writing by any party of interest, Owner will be required to give bond for the proper performance of said duties.

# 5.09 Acceptance of Bonds and Insurance; Option to Replace

A. If any objection is made concerning coverage afforded by or other provisions of the bonds or insurance required to be purchased and maintained by either party in accordance with Article 5 based on non-conformance with the Contract Documents by either Owner or Contractor, the objecting party must inform the other party in writing within ten days of receipt of the certificates required by Paragraph 2.01.B. Both Owner and Contractor will be responsible for providing the other with additional information concerning insurance provided. If either of the parties does not purchase or maintain all of the required bonds and insurance as identified in the Contract Documents, that party must inform the other party in writing of their failure to purchase or maintain adequate bonds or insurance before the start or during the Work, before any change is made to the required coverage.

# 5.10 Partial Utilization, Acknowledgement of Property Insurer

A. If it becomes necessary for the Owner to occupy a portion or portions of the Work prior to Substantial Completion of all of the Work, as described in Paragraph 14.05, occupancy must not begin until after the insurers providing property insurance have been notified in writing and make the necessary changes in coverage. The property insurance shall not be canceled or allowed to lapse due to partial use or occupancy.

051.18.2.22

# <u>ARTICLE 6 – CONTRACTOR'S RESPONSIBILITIES</u>

6.01 Supervision and Superintendence

- A. The Contractor, in accordance with the Contract Documents, is responsible for competently supervising, inspecting and directing the Work. Means, methods, techniques, sequences and procedures of construction will be the responsibility of the Contractor. Any negligence by the Owner or Engineer in the design or specification of a specific means, method, technique, sequence or procedure of construction shown and required by the Contract Documents will not be the responsibility of the Contractor.
- B. The Contractor must provide a competent resident superintendent to reside over the Work at all times. The superintendent must not be replaced without written notice to Owner and Engineer, unless extraordinary circumstances come about.
- C. The Contractor shall have the sole responsibility of contacting any Utility owner with any facilities present at the site prior to any excavation. Contractor shall also be responsible for contacting Utilities Protection Center of Georgia prior to any excavation at the site.

#### 6.02 Labor; Working Hours

- A. It will be the responsibility of the Contractor to provide competent, qualified personnel to survey and lay out the Work as well as perform construction as mandated by the Contract Documents. While on the Site, the Contractor must display good discipline and order at all times during the progress of the Work.
- B. All Work at the Site shall be performed during regular working hours unless required for the safety or protection of persons, the Work or the Site and except as stated in the Contract Documents. Contractor shall notify Owner in writing no later than the Wednesday prior to the Saturday, Sunday or legal holiday that Contractor intends to work. Owner's written consent is required for Contractor to be permitted to work on a Saturday, Sunday or any legal holiday.

#### 6.03 Services, Materials, and Equipment

- A. Unless otherwise specified in the Contract Documents, Contractor shall provide and assume full responsibility for all services, materials, equipment, labor, transportation, construction equipment and machinery, tools, appliances, fuel, power, light, heat, telephone, water, sanitary facilities, temporary facilities, and all other facilities and incidentals necessary for the performance, testing, start-up, and completion of the Work.
- B. All materials and equipment to be used for the Work shall be of good quality and new, unless otherwise noted in the Contract Documents. All of the special warranties and guarantees required by the Specifications will be to the benefit of Owner. Contractor may be required by Engineer to provide evidence to verify the source, kind and quality of all materials and equipment used for the Work.
- C. All materials and equipment shall be stored, installed, applied, erected, protected, connected, cleaned, used and conditioned to the specifications required by the Supplier unless otherwise noted in the Contract Documents.

#### 6.04 *Progress Schedule*

- A. The Progress Schedule shall be followed by the Contractor as provided by Paragraph 2.07. Progress Schedule may be adjusted from time to time as provided below.
  - 1. Any proposed adjustments in the Progress Schedule that will not result in a change in the Contract Times must be submitted to Engineer by Contractor.
  - 2. Contractor shall submit any proposed adjustments to the Progress Schedule, which will change the Contract Times, in accordance with the requirements of Article 12. A Change Order is required for adjustments in Contract Times.

# 6.05 Substitutes and "Or-Equals"

- A. Any time an item of material or equipment is specified by using the name of a proprietary item or naming a particular Supplier in the Contract Documents, the intent is to establish the type, function, appearance and quality required. No substitution will be permitted if the description of an item of material or equipment is followed by words reading that no like, equivalent, "orequal" item or no substitution permitted. For all other items of different Suppliers, Contractor may submit for Engineer to review under the following circumstances:
  - 1. "Or-Equal" Items: If an item of material or equipment proposed by Contractor is determined by the Engineer to be functionally equal and sufficiently similar to the named item and result in no change in related Work, that item of material or equipment will be considered "or-equal" by the Engineer. The review and approval of an "or-equal" item does not require compliance with some or all of the requirements for approval of proposed substitute items. A proposed item of material or equipment will be considered equal to an item so named if:
    - a. Engineer determines through reasonable judgment that:
      - 1) at minimum it is equal in materials of construction, quality, durability, appearance, strength and design characteristics;
      - 2) it will achieve the desired results of the Project as a whole and it will equally and reliably perform the desired function; and
      - 3) it is available for responsive service and has an established record of performance.
    - b. If approved for and used for the completion of the Work, Contractor certifies that:
      - 1) change will not result in an increase in cost for the Owner or an increase in Contract Times; and
      - 2) this item will significantly conform to the detailed requirements in the Contract Documents of the item name.

#### 2. Substitute Items:

- a. If an item of material or equipment proposed by Contractor is determined by Engineer to not qualify as an "or-equal" item as described in Paragraph 6.05.A.1, it will be considered a proposed substitute item.
- b. It will be the responsibility of the Contractor to provide Engineer with sufficient information of the item of material or equipment for the determination of its equivalence and its ability to serve as a substitute. Engineer will only accept requests for review of proposed substitute items of material or equipment from the Contractor.
- c. Paragraph 6.05.A.2.d sets the requirements for review by Engineer.
- d. It is the responsibility of the Contractor to apply for review of a proposed substitute item of material or equipment in writing. The application:
  - 1) shall certify that the proposed substitute item will:
    - a) sufficiently perform the functions and accomplish all of the intended results of the general design,
    - b) be similar in substance to that specified, and
    - c) be intended for the same use as that specified;

## 2) will state:

- a) any effect the use of the proposed substitute item will have on the date of Substantial Completion,
- b) if use of the proposed substitute item will require an adaptation to the design of the Work and therefore result in a change to any of the Contract Documents, and
- c) if use of the proposed substitute item will result in any license fee or royalty payment;

#### 3) will identify;

- a) any and all differences between the specified item and the proposed substitute item, and
- b) all available sales, engineering, repair, maintenance and replacement services; and
- 4) shall include an itemized estimate of all costs or credits resulting either

directly or indirectly due to the use of the proposed substitute item. The costs of redesign and any claims by other contractors that will be affected by the change must be included in the itemized estimate.

- B. Substitute Construction Methods or Procedures: Contractor may use a substitute means, method, technique, sequence or procedure of construction in place of a specific means, method, technique, sequence or procedure of construction, which is required by the Contract Documents, only with Engineers approval. It will be the responsibility of the Contractor to submit adequate information to Engineer for the determination of the equivalency of the proposed substitute to the requirements of the Contract Documents. The requirements for review will be similar to the requirements in Paragraph 6.05.A.2.
- C. Engineer's Evaluation: A reasonable amount of time will be allowed for Engineer to evaluate each and every proposal made pursuant to Paragraphs 6.05.A and 6.05.B. Additional data about the proposed substitute item may be required by the Engineer and shall be furnished by the Contractor. The sole judge of acceptability will be the Engineer. At the conclusion of Engineer's review a determination will be identified by a Change Order for a substitute and an approved Shop Drawing for an "or-equal". Contractor will be notified in writing of any negative determination.
- D. *Special Guarantee:* Contractor may be required, at their own expense, to acquire a special performance guarantee or other surety with respect to any substitute.
- E. Engineer's Cost Reimbursement: Engineer will document all costs incurred by Engineer during the evaluation of a proposed substitute in accordance with Paragraphs 6.05.A.2 and 6.05.B. Regardless if Engineer approves a proposed or submitted substitute, Contractor shall compensate Engineer for all costs incurred during the evaluation of such proposed substitute. Contractor shall also compensate Engineer for all costs incurred from making adjustments in the Contract Documents that are consequential to the acceptance of a proposed substitute.
- F. *Contractor's Expense:* Any expenses incurred by Contractor from providing data in support of any proposed substitute or "or-equal" will be the responsibility of the Contractor.

#### 6.06 Concerning Subcontractors, Suppliers, and Others

- A. Any Subcontractor, Supplier or other individual or entity that Owner has a reasonable objection against shall not be employed by the Contractor. Contractor shall not be required by Owner to employ any Subcontractor, Supplier or other individual or entity against whom Contractor has any reasonable objection.
- B. Contractor may be required by the Supplementary Conditions to provide Owner with the identity of certain Subcontractors, Suppliers or other individuals or entities at a date prior to the Effective Date of the Agreement in order for Owner to approve of Subcontractors, Suppliers or other individuals or entities. After investigation, Owner's acceptance (either in writing or by not making a written objection by the date for acceptance in the Bidding Documents or the Contract Documents) may be revoked due to reasonable objection. If Owner rejects any Subcontractor, Supplier or other individual or entity, it will be the responsibility of the Contractor to submit to

Owner an acceptable replacement. A Change Order will be issued if an adjustment to the Contract Price is necessary due to the replacement. Owner maintains the right to reject defective Work regardless of acceptance of any Subcontractor, Supplier or other individual or entity.

- C. All acts and omissions of the Subcontractors, Suppliers and other individuals or entities while performing or furnishing any of the Work will be the sole responsibility of the Contractor. Nothing in the Contract Documents shall:
  - 1. create any contractual relationship between Owner or Engineer and any such Subcontractor, Supplier or other individual or entity for the benefit of any such Subcontractor, Supplier or other individual or entity.
  - 2. obligate Owner to pay or see to the payment of any such Subcontractor, Supplier or other individual or entity unless required by Laws and Regulations.
- D. Scheduling and Coordinating the Work of Subcontractors, Suppliers and other individuals or entities performing or furnishing any of the Work, whether in direct or indirect Contract with Contractor, will be the sole responsibility of Contractor.
- E. Subcontractors, Suppliers and other individuals or entities will be required by Contractor to communicate with Engineer through Contractor.
- F. Contractor will not be restricted by the divisions and sections of the Specifications and identifications of any Drawings for the division of any Work among Subcontractors or Suppliers or delineating the Work to be performed by any specific trade.
- G. Subcontractor or Supplier will be bound to the applicable terms and conditions of the Contract Documents for all Work performed for Contractor by an appropriate agreement between Contractor and the Subcontractor or Supplier. Any agreement between Contractor and Subcontractor or Supplier who is listed as a loss payee on the property insurance according to Paragraph 5.06, will contain provisions that waive all rights of the Subcontractor or Supplier against Owner, Engineer, Contractor and all other loss payees listed in the Supplementary Conditions for any and all losses and damages caused by, relating to, resulting from or arising out of any of the perils or causes of loss covered by such policies and any other property insurance applicable to the Work. Contractor must obtain any separate waiver forms that may be required by insurers of any such policies and have them signed by any Subcontractor or Supplier.

#### 6.07 Patent Fees and Royalties

- A. If any invention, design, process, device or product, which is subject to patent rights or copyrights of others, is used in the performance of the Work or is incorporated into the Work the Contractor will be responsible to pay all license fees and royalties and assume all costs incident therefore.
- B. If any invention, design, process, product or device is specified in the Contract Documents, but

not identified as being subject to payment of any license fee or royalty as required by patent rights or copyrights, for incorporation in the Work or in the performance of the Work and results in any claims, costs, losses and damages due to patent rights or copyright infringement, Owner shall indemnify and hold harmless Contractor to the fullest extent permitted by Laws and Regulations against all claims, costs, losses and damages.

C. If any invention, design, process, product or device not specified in the Contract Documents is incorporated in the Work or in the performance of the Work and results in any claims, costs, losses and damages due to patent rights or copyright infringement, Contractor shall indemnify and hold harmless Owner and Engineer to the fullest extent permitted by Laws and Regulations against all claims, costs, losses and damages.

#### 6.08 Permits

A. It is the responsibility of the Contractor to obtain and pay for all construction permits and licenses unless noted otherwise in the Supplementary Conditions. Owner will assist Contractor in obtaining any necessary permits and licenses. Any governmental charges and inspection fees necessary for the execution of the Work, which are applicable at the time of opening of Bids or, in the case of no Bids, on the Effective Date of the Agreement, shall be paid by the Contractor. Any charges resulting from connections providing permanent service to the Work by utility owners shall be paid by the Owner.

# 6.09 Laws and Regulations

- A. It is the responsibility of the Contractor to comply and give all notices required by all Laws and Regulations that apply to the performance of the Work. Neither Owner nor Engineer shall be responsible for ensuring Contractor's compliance with any Laws or Regulations unless required by Laws and Regulations.
- B. Contractor shall bear all claims, costs, losses and damages that arise due to or relating to the performance of the Work in which Contractor knows or has reason to know that it is contrary to any Laws or Regulations. It is not the responsibility of the Contractor to verify that the Specifications and Drawings are in accordance with Laws and Regulations, but the Contractor is still obligated to the requirements of Paragraph 3.03.
- C. An adjustment in Contract Price or Contract Times will be made if any changes in Laws or Regulations that effect either the cost or time of performance of the Work arise that were not known at the time of opening of Bids. A Claim may be made as provided in Paragraph 10.05 if Owner and Contractor are unable to agree upon any necessary adjustment.

# 6.10 Taxes

A. All sales, consumer, use and other similar taxes required to be paid in accordance with the Laws and Regulations of the place of the Project that are applicable during the performance of the Work shall be paid by the Contractor.

051.18.2.22

# 6.11 *Use of Site and Other Areas*

# A. Limitation on Use of Site and Other Areas:

- All construction equipment, stored materials and equipment and the operations of
  workers shall be confined to the Site and other areas permitted by Laws and Regulations
  by the Contractor. The Site and other areas shall never be encumbered with construction
  equipment or other materials or equipment. Any damage to any land or area or to the
  owner or occupant of land or any adjacent land or area that occurs from the performance
  of the Work will be the sole responsibility of the Contractor.
- 2. It is the responsibility of the Contractor to promptly settle by negotiation or resolve by arbitration or other dispute resolution process any claim made by any owner or occupant as a result of the performance of the Work.
- 3. If any claim or action, legal or equitable, brought by any such owner or occupant is made based upon Contractor's performance of the Work, Contractor shall indemnify and hold harmless Owner and Engineer to the fullest extent permitted by Laws and Regulations against all claims, costs, losses and damages.
- B. *Removal of Debris during Performance of the Work:* Contractor shall keep the Site and other areas void of any waste materials, rubbish and other debris throughout the progress of the Work.
- C. Cleaning: It is the responsibility of the Contractor to clean the Site and the Work and make it ready for use by Owner prior to Substantial Completion of the Work. Contractor shall also remove all tools, appliances, construction equipment and machinery and surplus materials from the Site and all property not shown for modification by the Contract Documents shall be restored to original condition at the conclusion of the Work.
- D. Loading Structures: Contractor shall not load or permit the loading of any structure in a way that may compromise the structure. No part of the Work or any adjacent property shall be subjected to stresses or pressures that will endanger it by the Contractor.

## 6.12 Record Documents

A. Record Documents shall be maintained in accordance with Section 01720.

# 6.13 Safety and Protection

- A. It is the sole responsibility of the Contractor to initiate, maintain and supervise all safety programs and precautions that are related to the Work. It is the responsibility of Subcontractors to perform their work in a manner that provides safety for the persons or property that is part of their work. Subcontractors shall also comply with all applicable safety Laws and Regulations. It is the responsibility of the Contractor to provide all necessary protection for the prevention of damage, injury or loss to as well as taking all necessary precautions for the safety of:
  - 1. all persons who may be affected by the Work who are present at the Site;

- 2. all work, materials and equipment that will be incorporated into the Work, whether on or off the Site in storage; and
- 3. other property, including trees, shrubs, lawns, walks, pavements, roadways, structures, utilities and Underground Facilities that are not designated for removal, relocation or replacement during the course of construction, at or adjacent to the Site.
- B. All applicable Laws and Regulations related to the safety of persons or property, or the protection of persons or property from damage, injury or loss must be complied with by Contractor. All safeguards that are necessary for the safety and protection of persons or property must be erected and maintained by Contractor. Owners of adjacent property, of Underground Facilities and utility owners must be notified by Contractor prior to execution of any Work that may affect them. Contractor shall also assist them in the protection, removal, relocation and replacement of their property.
- C. Owner's safety programs are identified in Part 20 of the Supplementary Conditions and will be the safety requirements the Contractor shall comply with.
- D. Owner and Engineer shall be notified of any specific requirements of Contractor's safety program in order to ensure compliance by Owner's and Engineer's employees while at the Site.
- E. Contractor shall remedy all damage, injury or loss to any property referred to in Paragraph 6.13.A.2 or 6.13.A.3 that is caused, directly or indirectly, in whole or in part by Contractor, any Subcontractor, Supplier or any other individual or entity that is employed directly or indirectly by any of them for the performance of any of the Work or anyone whose acts any of them may be liable for. Contractor shall not be liable for any damage, injury or loss that is attributable to the fault of Drawings, Specifications, acts or omissions of Owner, Engineer or anyone employed by the Owner and is not attributable to the fault or negligence of Contractor, Subcontractor, Supplier or other individual or entity directly or indirectly employed by any of them.
- F. Contractor shall initiate, maintain and supervise all safety precautions and protection of the Work until Engineer issues Contractor with a written notice, in accordance with Paragraph 14.07.B, stating the Work is acceptable and complete.

## 6.14 Safety Representative

A. An experienced and qualified safety representative shall be designated to the Site by the Contractor. The safety representative shall be responsible for maintaining and supervising safety precautions and programs and the prevention of accidents.

# 6.15 Hazard Communication Programs

A. It is the responsibility of the Contractor to coordinate between and among employers at the Site the exchange of material safety data sheets or other hazard communication information in accordance with Laws or Regulations.

051.18.2.22

# 6.16 Emergencies

A. Contractor shall act in a way to prevent damage, injury or loss of persons, the Work, property at the Site or adjacent to the Site in the event of any emergency that may affect the safety or protection of such. If Contractor believes any significant changes in the Work or variations from the Contract Documents have resulted from action taken by Contractor in response to such an emergency Contractor shall promptly provide Engineer with written notice. Engineer may issue a Change Order or a Work Change Directive if Engineer determines that the action taken by Contractor in response to such an emergency requires a change in the Contract Documents.

# 6.17 *Shop Drawings and Samples*

- A. Shop Drawings and Samples shall be submitted by Contractor to Engineer as required by Paragraph 2.07 in accordance with Section 01340. Each submittal will be identified as required by Engineer.
  - 1. Shop Drawings:
    - a. Submit required number of copies as specified in Section 01340.
    - b. Shop Drawings will clearly and completely show quantities, dimensions, specified performance, design criteria, materials and similar data to allow Engineer to review and determine, as required by Paragraph 6.17.D, the services, materials and equipment proposed by Contractor.

# 2. Samples:

- a. Submit required number of Samples as specified in Section 01340.
- b. Supplier, material, pertinent data such as catalog numbers, intended use and other data shall be clearly identified for each Sample to allow Engineer to review the submittal as required by Paragraph 6.17.D.
- B. Any Work specified by the Contract Documents or the Schedule of Submittals to require a Shop Drawing or Sample and is performed prior to Engineer's review and approval of submittal will be the sole expense and responsibility of Contractor.
- C. Submittal Procedures:
  - 1. Prior to the submission of each Shop Drawing or Sample, Contractor shall have:

- a. reviewed and coordinated each Shop Drawing with the requirements of the Work and the Contract Documents:
- b. with respect to Shop Drawing or Samples, determined and confirmed all field measurements, quantities, dimensions, installation requirements, materials, catalog numbers, specified performance and design criteria and similar information;

- c. with respect to the specified application fabrication, shipping, handling, storage, assembly and installation, determined and confirmed the suitability of all materials offered pertaining to the performance of the Work; and
- d. determined and confirmed means, methods, techniques, sequences, procedures of construction and safety precautions that are the responsibility of the Contractor.
- 2. A stamp or written certification is required of each submittal to show that Contractor has satisfied the review and approval of submittals as required by the Contract Documents.
- 3. Contractor shall include written notice with any submittal to Engineer in which the Shop Drawing or Sample varies from the requirements of the Contract Documents. For Engineer's review and approval two notices are required: (i) one shall be a written communication that is separate from the Shop Drawings or Sample submittal and (ii) a notation made on each Shop Drawing or Sample.

# D. Engineer's Review:

- Shop Drawings and Samples will be reviewed by Engineer in a timely manner in accordance with the Schedule of Submittals. The review and approval will be to determine the conformity of the items covered by the submittals to the information given in the Contract Documents after installation or incorporation into the Work. The review will also determine if the items covered by the submittals will be compatible, as indicated by the Contract Documents, to the design concept of the finished Project as a functioning whole.
- 2. Means, methods, techniques, sequences, procedures of construction (unless specific means, methods, techniques, sequences or procedures of construction are required by the Contract Documents) or safety precautions or programs will not be reviewed or approved by Engineer.
- 3. Contractor shall be responsible for any variation from the requirements of the Contract Documents regardless of Engineer's review and approval except when Contractor has complied with the requirements of Paragraph 6.17.C.3 and written approval of each written notation incorporated in or accompanying the Shop Drawing or Sample has been given by Engineer. Contractor shall not be relieved from responsibility of complying with the requirements of Paragraph 6.17.C.1 regardless of review and approval by Engineer.

#### E. Resubmittal Procedures:

1. All corrections required by Engineer shall be made by Contractor and the required number of corrected copies of Shop Drawings shall be returned and if required new Samples submitted for review and approval.

051.18.2.22

# 6.18 Continuing the Work

A. During all disputes or disagreements with Owner, Contractor shall continue the Work and adhere to the Progress Schedule. Except with a written agreement between Owner and Contractor, no Work shall be delayed or postponed by any dispute or disagreement except as permitted by Paragraph 15.04.

# 6.19 Contractor's General Warranty and Guarantee

- A. All Work shall be warranted and guaranteed to Owner to be in accordance with the Contract Documents and free of defects by Contractor.
- B. Warranty and guarantee by the Contractor excludes defects or damage caused by:
  - 1. abuse, modification or improper maintenance or operation by any person other than Contractor, Subcontractor, Supplier or any other person who is the responsibility of the Contractor; or
  - 2. wear and tear as the result of normal usage.
- C. The Work shall be performed and completed in absolute accordance with the Contract Documents by the Contractor. The following will not establish an acceptance of Work that is not in accordance with the Contract Documents or a release of Contractor's requirements to execute the Work as required by the Contract Documents:
  - 1. observations by Owner or Engineer;
  - 2. any progress or final payment by Owner;
  - 3. any payment related to or issuance of a certificate of Substantial Completion by Owner or Engineer;
  - 4. Owner's use or occupancy of any part of the Work;
  - 5. a Shop Drawing or Sample submittal that is reviewed and approved by Engineer or the issuance of a notice of acceptability by Engineer;
  - 6. any inspection, test or approval by others; or
  - 7. the correction of any defective Work by Owner.

#### 6.20 Indemnification

A. Contractor shall indemnify and hold harmless Owner, Engineer and Owner's, officers, directors, members, partners, employees, agents, consultants and subcontractors to the fullest extent permitted by Laws and Regulations against all claims, costs, losses and damages that arise out of or are related to the performance of the Work provided that said claim, cost, loss or damage is a result of bodily injury, sickness, disease, death or injury to or destruction of tangible property, that includes the loss of use that is attributable to an omission or negligent act by Contractor,

- Subcontractor, Supplier or other individual directly or indirectly employed by any of them for the performance of any of the Work or anyone whose acts they are liable for.
- B. The indemnification obligation under Paragraph 6.20.A shall not be limited based on the amount or type of damages, compensation or benefits payable in any way by or for Contractor or any Subcontractor, Supplier or other individual or entity under workers' compensation acts, disability benefit acts or other employee benefit acts for any and all claims against Owner, Engineer or any of Owner's officers, directors, members, partners, employees, agents, consultants or subcontractors by any employee of Contractor, Subcontractor, Supplier or any individual or entity directly or indirectly employed by any of them for the performance of any of the Work.
- C. The Contractor shall not be obligated to indemnify, under Paragraph 6.20.A, the liability of the Engineer arising from:
  - 1. the preparation, failure to prepare, approval or failure to approve of maps, Drawings, opinions, reports, surveys, Change Orders, designs or specifications; or
  - 2. giving or failing to give instructions or directions, if that is the primary cause of the injury or damage.

# 6.21 Delegation of Professional Design Services

- A. The obligations of providing professional design services will not be the responsibility of the Contractor except when: (i) Contract Documents require such services for a portion of the Work or (ii) the execution of Contractor's responsibilities for construction means, methods, techniques, sequences and procedures require such services. Contractor shall not be required to provide professional services that will result in the violation of any applicable law.
- B. Engineer will specify all performance and design criteria that must be satisfied if professional design services or certifications by a design professional, in relation to systems, materials or equipment, are required of Contractor by the Contract Documents. All service or certifications that are the responsibility of the Contractor must be provided by a properly licensed professional. All drawings, calculations, specifications, certifications, Shop Drawings and other submittals must bear the signature and seal of such professional. If Shop Drawings and other submittals that are related to the Work are prepared by others and designed or certified by such professional, such professional's written approval must be submitted to Engineer.
- C. Provided Owner and Engineer have specified to Contractor all performance and design criteria that such service must satisfy, Owner and Engineer shall be allowed to rely on the services, certifications or approvals performed by such design professionals based on the adequacy, accuracy and completeness of the design.
- D. In accordance with this Paragraph 6.21, review and approval of design calculations and drawings by Engineer will only be to ensure that such calculations and drawings conform to the performance and design criteria and concept expressed in the Contract Documents. Review and approval by Engineer of Shop Drawings and other submittals (except design calculations and

design drawings) will be performed only for the purpose stated in Paragraph 6.17.D.1.

E. The adequacy of the Contract Documents for performance or design criteria shall not be the responsibility of Contractor.

# **ARTICLE 7 – OTHER WORK AT THE SITE**

## 7.01 Related Work at Site

- A. Owner is allowed to perform other work at the Site that is related to the Project with Owner's employees, through other direct contracts or by utility owners. If no note of such other work is in the Contract Documents, then:
  - 1. Contractor will be given written notice prior to any such other work starting; and
  - 2. if no adjustment in the Contract Price or Contract Times, as a result of such other work, can be agreed upon by Owner and Contractor, a Claim may be made as provided in Paragraph 10.05.
- B. Contractor shall afford proper and safe access to the Site, reasonably accommodate the introduction and storage of materials and equipment and the execution of such other work and coordinate such work with their Work to each contractor who is a party to a direct contract with Owner, each utility owner and Owner if Owner's employees are performing any of the other work. Any cutting, fitting or patching of the Work that is required to properly connect or integrate such other work will be the responsibility of the Contractor. Any cutting, excavating or altering of any work of others shall not be allowed by Contractor; however with written consent of Owner and others whose work will be affected Contractor may cut or alter others' work. The duties and responsibilities in this Paragraph of Contractor are for the benefit of other contractors and utility owners while provisions in said direct contracts between Owner and such utility owners and other contractors exist for the benefit of Contractor.
- C. It is the responsibility of Contractor to inspect work of others if any part of Contractor's Work is dependent upon the proper execution or results of work performed by others under this Article 7. If any delays, defects or deficiencies are found in the work of others that render it unavailable or unsuitable for the proper execution and results of Contractor's Work, Contractor must promptly notify Engineer in writing. If Contractor fails to report any delays, defects or deficiencies it will constitute Contractor's acceptance of such other work as satisfactory for integration with Contractor's Work except in the case of latent defects and deficiencies in such other work.

## 7.02 Coordination

- A. If the performance of other work on the Project at the Site is intended to be contracted to others by Owner, the following will be stated in the Supplementary Conditions:
  - 1. identification of the individual or entity who will have the authority and responsibility for coordination of activities between the various contractors;

- 2. itemization of specific matters covered by such authority and responsibility;
- 3. extent of such authority and responsibilities.
- B. Authority and responsibility for coordination will be the responsibility of the Owner unless otherwise provided for in the Supplementary Conditions.

# 7.03 Legal Relationships

- A. Paragraphs 7.01.A and 7.02 do not apply to any utilities that are not under the control of Owner.
- B. Other contractors under direct contract to Owner are liable to Owner and Contractor, under Paragraph 7.01, for contractor's wrongful actions or inactions that result in direct delay and disruption costs incurred by Contractor.
- C. Contractor under direct contract to Owner is liable to Owner and any other contractor for Contractor's wrongful actions or inactions that result in direct delay and disruption costs incurred by such other contractor.

# **ARTICLE 8 – OWNER'S RESPONSIBILITIES**

- 8.01 *Communications to Contractor* 
  - A. Unless otherwise stated in these General Conditions, all communications of Owner to Contractor shall be directed through Engineer.
- 8.02 Furnish Data
  - A. Data required of Owner under the Contract Documents shall be promptly furnished by Owner.
- 8.03 Pay When Due
  - A. As provided in Paragraphs 14.02.C and 14.07.C, Owner shall make payments when they are due to Contractor.
- 8.04 Lands and Easements; Reports and Tests
  - A. As set forth in Paragraphs 4.01 and 4.05, Owner shall provide lands, easements and engineering surveys for the establishment of reference points. Owner shall identify and make available copies of reports of explorations and tests of subsurface conditions and drawings of physical conditions that relate to the existing surface or subsurface structures at the site for Contractor as required by Paragraph 4.02.

051.18.2.22

8.05 Insurance

A. All responsibilities of Owner to purchase or maintain liability and property insurance are set forth in Article 5.

# 8.06 Change Orders

- A. As indicated in Paragraph 10.03, it is the obligation of the Owner to execute Change Orders.
- 8.07 Inspections, Tests, and Approvals
  - A. Paragraph 13.03.B indicates the responsibility of the Owner with respect to certain inspections, tests and approvals.
- 8.08 Limitations on Owner's Responsibilities
  - A. Means, methods, techniques, sequences, procedures of construction, safety precautions and programs or failure of Contractor to comply with Laws and Regulations that apply to the performance of the Work will not be the responsibility of Owner to supervise, direct or have control or authority over. Contractor's failure to execute the Work as required by the Contract Documents will not be the responsibility of the Owner.
- 8.09 Undisclosed Hazardous Environmental Conditions
  - A. Responsibility to an undisclosed Hazardous Environmental Condition of the Owner is described in Paragraph 4.06.
- 8.10 Evidence of Financial Arrangements
  - A. If Contractor request, Owner is required to provide Contractor with evidence that financial arrangements have been made that will satisfy Owner's obligations under the Contract Documents.
- 8.11 *Compliance with Safety Program* 
  - A. In accordance with Paragraph 6.13.D, Contractor must inform Owner of any specific safety requirements mandated by the Contractor at the Site. All employees and representatives of the Owner must comply with any specific applicable requirements of Contractor's safety programs while at the site.

# **ARTICLE 9 – ENGINEER'S STATUS DURING CONSTRUCTION**

- 9.01 Owner's Representative
  - A. Engineer will be Owner's representative throughout the construction of the project. The duties and responsibilities of Engineer, as Owner's representative, are described in the Contract Documents.

051.18.2.22

9.02 Visits to Site

- A. Engineer or a representative of the Engineer will routinely visit the Site to observe the progress that has been made and the quality of the various aspects of Contractor's executed Work. Engineer will determine from such visits, for the benefit of Owner, if the Work is progressing as required by the Contract Documents. Engineer will help keep Owner abreast of the progress of the Work and will strive to protect Owner from defective Work.
- B. Engineer will not supervise, direct, control, have authority over or be responsible for Contractor's means, methods, techniques, sequences, procedures of construction or the safety precautions and programs incident to the Project or for Contractor's failure to conform with Laws and Regulations relevant to the performance of the Work.

# 9.03 Project Representative

A. Engineer may utilize a Resident Project Representative to allow more extensive observation of the Work. The authority and responsibility of such Resident Project Representative will be as provided in the Supplementary Conditions. The Resident Project Representative's responsibilities will be limited as provided in Paragraph 9.09.

#### 9.04 Authorized Variations in Work

A. Minor variations, which cause no adjustment in the Contract Price or Contract Times, in the Work from the requirements of the Contract Documents but that are compatible with the concept of the Project as a functioning whole may be authorized by Engineer. Any minor variations made will be accomplished by a Field Order and will be binding on Owner and Contractor, who will be responsible for the prompt performance of the Work. If an adjustment in the Contract Price or Contract Times is believed to be necessary by Owner or Contractor, a Claim may be made as provided in Paragraph 10.05.

# 9.05 Rejecting Defective Work

A. If any Work is believed to be defective, not in conformance to the Contract Documents or compromise the integrity of the design concept of the completed Project as a functioning whole as indicated by the Contract Documents by Engineer, Engineer shall have the authority to reject any such work. As provided in Paragraph 13.04, Engineer will also have authority to have special inspections or testing of the Work, regardless if the Work is fabricated, installed or completed.

#### 9.06 Shop Drawings, Change Orders, and Payments

- A. See Paragraph 6.17 for Engineer's authority or limitations of authority in regards to Shop Drawings and Samples.
- B. See Paragraph 6.21 for Engineer's authority or limitations of authority in regards to design calculations and design drawings submitted in response to a delegation of professional design services.

- C. See Articles 10, 11 and 12 for Engineer's authority in regards to Change Orders.
- D. See Article 14 for Engineer's authority in regards to Applications for Payment.

# 9.07 Determination for Unit Price Work

A. The quantities and classifications of Unit Price Work performed by Contractor will be determined by Engineer. Before rendering a recommendation of an Application for Payment, Engineer will review Engineer's preliminary determinations on quantities and classifications of Unit Price Work with Contractor. Subject to the provisions of Paragraph 10.05, Engineer's decision will be final and binding upon Contractor.

# 9.08 Decisions on Requirements of Contract Documents and Acceptability of Work

- A. Responsibility for initial interpretation of the requirements of the Contract Documents and judgment of the acceptability of the Work will be the Engineer's. Engineer must be contacted within thirty days by Owner or Contractor about any concern that arises prior to the date final payment is due in regards to the acceptability of the Work and the interpretation of the requirements of the Contract Documents as they pertain to the performance of the Work.
- B. A written decision will promptly be rendered by Engineer on the issue. A Claim may be made under Paragraph 10.05 if an adjustment in the Contract Price or Contract Time is believed to be necessary. For the purposes of Paragraph 10.05.B, the date of Engineer's decisions shall be the date of the event giving rise to the issues referenced.
- C. Subject to the provisions of Paragraph 10.05, Engineer's written decision on the issue will be final and binding on Contractor.

#### 9.09 *Limitations on Engineer's Authority and Responsibilities*

- A. No decision made by Engineer in the exercise or non-exercise of authority or responsibility of Engineer shall create, impose or give rise to any duty in contract, tort or otherwise owed by Engineer to Contractor, Subcontractor, Supplier, other individual or entity or any surety for or employee or agent of any of them.
- B. Contractor's means, methods, techniques, sequences, procedures of construction, safety precautions and programs or any failure to comply with Laws and Regulations that apply to the performance of the Work will not be supervised, directed or controlled by Engineer and Engineer will not be responsible or have authority of any such activity of Contractor. Any failure in the performance of the Work in accordance with the Contract Documents by Contractor will not be the responsibility of Engineer.
- C. No acts or omissions of Contractor or any Subcontractor, Supplier or other individual or entity performing any of the Work will be the responsibility of the Engineer.
- D. The review of the final Application of Payment and accompanying documentation and all maintenance and operating instructions, schedules, guarantees, bonds, certificates of inspection,

tests and approvals, and other documentation to be delivered by Paragraph 14.07.A will be performed by Engineer to generally determine that their content or certified results satisfy the requirements of the Contract Documents.

# **ARTICLE 10 – CHANGES IN THE WORK; CLAIMS**

## 10.01 Authorized Changes in the Work

- A. Owner or Engineer may order additions, deletions or revisions in the Work through a Change Order or a Work Change Directive at any time or from time to time without invalidating the Contract and without notification to any surety. Contractor shall promptly proceed with the Work involved which will be performed under the applicable conditions of the Contract Documents upon receipt of any such document.
- B. An adjustment in Contract Price or Contract Times will be made if any changes in Laws or Regulations that effect either the cost or time of performance of the Work arise that were not known at the time of opening of Bids. Claim may be made as provided in Paragraph 10.05 if Owner and Contractor are unable to agree upon any necessary adjustment.

# 10.02 Unauthorized Changes in the Work

A. Contract Price will not be increased and Contract Times will not be extended for any work Contractor performs that is not a requirement of the Contract Documents as amended, modified or supplemented in Paragraph 3.04 except in the case of: (i) any emergency as provided in Paragraph 6.16, or (ii) uncovering Work as provided in Paragraph 13.04.D.

# 10.03 Execution of Change Orders

- A. Change Orders endorsed by Engineer covering the following shall be executed by Owner and Contractor:
  - 1. changes in the Work which are: (i) ordered by Owner pursuant to Paragraph 10.01.A, (ii) required due to acceptance or correction of defective Work under Paragraphs 13.08.A or 13.09, or (iii) agreed to by both parties;
  - 2. changes, which are agreed upon by both parties, in the Contract Price or Contract Times, including any undisputed sum or amount of time for Work actually performed in accordance with a Work Change Directive; and
  - 3. changes which embody the substance of any written decision rendered by Engineer in the Contract Price or Contract Times in accordance with Paragraph 10.05; given that, an appeal may be taken from any such decision in accordance with the provisions of the Contract Documents and applicable Laws and Regulations in lieu of executing any such Change Order, but Contractor shall adhere to the Progress Schedule and continue the execution of the Work during any such appeal in accordance with Paragraph 6.18.A.

10.04 *Notification to Surety* 

A. It will be the responsibility of the Contractor to notify any surety, if the provisions of any bond require notification, of any change that affects the general scope of the Work or the requirements of the Contract Documents. Each applicable bond must be adjusted to reflect the effect of any such change.

#### 10.05 *Claims*

- A. *Engineer's Decision Required:* All Claims, except those waived in accordance with Paragraph 14.09, shall be submitted to Engineer for decision. Owner or Contractor must refrain from exercising any rights or remedies they have under the Contract Documents or by Laws and Regulations applicable to such Claims until a decision by Engineer has been rendered.
- B. *Notice:* Engineer shall be delivered written notice that states the nature of each Claim after the start but no later than sixty days after the event. The party making the Claim will bear the responsibility of substantiating the Claim. The Claimant shall deliver to Engineer and other party to the Contract, within sixty days after the start of such event, notice of the amount or extent of the Claim along with all supporting data. Paragraph 12.01.B provides the requirements for a Claim for an adjustment in Contract Price. Paragraph 12.02.B provides the requirements for a Claim for an adjustment in Contract Times. Claimant's written statement for each Claim must state that the claimed adjustment is the entire adjustment claimant believes it is entitled to as a result of such event.
- C. *Engineer's Action:* Each Claim will be reviewed by Engineer within thirty days after receipt of last submittal by Contractor and take one of the following actions in writing:
  - 1. deny the Claim in whole or in part; or
  - 2. approve the Claim.
- D. The Claim shall be considered denied if Engineer takes no action on a Claim within thirty days.
- E. Owner or Contractor will be bound to Engineer's written action under Paragraph 10.05.C or denial in accordance with Paragraphs 10.05.C.3 or 10.05.D, unless Owner or Contractor invoke the dispute resolution procedure presented in Article 16 within thirty days of said action or denial.
- F. If Claim for an adjustment in Contract Price or Contract Times is not submitted as required by this Paragraph 10.05, the Claim will be considered invalid.

# ARTICLE 11 – COST OF THE WORK; ALLOWANCES; UNIT PRICE WORK

#### 11.01 *Cost of the Work*

A. *Costs Included:* The sum of all cost necessarily incurred and paid by Contractor in the proper performance of the Work, except any excluded by Paragraph 11.01.B, will be termed Cost of the Work. Contractor will only be reimbursed those additional or incremental costs

when the value of any Work covered by a Change Order or when a Claim for an adjustment in Contract Price is determined based on Cost of the Work. Unless agreed to by Owner in writing, such costs shall not exceed prevailing costs in the locality of the Project, shall exclude all of the costs itemized in Paragraph 11.01.B and shall include only the following:

- 1. payroll cost for persons under the direct employment of Contractor who are performing the Work under schedules of job classifications agreed to by Owner and Contractor. Included in such employees are superintendents, foremen and other full time employed personnel performing the Work. For employees who are not employed full time on the Work, payroll costs shall be allotted based on the time they spend on the Work. Included in payroll costs shall be, but not limited to, salaries and wages plus the cost of fringe benefits, which shall include social security contributions, unemployment, excise and payroll taxes, workers' compensation, health and retirement benefits, bonuses, sick leave and vacation and holiday pay. To the extent authorized by Owner, any overtime expenses accumulated due to performing Work outside of regular working hours on Saturday, Sunday or legal holidays shall be included in the above.
- 2. cost of all equipment and materials Contractor furnishes and incorporates into the Work, including any costs associated with the transportation and storage of such materials and any field service cost of Supplier in connection with materials. Contractor shall accrue all cash discounts except when Owner deposits funds with Contractor (Owner will accrue cash discounts in this case), which Contractor uses to make payments. Owner shall accrue all trade discounts, rebates, refunds and returns from sale of surplus materials and equipment. It is the responsibility of the Contractor to make provisions so that trade discounts, rebates, refunds and returns from sale of surplus materials and equipment may be obtained.
- 3. all payments made to Subcontractors by Contractor for the performance of Work by Subcontractors. Owner may require Contractor to obtain competitive bids from subcontractors and deliver such bids to Owner for acceptance. If a subcontract established that payment of Subcontractor will be based on Cost of the Work plus a fee, the Subcontractor's Cost of the Work and fee shall be determined as provided in this Paragraph 11.01.
- 4. cost associated with the employment of special consultants (including but not limited to engineers, architects, testing laboratories, surveyors, attorneys and accountants) for any services that specifically relate to the Work.
- 5. Supplemental costs shall include the following:
  - a. expenses incurred in the discharge of duties connected with the Work by Contractor's employees due to necessary transportation, travel and subsistence.
  - b. cost associated with the transportation and maintenance of all materials, supplies, equipment, machinery, appliances, and office and temporary facilities at the site. Cost of any hand tools, not owned by workers, which are consumed through the performance of the Work and cost, less market value, of items that remain the

property of Contractor.

- c. cost accrued due to the rental of all construction equipment and machinery and costs associated with transportation, loading, unloading, assembly, dismantling and removal of such, whether rented from Contractor or others in accordance with a rental agreement that is approved by Owner. Said rental agreements will provide all requirements for such cost. At the time that such equipment, machinery or parts are no longer mandatory for the execution of the Work the rental of such shall conclude.
- d. taxes related to the Work such as sales, consumer, use and other similar taxes imposed by Laws and Regulations for which Contractor is liable.
- e. lost deposits that occur for causes other than negligence by Contractor, Subcontractor or anyone directly or indirectly employed by any of them or anyone whose acts they may be liable for. All royalty payments and fees for permits and licenses required for the performance of the Work.
- f. losses and damages related to the performance of the Work sustained by Contractor that are caused by damage to the Work and are not compensated by insurance or otherwise, provided the losses and damages occurred for reasons other than the negligence of Contractor, Subcontractor or anyone directly or indirectly employed by any of them or anyone whose acts they may be liable for. Such losses shall include settlements made with the written consent and approval of Owner. Cost of the Work will not include such losses, damages and expenses for the determination of Contractor's fee.
- g. cost of utilities, fuel and sanitary facilities at the Site.
- h. any minor expenses related to the performance of the Work such as telegrams, long distance telephone calls, telephone service at the Site, express and courier services and similar petty cash items.
- i. cost of premiums for all bonds and insurance the Contract Documents require Contractor to purchase and maintain.
- B. Costs Excluded: None of the following items will be included in the term Cost of the Work:
  - 1. cost resulting from payroll and other compensation of Contractor's officers, executives, principals, general managers, safety managers, engineers, architects, estimators, attorneys, auditors, accountants, purchasing and contracting agents, expediters, timekeepers, clerks and other personnel employed by Contractor, whether at the Site or in Contractor's principal or branch office, for the general administration of the Work and not specifically included in the agreed upon schedule of job classifications referred to in Paragraph 11.01.A.1 or specifically covered by Paragraph 11.01.A.4.
  - 2. expenses incurred by Contractor's principal and branch offices excluding Contractor's

office at the Site.

- 3. all of Contractor's capital expenses including interest on Contractor's capital employed for the Work and charges due to delinquent payments against Contractor.
- 4. cost incurred as the result of negligence including but not limited to the correction of defective Work, disposal of materials or equipment mistakenly supplied and repairing any property damaged by Contractor, Subcontractor or anyone directly or indirectly employed by any of them or anyone for whose acts they may be liable.
- 5. other overhead or general expense cost of any kind and costs of any item not specifically and expressly included in Paragraphs 11.01.A.
- C. Contractor's Fee: Contractor's fee shall be determined as described in the Agreement when all of the Work is performed on the basis of cost-plus. Contractor's fee shall be determined as required by Paragraph 12.01.C whenever the value of any Work, that is the subject of a Change Order or an adjustment in Contract Price due to a Claim, is determined on the basis of Cost of the Work.
- D. *Documentation:* Contractor shall establish and maintain records in accordance with generally accepted accounting practices whenever the Cost of the Work, for any purpose, is to be determined based on Paragraphs 11.01.A and 11.01.B. Contractor shall submit all such records in itemized breakdown form together with supporting data to Owner.

# 11.02 Allowances

A. It is understood that all allowances named in the Contract Documents have been included in the Contract Price by Contractor and all Work shall be performed for such sums and by such persons or entities that are acceptable to Owner and Engineer.

#### B. Cash Allowances:

- 1. Contractor agrees that:
  - cost to Contractor for materials and equipment required by allowances to be delivered at the Site and all applicable taxes are included in the cash allowances;
     and
  - b. Contract Price includes Contractor's costs for unloading and handling on the Site, labor, installation, overhead, profit and other expenses contemplated for the cash allowances and not in the allowances, and Contractor's demand or request for additional payment of account such will be invalid.

# C. Contingency Allowance:

1. A contingency allowance is for the sole use of Owner to cover unanticipated costs and is agreed upon by Contractor.

D. A Change Order will be issued by Engineer that will reflect actual amounts owed to Contractor for the performance of Work covered by allowance prior to final payment and Contract Price will be adjusted accordingly.

# 11.03 Unit Price Work

- A. The Contract Price will be an amount equal to the sum of the unit price, for each item shown in Unit Price Work, times the approximate quantity of each item as shown in the Agreement wherever the Contract Documents show that the Work, either all or part, is to be Unit Price Work.
- B. The quantities of items of Unit Price Work are estimates and are for the sole purpose of comparisons of Bids and establishing an initial Contract Price. Engineer will determine the actual quantities and classifications of Unit Price Work that Contractor performs in accordance with Paragraph 9.07.
- C. Each unit price will be in an amount that Contractor considers being sufficient to cover overhead and profit for each identified item.
- D. A Claim for an adjustment in the Contract Price may be made in accordance with Paragraph 10.05 if:
  - 1. a difference in the material or a significant difference from the estimated quantity, shown in the Agreement, occurs for any item of Unit Price Work performed by Contractor,
  - 2. no corresponding adjustment to any other item of Work occurs; and
  - 3. Contractor believes the Contract Price should be increased to account for additional expenses sustained by Contractor or Owner believes a decrease in Contract Price is justifiable and an amount of increase or decrease is unable to be agreed upon.

# ARTICLE 12 – CHANGE OF CONTRACT PRICE; CHANGE OF CONTRACT TIMES

# 12.01 Change of Contract Price

- A. A Change Order is required for a change to the Contract Price. The party making Claim must submit written notice to Engineer for any Claim for an adjustment in the Contract Price in accordance with Paragraph 10.05.
- B. For any Work covered by a Change Order or any Claim for an adjusted Contract Price, the value will be determined as follows:

051.18.2.22

1. where unit prices for such Work are shown in the Contract Documents, by application of such unit prices to the quantities of the items involved in accordance with Paragraph 11.03; or

- 2. where unit prices for such Work are not shown in the Contract Documents, by a sum agreed upon by both parties; or
- 3. where unit prices for such Work are not shown in the Contract Documents and an agreement between parties cannot be reached under Paragraph 12.01.B.2, based on the Cost of the Work (determined in accordance with Paragraph 11.01) plus a Contractor's fee for overhead and profit (determined in accordance with Paragraph 12.01.C).
- C. Contractor's Fee: The fee for overhead and profit for Contractor shall be determined as follows:
  - 1. a mutually acceptable fixed fee; or
  - 2. in the instance an agreement on a fixed fee cannot be reached, then a fee established by the following percentages of different portions of the Cost of the Work:
    - a. Contractor's fee shall be fifteen percent for costs sustained under Paragraphs 11.01.A.1 and 11.01.A.2;
    - b. Contractor's fee shall be five percent for costs sustained under Paragraph 11.01.A.3;
    - c. where one or more tiers of subcontracts are based on Cost of the Work plus a fee and an agreement on a fixed fee cannot be reached, Paragraphs 12.01.C.2.a and 12.01.C.2.b intend for the Subcontractor who essentially performs the work to be paid a fee of fifteen percent of costs sustained by said Subcontractor in accordance with Paragraph 11.01.A.1 and 11.01.A.2 and that any Subcontractor or Contractor on a higher tier will each receive a fee of five percent of the amount paid to the next lower tier Subcontractor.
    - d. no fee shall be owed on the basis of costs itemized under Paragraphs 11.01.A.4, 11.01.A.5 and 11.01.B;
    - e. the quantity of credit permitted by Contractor to Owner for any change that results in a net decrease in cost will be the amount of the actual net decrease in cost plus a deduction in Contractor's fee by an amount equal to five percent of such net decrease; and
    - f. the modification in Contractor's fee shall be calculated on the basis of the net change in accordance with Paragraph 12.01.C.2.a through 12.01.C.2.e whenever additions and credits are associated with one change.

# 12.02 Change of Contract Times

A. A Change Order is required to change the Contract Times. The party making the Claim must submit written notice to Engineer for any Claim for an adjustment in the Contract Times in accordance with Paragraph 10.05.

B. The provisions of Article 12 will be used for the determination of any adjustment of the Contract Times covered by a Change Order or any Claim for an adjustment in the Contract Times.

# 12.03 Delays

- A. Contract Times will be extended any amount equal to the time lost for any Work that Contractor is prevented from completing because of a delay beyond the control of Contractor provided a Claim is made as provided in Paragraph 12.02.A. Delays that are considered outside of the control of Contractor include but are not limited to acts or neglect by Owner or utility owners or other contractors performing other work as described by Article 7, fires, floods, epidemics, abnormal weather conditions or acts of God.
- B. Contractor shall be entitled to a reasonable adjustment in the Contract Price or Contract Times if delays, disruptions or interference with the performance of the Work occur due to Owner or other contractors or utility owners performing other work for Owner as described in Article 7. An adjustment of the Contract Times is based on said adjustment being critical to Contractor's ability to execute the Work within the Contract Times.
- C. If the performance of the Work by Contractor is delayed due to fire, flood, epidemic, abnormal weather conditions, acts of God, acts or failures to act of utility owners not under the control of Owner, or other causes that are not the responsibility of and beyond control of Owner and Contractor, then a reasonable adjustment in Contract Times will be permitted to Contractor but only if such adjustment is required for the completion of the Work within the Contract Times. Said adjustment shall be the sole and exclusive remedy for delays, described in this Paragraph 12.03.C, sustained by Contractor.
- D. Any claims, costs, losses or damages sustained by Contractor on or in connection with any project or expected project will not be the liability of Owner, Engineer or Owner's officers, directors, members, partners, employees, agents, consultants or subcontractors.
- E. For delays within the control of Contractor, no adjustment in Contract Price or Contract Times will be entitled to Contractor. Delays caused to and within the control of a Subcontractor or Supplier shall be considered delays within the control of Contractor.

# <u>ARTICLE 13 – TESTS AND INSPECTIONS; CORRECTION, REMOVAL OR ACCEPTANCE OF</u> DEFECTIVE WORK

13.01 Notice of Defects

A. Engineer will promptly notify Contractor of all defective Work that Owner or Engineer has actual knowledge of. Defective Work may be rejected, corrected or accepted as provided in this Article 13.

051.18.2.22

13.02 Access to Work

A. The Site and the Work will be accessible at reasonable times for observation, inspection and testing to Owner, Engineer, Owner's representatives and personnel, independent testing laboratories and governmental agencies with jurisdictional interests. It is the responsibility of the Contractor to advise them of Contractor's safety procedures and programs and provide them proper and safe conditions at the Site.

# 13.03 Tests and Inspections

- A. Engineer shall be notified in a timely manner of the readiness of the Work for all required inspections, tests or approvals. In order to facilitate required inspections or tests, Contractor shall cooperate with all inspection and testing personnel.
- B. Owner shall be responsible for employing and paying for the services of an independent testing laboratory to perform all inspections, tests or approvals required by the Contract Documents except;
  - 1. for inspections, tests or approvals covered by Paragraphs 13.03.C and 13.03.D below;
  - 2. that costs incurred related to tests or inspections conducted with regards to Paragraph 13.04.B shall be paid in accordance to Paragraph 13.04.C; and
  - 3. as otherwise specifically provided in the Contract Documents.
- C. Contractor is obligated for arranging and obtaining inspections, tests or approvals and pay all cost in connection with any Laws or Regulations, with jurisdiction, that require any Work to be specifically inspected, tested or approved by an employee or representative of such public body. All such required certificates of inspection or approval shall be delivered to Engineer by Contractor.
- D. It is the responsibility of Contractor to arrange, obtain and pay all costs associated with inspections, tests or approvals required for Owner's and Engineer's approval of materials or equipment for incorporation into the Work; or acceptance of materials, mix designs or equipment submitted for approval prior to purchase by Contractor for incorporation into the Work. Owner and Engineer must approve of organization that performs such inspections, tests or approvals.
- E. Contractor shall uncover any Work, at Engineer's request, that is to be inspected, tested or approved that has been covered by Contractor without written agreement of Engineer.
- F. Contractor shall bear all expenses related to uncovering Work as provided in Paragraph 13.03.E except if Engineer has been notified by Contractor of the intent to cover Work and Engineer did not promptly respond to such notice.

# 13.04 Uncovering Work

A. Contractor, at own expense, shall uncover any Work that was covered in disregard to written request of Engineer, if Engineer requests such Work to be uncovered.

- B. If it is deemed necessary or advisable by Engineer for covered Work to be observed by Engineer or inspected or tested by others, Contractor, at request of Engineer, shall uncover, expose or make available for observation, inspection or testing as required by Engineer the area of Work in question. Labor, material and equipment for the uncovering of such Work shall be provided by Contractor.
- C. Contractor shall pay all claims, costs, losses and damages occurring as a result of uncovering, exposure, observation, inspection, testing and satisfactory replacement or reconstruction of any uncovered Work that is found to be defective and a suitable decrease in the Contract Price shall be permitted for Owner.
- D. Contractor shall be entitled to an increase in the Contract Price or an extension of the Contract Times, or both, for any uncovering, exposure, observation, inspection, testing, replacement and reconstruction resulting from uncovering Work that is not found to be defective.

## 13.05 Engineer May Stop the Work

A. Engineer may direct Contractor to cease the Work, if the Work is discovered to be defective due to failure to provide adequately skilled workers or appropriate materials or equipment or the Work was performed in a way that results in the completed Work being incompatible with the Contract Documents, until Contractor eliminates the cause for such order. The right of Engineer to cease the Work shall not require Engineer to implement this right for the benefit of Contractor, Subcontractor, Supplier, other individual or entity or any employee or agent of any of them.

# 13.06 Correction or Removal of Defective Work

- A. Contractor shall correct all defective Work, promptly after receipt of written notice, whether or not fabricated, installed or completed; if Engineer rejects any of the Work, Contractor shall remove it from the Project and replace it with Work that is acceptable. All claims, costs, losses and damages that are consequential to the correction or removal of defective Work shall be paid for by Contractor.
- B. Contractor shall not act in a way that would void or otherwise impair Owner's special warranty and guarantee while correcting defective Work as required by this Paragraph 13.06 or Paragraph 13.07.

#### 13.07 Correction Period

- A. If any Work or any repair of damages to land or areas made accessible for Contractor's use by Owner or permitted by Laws and Regulations, as anticipated in Paragraph 6.11.A is found to be defective within one year after the date of Substantial Completion or by any specific provision of the Contract Documents, without cost to Owner and in agreement with Owner's written instructions, Contractor shall:
  - 1. restore such defective land or areas; or

- 2. correct such defective Work; or
- 3. remove any defective Work that has been rejected by Owner and replace it with Work that is not defective; and
- 4. correct, repair or remove and replace any damage to other Work, to the work of others or other land or areas damaged as a result.
- B. Owner may have defective Work corrected or repaired or have rejected Work removed and replaced if Contractor fails to punctually fulfill the terms of Owner's written instructions or in the case of an emergency where any deferral would cause risk of loss or damage. Contractor shall pay all claims, costs, losses and damages that result from such correction or repair or removal and replacement.
- C. The correction period for any specific item of equipment that is put into continuous service prior to Substantial Completion of all the Work may begin to run from an earlier date if provided for in the Specifications.
- D. The correction period for the correction or removal and replacement, under this Paragraph 13.07, of defective Work shall be lengthened for an additional one year starting at a time after such correction or removal and replacement has been satisfactorily completed.
- E. The requirements of Contractor under this Paragraph 13.07 are in addition to any other requirement or warranty. The provisions of this Paragraph 13.07 shall not be interpreted as a substitute for or a waiver of the provisions of any relevant statute of limitation or repose.

#### 13.08 Acceptance of Defective Work

A. Owner may accept defective Work rather than require correction or removal and replacement. It is the responsibility of Contractor to pay all claims, costs, losses and damages attributable to Owner's and Engineer's assessment of and decision to accept such defective Work and for the reduced value of the Work to the extent not otherwise paid by Contractor pursuant to this sentence. If Owner accepts defective Work, the required revisions in the Contract Documents will be completed with the issuance of a Change Order, and Owner shall be eligible for an appropriate reduction in the Contract Price that accounts for the reduced value of the accepted Work.

#### 13.09 Owner May Correct Defective Work

- A. If defective Work is neither corrected nor removed and replaced by Contractor within a reasonable time after written notice from Engineer, or if performance of the Work by Contractor is not in accordance with the Contract Documents, or if compliance with any provision of the Contract Documents is not made by Contractor, Owner has the right to correct or remedy any such deficiency after seven days written notice to Contractor.
- B. Owner shall proceed expeditiously with the rights and remedies provided by this Paragraph

- 13.09. Owner may dismiss Contractor from the Site, take control of all or part of the Work and discontinue Contractor's performance of the Work, take possession of Contractor's tools, appliances, construction equipment and machinery at the Site and incorporate into the Work all materials and equipment stored at the site that Owner has paid Contractor but which are stockpiled in a different place, in conjunction with corrective or remedial action. Owner, Engineer, Owner's and Engineer's representatives, agents and employees and Owner's other contractors shall be permitted admission to the Site by Contractor to exercise the rights and remedies provided by this Paragraph.
- C. Contractor will be charged with all claims, costs, losses and damages incurred or sustained by Owner in the exercise of the rights and remedies provided by this Paragraph 13.09 and the required revisions in the Contract Documents will be completed with the issuance of a Change Order; and Owner shall be eligible for an appropriate reduction in the Contract Price. Such claims, costs, losses and damages will comprise of but will not be limited to all costs associated with repair or replacement of work of others that is destroyed or damaged by correction, removal or replacement of Contractor's defective Work.
- D. An extension of the Contract Times will not be granted to Contractor for any delay in the performance of the Work that results due to the exercise by Owner of Owner's rights and remedies under this Paragraph 13.09.

# ARTICLE 14 – PAYMENTS TO CONTRACTOR AND COMPLETION

14.01 Schedule of Values

A. The basis for progress payments is established by the Schedule of Values, as provided in Paragraph 2.07.A, and will be integrated into a form of Application for Payment that is satisfactory to Engineer.

# 14.02 Progress Payments

- A. *Applications for Payments:* 
  - 1. An Application for Payment shall be submitted to Engineer for evaluation at a minimum of twenty days prior to date established in the Agreement for each progress payment. The Application for Payment shall be signed and completed by Contractor and shall account for all accomplished Work as of the date of the Application and the documentation required by the Contract Documents shall accompany the Application. If Contractor request payment for materials and equipment that have not yet been integrated into the Work but have been transported and stockpiled at the Site or another location, a bill of sale, invoice or other document that demonstrates materials and equipment have been received by Owner free and clear of all Liens and evidence that the stored materials and equipment have the applicable insurance coverage or other arrangement to protect Owner's interest, all of which must be acceptable to Owner, shall supplement the Application for Payment.
  - 2. Each Application, beginning with the second Application for Payment, shall contain an

affidavit of Contractor that states that all preceding progress payments have been used on account to release Contractor's responsibilities connected with preceding Applications for Payment.

3. The Agreement will establish the extent of retainage with respect to progress payments.

# B. Review of Applications:

- 1. Engineer shall specify in writing either acceptance of payment or return Application to Contractor indicating Engineer's justification for refusal of payment within ten days after receipt of each Application for Payment. Contractor may make required corrections and resubmit Application if Engineer refuses original Application.
- 2. Engineer's acceptance of any payment requested in an Application for Payment will be founded on Engineer's inspection of the completed Work and on Engineer's evaluation of the Application for Payment and the associated data and schedules, that to the best of Engineer's knowledge, information and belief:
  - a. the Work has advanced to the point specified;
  - b. the quality of the Work conforms to the requirements of the Contract Documents (conditional to an assessment of the Work as an operational whole preceding or upon Substantial Completion, the results of any ensuing tests required by the Contract Documents, a final judgment of quantities and classifications for Unit Price Work under Paragraph 9.07); and
  - c. Contractor has satisfied all circumstances required for the Work that precede payment of Contractor
- 3. Engineer's evaluation of Contractor's Work for the determination of payment will not constitute responsibility on Engineer for:
  - a. supervision, direction or control of the Work; or
  - b. the means, methods, techniques, sequences or procedures of construction or the safety precautions and programs required for the performance of the Work; or
  - c. failure of Contractor to abide by Laws and Regulations pertinent to Contractor's performance of the Work; or
  - d. examination to determine how or for what purposes Contractor has used the moneys paid on account of the Contract Price; or
  - e. determination that title to any of the Work, materials or equipment has progressed to Owner free and clear of any Liens.
- 4. Engineer may reject the whole or any part of any payment due to subsequently

discovered evidence or as a result of subsequent inspections or tests to protect Owner from loss because:

- a. the Work is defective or previously completed Work has been damaged and requires correction or replacement;
- b. Change Orders have reduced the Contract Price;
- c. Owner has corrected defective Work or finished Work in accordance with Paragraph 13.09; or
- d. Owner has knowledge of the occurrence of any of the events shown in Paragraph 15.02.A.

## C. Payment Becomes Due:

1. The amount settled on in the Application for Payment will be owed thirty days after the agreement on the amount due for that progress period and will be paid by Owner to Contractor.

# D. Reduction in Payment:

- 1. Owner may withhold full payment as a result of:
  - a. claims made against Owner as a result of Contractor's performance or furnishing of the Work:
  - b. Liens filed in relation to the Work, not including when Contractor has provided a bond to specifically secure the satisfaction and discharge of such Liens;
  - c. other items warranting Owner to a set-off against the proposed amount; or
  - d. Owner's knowledge of the existence of any of the events described in Paragraph 15.02.A.
- Owner will promptly provide Contractor written notice for any refusal to make payment of the full or partial amount in the Application for Payment. Written notice will outline the reasons for nonpayment. After Contractor corrects the reasons outlined in such written notice, Owner shall promptly compensate Contractor the amount withheld or any adjustment agreed to by Owner and Contractor.
- 3. If Owner's refusal of payment is discovered to be unwarranted, the amount unjustly withheld shall be treated in accordance with Paragraph 14.02.C.1 and subject to interest as provided in the Agreement.

14.03 Contractor's Warranty of Title

A. Contractor warrants and guarantees that title to all Work, materials and equipment included in any Application for Payment, whether or not integrated into the Project, will pass to Owner free and clear of all Liens prior to the time of payment.

# 14.04 Substantial Completion

- A. Contractor shall advise Engineer in writing when Contractor deems the entire Work ready for its proposed use (excluding items specifically noted by Contractor as unfinished) and request Engineer release a certificate of Substantial Completion.
- B. Owner, Engineer and Contractor shall perform an inspection of the Work to determine the status of completion soon after Contractor's request for a certificate of Substantial Completion. Engineer will advise Contractor in writing of any reasons Engineer believes is preventing the Work from being substantially complete.
- C. Engineer shall execute and deliver a definitive certificate of Substantial Completion, fixing the date of Substantial Completion, when Engineer deems the Work substantially complete. Engineer shall attach a tentative list of items to be completed or corrected prior to final payment.
- D. At the time of delivery of the tentative certificate of Substantial Completion, Engineer will deliver to Owner and Contractor a written recommendation as to division of responsibilities, pending final payment, between Owner and Contractor with respect to security, operation, safety, and protection of the Work, maintenance, heat, utilities, insurance, and warranties and guarantees.
- E. After the date of Substantial Completion, Owner shall have the right to dismiss Contractor from the Site with the exception of allowing Contractor reasonable admission to remove Contractor's property and complete or correct items on the tentative list.

#### 14.05 Partial Utilization

- A. Owner may use or occupy any substantially completed part of the Work which has explicitly been identified in the Contract Documents or which Engineer and Contractor agree is an independently operational and usable part of the Work. Contractor's performance of the Work must not be inhibited due to Owner's use or occupancy of such Work. Owner's use or occupancy of such Work is subject to the following conditions:
  - 1. If in the judgment of Engineer any part of the Work is ready for its proposed use and substantially complete, Engineer may request in writing for Contractor to allow Owner to use or occupy such Work. The procedures of Paragraph 14.04.A through E will be followed by Owner and Contractor if and when Contractor agrees that such part of the Work is substantially complete.
  - 2. Contractor may request Engineer to issue a certificate of Substantial Completion at any time Contractor believers any such part of the Work is ready for its intended use and substantially complete by informing Owner and Engineer in writing of such conditions.

- 3. Owner, Engineer and Contractor shall make an inspection of that part of the Work to determine its status of completion within a reasonable amount of time after such request. Engineer will notify Contractor in writing if Owner or Engineer does not consider that part of the Work substantially complete and reasoning will be provided in the written notification. The provisions of Paragraph 14.04 will pertain with respect to certification of Substantial Completion of that part of the Work if Engineer deems that part of the Work to be substantially complete.
- 4. Use or occupancy or separate operation of part of the Work shall not take place preceding compliance with the requirements of Paragraph 5.10 regarding property insurance.

# 14.06 Final Inspection

A. Owner, Engineer and Contractor will make a final inspection following written notice from Contractor that the entire Work or an agreed upon portion is complete. Contractor will be furnished with notification in writing from Engineer of all specifics revealed during the inspection signifying the Work is incomplete or defective. Incomplete or defective Work shall promptly be remedied by Contractor.

# 14.07 Final Payment

# A. Application for Payment:

- After the amendments identified in the final inspection have been rectified and completed, in the opinion of Engineer, by Contractor and Contractor has delivered, as required by Contract Documents, all maintenance and operating instructions, schedules, guarantees, bonds, certificates or other evidence of insurance, certificates of inspection, marked-up record documents and other documents, an application for final payment may be made by Contractor.
- 2. Contractor shall include with the final Application for Payment (except as formerly delivered):
  - a. all documentation required by the Contract Documents, along with but not limited to the proof of insurance required by Paragraph 5.05.B.6;
  - b. approval of the surety, if any, to final payment;
  - c. a list of all Claims, unresolved in the opinion of Contractor, against Owner; and
  - d. complete and legally effective releases or waivers (acceptable to Owner) of all Lien rights arising out of or Liens filed in association with the Work.
- 3. In lieu of the releases or waivers of Liens identified in Paragraph 14.07.A.2 and as permitted by Owner, Contractor may provide receipts or releases in full and an affidavit of Contractor that: (i) the releases and receipts containing all labor, services, material and equipment for which a Lien could be filed; and (ii) all payrolls, material and equipment

bills and other indebtedness related to the Work that could become the obligation of Owner, or which could result in liens or other burdens on Owner's property, have been paid or otherwise fulfilled. Contractor may furnish a bond or other collateral that is acceptable to Owner to indemnify Owner against any Lien that may result due to any Subcontractor or Supplier's failure to provide such a release or receipt in full.

# B. Engineer's Review of Application and Acceptance:

1. If Owner and Engineer are satisfied that the Work has been finished and Contractor's other responsibilities under the Contract Documents have been satisfied, based on observations by Engineer of the Work throughout construction and final inspection and Engineer's examination of the final Application for Payment and complementary documentation in accordance with the Contract Documents, Engineer, within ten days after receiving final Application for Payment, will provide Contractor with written notice that the Work is acceptable in accordance with Paragraph 14.09. Otherwise, Engineer will send the Application for Payment to Contractor and specify in writing Engineer's reason for rejection of final Application for Payment. In this case Contractor shall resubmit the Application for Payment after fulfilling the required amendments.

# C. Payment Becomes Due:

1. The amount owed Contractor, less any amount Owner is entitled to, will become due and will be paid by Owner to Contractor thirty days after the Owner is given the final Application for Payment and associated documentation.

# 14.08 Final Completion Delayed

A. If final fulfillment of the Work is considerably delayed and Contractor is not responsible for the delay, Owner shall make payment of the balance due for the portion of Work fully completed and accepted upon receipt of Contractor's final Application for Payment without terminating the Contract. Contractor shall submit the written consent of the surety for the payment of the balance due for the portion of the Work completed and accepted to Owner with the Application for such payment if the remaining balance retained by Owner for unfinished or erroneous Work is less than the retainage specified in the Agreement. Such payment shall be made by Owner in accordance with the terms and conditions governing final payment except that it shall not constitute a waiver of Claims.

# 14.09 Waiver of Claims

- A. The making and acceptance of final payment will constitute:
  - 1. Owner waiving all Claims against Contractor, other than all Claims resulting from unsettled Liens, due to defective Work emerging after final inspection in accordance with Paragraph 14.06, due to failure in compliance with the Contract Documents or any specified terms of any special guarantees included in Contract Documents or due to remaining requirements of Contractor under the Contract Documents; and

2. Contractor waiving all Claims against Owner except any previously made Claims conforming with the requirements herein and specifically recognized in writing by Owner as still being unresolved.

# <u>ARTICLE 15 – SUSPENSION OF WORK AND TERMINATION</u>

## 15.01 Owner May Suspend Work

A. Owner may interrupt the Work or any section of the Work, at any time and without reason, by informing Contractor in writing. Owner will establish the date Contractor may proceed with the Work and suspension of Work may not exceed ninety successive days. An adjustment in the Contract Price or an extension of the Contract Times shall be allowed to Contractor if a Claim is made by Contractor in accordance with Paragraph 10.05.

## 15.02 Owner May Terminate for Cause

- A. If any one or more of the following occurrences transpire, termination for cause will be justified:
  - 1. continuous failure by Contractor in the performance of the Work as required by the Contract Documents (including, but not limited to, failure to furnish adequately skilled workers or appropriate materials or equipment or failure to comply to the Progress Schedule established under Paragraph 2.07 as modified from time to time in accordance with Paragraph 6.04);
  - 2. disregard by Contractor of Laws or Regulations of any public body with jurisdiction;
  - 3. continual disregard by Contractor of the authority of Engineer; or
  - 4. violation by Contractor of any requirements of the Contract Documents in any significant way.
- B. If one or more of the occurrences described in Paragraph 15.02.A transpire, after giving Contractor seven days written notice of its intent to discontinue the services of Contractor, Owner may:
  - 1. without liability to Contractor for trespass or conversion, dismiss Contractor from the Site and seize possession of the Work and Contractor's tools, appliances, construction equipment and machinery at the Site and use to the full amount that they may possibly be used by Contractor.
  - 2. integrate into the Work all materials and equipment stockpiled at the site or for which Owner has compensated Contractor but may be stocked in another place; and
  - 3. execute the Work as Owner may consider appropriate.
- C. Contractor shall not be eligible to accept any additional payment until the Work is concluded if Owner proceeds as specified in Paragraph 15.02.B. If the unpaid balance of the Contract Price

exceeds all claims, costs, losses and damages sustained by Owner resulting from finishing the Work, such excess will be paid to Contractor. If such claims, costs, losses and damages exceed the unpaid balance, Contractor shall pay the difference to Owner. Owner shall not be obligated to acquire the lowest price for the performance of the Work when utilizing any rights or remedies under this Paragraph.

- D. Despite Paragraphs 15.02.B and 15.02.C, if Contractor commences, within seven days of receipt of notice of intent to terminate, to remedy Contractor's failure to execute and progresses diligently to rectify such failure within thirty days of receipt of said notice, Contractor's services will not be terminated.
- E. The termination of Contractor's services by Owner will not affect any rights or remedies existing or which may later accumulate of Owner against Contractor. Liability of Contractor will not be released as a consequence of any retention or payment of moneys due to Contractor by Owner.
- F. If and to the degree that Contractor has provided a performance bond under the provisions of Paragraph 5.01.A, the termination process of that bond shall override the provisions of Paragraphs 15.02.B and 15.02.C.

#### 15.03 Owner May Terminate For Convenience

- A. Owner may, without cause and prejudice to any other right or remedy of Owner, terminate the Contract after seven days written notice to Contractor. In such an event, Contractor shall be paid for:
  - 1. Work that was accomplished and accepted preceding the effective date of termination and executed as required by Contract Documents, including fair and equitable sums for overhead and profit on such Work;
  - costs incurred preceding the effective date of termination in the performance of services and supplying labor, materials, or equipment in accordance with the Contract Documents related to unfinished Work, plus fair and reasonable sums for overhead and profit on such expenses;
  - 3. all claims, costs, losses, and damages sustained during the settlement of terminated contracts with Subcontractors, Suppliers, and others; and
  - 4. equitable expenses directly resulting from the termination.
- B. Projected profits or revenue or other economic loss subsequent to such termination shall not be paid to Contractor.

#### 15.04 Contractor May Stop Work or Terminate

A. If, through no act or fault of Contractor, (i) the Work is postponed by Owner or under an order of court or other public authority for more than ninety consecutive days, or (ii) Owner fails to respond to any Application for Payment within thirty days after submission, or (iii) Owner fails

for thirty days to pay Contractor any sum determined to be due, then Contractor may, after seven days written notice to Owner and if Owner does not provide a solution to such failure within such times, terminate the Contract and recoup payment from Owner on the terms provided in Paragraph 15.03.

B. Instead of terminating the Contract and without prejudice to any other right or remedy, if Owner has failed to respond to Application for Payment within thirty days of submission or failed to pay Contractor a sum that has been determined to be due within thirty days, Contractor may, seven days after written notice to Owner, cease all Work pending all payments, including any accrued interest, owed to Contractor by Owner. The intent of Paragraph 15.04 is not to prevent Contractor from making a Claim under Paragraph 10.05 for an adjustment in Contract Price or Contract Times or any expenses or damages consequential to Contractor's discontinuance of Work as allowed by this Paragraph.

# **ARTICLE 16 – DISPUTE RESOLUTION**

#### 16.01 Methods and Procedures

- A. Mediation of any Claim made under Paragraph 10.05 may be requested by Owner or Contractor preceding a final and binding verdict. Such mediation will be overseen by the Construction Industry Mediation Rules of the American Arbitration Association in effect as of the Effective Date of the Agreement. The request for mediation shall be submitted in writing to the American Arbitration Association and the other party to the Contract. Timely submission of the request shall stay the effect of Paragraph 10.05.E.
- B. Owner and Contractor shall engage in the mediation process in good faith. The process shall be completed by sixty days of the filing of the request. The date of termination of the mediation shall be decided by application of the mediation rules indicated above.
- C. If the Claim is not resolved by mediation, Engineer's action under Paragraph 10.05.C or a denial pursuant to Paragraphs 10.05.C.3 or 10.05.D shall become final and binding 30 days after termination of the mediation unless, within that time period, Owner or Contractor:
  - 1. elect in writing to invoke any dispute resolution process provided for in the Supplementary Conditions; or
  - 2. agree with the other party to submit the Claim to another dispute resolution process; or
  - 3. gives written notice to the other party of the intent to submit the Claim to a court of competent jurisdiction.

# **ARTICLE 17 – MISCELLANEOUS**

#### 17.01 Giving Notice

A. Whenever written notice is required by the Contract Documents, it will be considered to have been properly executed if it is:

- 1. given in person to either the individual, member of the firm or an officer of the corporation for whom it is intended; or
- 2. delivered to or sent by registered or certified mail, postage prepaid to the latest business address known to the giver of the notice.

## 17.02 Computation of Times

A. Whenever any period of time is expressed in days in the Contract Documents, it will be figured by excluding the first and including the last day of such period. If the last day of such period falls on a Saturday, Sunday or legal holiday, said day will be excluded from the calculation.

#### 17.03 Cumulative Remedies

A. The duties and obligations established by the General Conditions and the rights and remedies existing in the General Conditions to the parties are supplementary to and are not to be interpreted in any way as a limitation of, any rights and remedies applicable to any or all of them which are established or available by Laws or Regulations, by special warranty or guarantee, or by other provisions of the Contract Documents. The provisions of this Paragraph will be as effective as if repeated specifically in the Contract Documents in connection with each particular duty, obligation, right and remedy to which they pertain.

# 17.04 Survival of Obligations

A. All representations, indemnifications, warranties, and guarantees made in, mandated by, or given in accordance with the Contract Documents, as well as all ongoing responsibilities specified in the Contract Documents, will continue beyond final payment, completion, and acceptance of the Work or termination or completion of the Contract or termination of the services of Contractor.

# 17.05 Controlling Law

A. This Contract is to be governed by the State of Georgia.

## 17.06 Headings

A. Article and paragraph headings are inserted for convenience only and do not constitute parts of these General Conditions.

#### **END OF SECTION 00700**

# **SECTION 00800 - SUPPLEMENTARY CONDITIONS**

# **INDEX**

PART 1:	TERMS	2
PART 2:	SCOPE OF THE WORK	2
PART 3:	CONSTRUCTION DRAWINGS	2
PART 4:	COMMENCEMENT AND COMPLETION OF WORK	
PART 5:	SUBSTITUTIONS OR "OR EQUAL"	2 2 3
PART 6:	REPORTS AND DRAWINGS USED BY THE ENGINEER	
PART 7:	INSURANCE	3 5
PART 8:	CERTIFICATES OF INSURANCE	5
PART 9:	BUILDER'S RISK	5
PART 10:	ACCIDENTS	5
PART 11:	HOLD HARMLESS CLAUSE	6
PART 12:	CONTRACTOR'S STATUS	6
PART 13:	CONTRACTOR'S AFFIDAVIT	6
PART 14:	RESIDENT PROJECT ENGINEER	6
PART 15:	ACCESS FOR INSPECTION	6
PART 16:	UTILITIES	6
PART 17:	ADJUSTMENT OF DISCREPANCIES	7
PART 18:	MEASUREMENT AND PAYMENT	7
PART 19:	MODIFICATION OF QUANTITIES	7
PART 20:	SAFETY AND HEALTH REGULATIONS	7
PART 21:	SANITARY CONVENIENCES	7
PART 22:	UTILITY SERVICES	7
PART 23:	ENVIRONMENTAL IMPACT	8
PART 24:	CONSTRUCTION STAKEOUT	8
PART 25:	INSTRUCTION OF OWNER'S EMPLOYEES	9
PART 26:	RECORD DATA AND DRAWINGS	9
PART 27:	PROPERTY CORNERS	9
PART 28:	RESTORATION	10
PART 29:	MAINTENANCE DURING CONSTRUCTION	10
PART 30:	BARRICADES, DANGER, WARNING AND DETOUR SIGNS	10
	HIGH VOLTAGE ACT	11
PART 32:	BUY AMERICAN	11

#### PART 1: TERMS

1.1 The terms used in this Supplementary Conditions have been defined in the General Conditions and will be interpreted according to the definition given to them in the General Conditions.

## PART 2: SCOPE OF THE WORK

2.1 See Section 00100 and 012000 for descriptions of the scope of work.

Intersection Improvements to construct new roundabout at the intersection of White Road and Peavy Road. in Byron, GA. Includes utility relocation's, etc.

#### PART 3: CONSTRUCTION DRAWINGS

3.1 The Work shall conform to the following Construction Drawings:

City of Byron: White Road at Peavy Road Intersection Improvements Cover and Sheets C-1 thru CK-10

# PART 4: COMMENCEMENTS AND COMPLETION OF WORK

- 4.1 Contractor shall begin work within ten (10) days once the Notice to Proceed has been issued. Contractor's work shall be finished within **240 consecutive calendar days**.
- 4.2 If Contractor is unable to execute the work with enough persistence to insure each portion of the work is accomplished within the time shown on the above schedule, in addition to any extensions created in conformance with Article 12 of the General Conditions; and, if Owner does not implement his reservations as provided for in Article 13, Contractor shall proceed with the work in which case liquidated damages for the delay will be unattainable. If applicable, liquidated damages will be required of the Contractor in the amount of \$500 per each day of delay of the work until the work is completed.

# PART 5: SUBSTITUTIONS OR "OR EQUAL"

- 5.1 Substitutes for the major equipment manufacturers shown in the Technical Specifications may be used. Submittals on "Or Equal" substitutes must be provided, within forty-eight (48) hours after the Bid Opening, to Engineer.
  - A. Submittals for every proposed type of equipment shall contain illustrative drawings; specifications, descriptive brochures, installation lists, weights, metal thickness of principal components; drive arrangements; torques; power requirements; performance curves; installation requirements; availability of spare parts; local service capability and other items essential for Engineer to conclude if the required Specifications will be satisfied.
  - B. The price bid of the substitute equipment shall not be used in the calculation of the base bid but

shall be bid as a deductible or additive alternate and shown on the forms supplied in the Proposal.

- C. A five (5) year warranty shall be mandatory on all substitute equipment/materials/systems in the form of a bond or other equivalent surety, except when allowed by the Engineer. Such warranty/surety shall be equal to the total amount of the Contractor's purchase order and shall account for installation and service for the substitute. Owner can demand adjustment or replacement if the substitute fails to operate adequately in regards to mechanical integrity or performance. Owner can also apply any or all of the bond/surety to adjust or replace the system or segments of the system, including alterations to another process, if the supplier/manufacturer fails to rectify the defects/performance.
- D. The action and performance of the substitute supplier/manufacturer shall be the liability of Contractor pending acceptable performance and afterward during the one (1) year warranty period as stated in the Contract Documents. After this, to cover the remaining period of two (2) years, Owner can take action on the system supplier's bond/surety. The bond/surety must be tended preceding acceptance of the substitute.
- E. Engineer reserves the right to accept or reject any and all substitutes suggested by Contractor.
- F. Award of the contract will be based on the lowest qualified base bid including any substitutes chosen by Engineer.
- G. The cost of all re-design and cost of all structural, mechanical and electrical changes shall be contained in the adjustment of the price bid for the substitute equipment if Engineer establishes that the considered item will not conform to the design.

# PART 6: REPORTS AND DRAWINGS USED BY THE ENGINEER

- 6.1 Engineer has utilized the following for the development of the Drawings and Specifications:
  - A. The subsequent reports of explorations and tests of sub-surface conditions at the site of the Work:
    - 1. N/A
  - B. The subsequent drawings of physical conditions in or pertaining to existing surface and subsurface structures, excluding Underground Facilities, which are adjacent to the site of the Work:
    - 1. N/A

# PART 7: INSURANCE

7.1 The limits of liability for the insurance required by Paragraph 5.04 of the General Conditions shall supply coverage at minimum equal to the following amounts or exceeding where mandated by Laws and Regulations:

A. Workers' Compensation, etc. under Paragraphs 5.04.A.1 and 5.04.A.2 of the General Conditions:

1. State: Statutory

2. Applicable Federal: Statutory

3. Employer's Liability: \$200,000

- 4. Owner shall be shown by Contractor as additionally insured.
- B. Comprehensive General Liability (under Paragraphs 5.04.A.3 through 5.04.A.5 of the General Conditions):
  - 1. Combined single limit for Bodily Injury and Property Damage:

\$1,000,000 Each Occurrence Or combined single limit \$2,000,000

General Aggregate \$2,000,000 Operations Aggregate \$2,000,000

- 2. Property Damage liability insurance will furnish Explosion, Collapse and Underground coverages where relevant.
- 3. Personal Injury, with employment exclusion deleted

\$1,000,000 Annual Aggregate \$1,000,000 Each Occurrence

4. Excess/Umbrella Policy

General Aggregate \$2,000,000 Each Occurrence \$1,000,000

C. Comprehensive Automobile Liability (under Paragraph 5.04.A.6 of the General Conditions):

Bodily Injury:

\$1,000,000 Each Person \$1,000,000 Each Occurrence

Property Damage:

\$100,000 Each Occurrence

Or combined single limit of \$200,000

Or a Combined Single Limit Per Accident \$2,000,000

For Property Damage and Bodily Injury of

7.2 Builders Risk Insurance (Installation Coverage):

100% completed value based on the insurable part of the project.

# 7.3 Contractual Endorsement:

- A. In accordance with Paragraph 5.04.B.3 of the General Conditions, the Contractual Liability shall furnish coverage equal to or greater than the following amounts:
  - 1. Bodily Injury:

\$500,000

Each Occurrence

2. Property Damage: \$100,000

\$N/A

Each Occurrence Annual Aggregate

# PART 8: CERTIFICATES OF INSURANCE

- 8.1 Certificates satisfactory to Owner shall accompany the signed Contract Documents when they are transmitted to the Owner for execution. The following statement shall be included with such certificates: "Coverages afforded under the policies will not be cancelled unless written notice has been given to the Owner at a minimum of thirty days preceding cancellation, as evidenced by receipts of registered or certified mail".
- 8.2 Owner shall be listed as Additional Insured on the Contractor's General Liability Policy and Builder's Risk in equivalent amounts required by the Contractor in addition to the requirements of Paragraph 5.3 to the General Conditions.

## PART 9: BUILDER'S RISK

9.1 Builder's Risk Insurance shall be obtained and sustained through the life of the Contract Agreement by Contractor to protect the interests of the Owner, Contractor and Sub-Contractors against loss by fire, vandalism, malicious mischief and all hazards contained in a standard "All Risk" Coverage endorsement. The amount of such insurance shall at all times be equivalent to or exceeding the full amount of the Contract. The policies shall be in the name of the Owner and Contractor.

# PART 10: ACCIDENTS

10.1 Contractor shall furnish, at the site, medical equipment and facilities that will adequately provide first-aid service to anyone who could become injured in association with the work. All accidents resulting in death, personal injury or property damage resulting from or in connection with the performance of the Work shall be disclosed to Owner and Engineer in writing, including full details and statement of witnesses, by Contractor whether accident occurs on or adjacent to the site. Additionally, if death or serious injuries or damages result from accident, the accident shall be immediately reported by telephone or messenger to Contractor and any subcontractor on account of any accident. Contractor shall promptly report the full details of the facts in writing to Owner and Engineer. Contractor shall inform his superintendent, foreman, or whoever is present at the site of the work the name and phone

number of the hospital and the name and phone number of the doctor he intends to use in the event of an accident.

# PART 11: HOLD HARMLESS CLAUSE

11.1 Contractor agrees to hold harmless, indemnify and defend Owner and Owner's agents, architects, engineers and employees from and against any and all claims, losses, damages, demands, causes of action and any and all associated costs and expenses, of every kind and character, emerging out of, incidental to, or resulting directly or indirectly from the Contractor's performance of the work described in this contract, regardless if such loss, damage, injury or liability is furthered by the negligence of the Owner, Engineer, its agents, architects, engineers or employees, except that Contractor shall have no liability for damages or the costs incidental to such caused by the sole negligence of the Owner, Engineer, its agents, architects, engineers or employees. Any and all Subcontractors will be obligated by Contractor to comply with the requirements of this clause preceding any work and agrees to guarantee that this clause is in conformity with the insurance provisions of the contract.

#### PART 12: CONTRACTOR'S STATUS

12.1 It is established that Contractor shall maintain the status of an Independent Contractor and the employees of Contractor are not employees of the Owner.

#### PART 13: CONTRACTOR'S AFFIDAVIT

13.1 After finishing the work and preceding final payment and settlement of all sums due in accordance with the contract, Contractor will submit to Owner, a Contractor's Affidavit in the typical form submitted by Contractor under the laws of the State of Georgia declaring that all bills for labor, materials and services in conjunction with said contract have been paid in full, recognizing receipt of the contract price and affirming that there are not unsettled claims under said contract which could become a lien on the real estate emerging out of said contract.

# PART 14: RESIDENT PROJECT ENGINEER

14.1 If considered essential, Owner upholds the right to furnish a Resident Project Engineer to insure the quality control of the Project and the conformance of the Work to the Plans and Specifications. Resident Project Engineer will act as Owner's Representative on the Project and will have the authority of the Owner as set forth in the Contract Documents.

#### PART 15: ACCESS FOR INSPECTION

15.1 Contract shall provide access for representatives of the Georgia Department of Natural Resources, Environmental Protection Division and the Georgia Department of Highways and Public Transportation for inspection of the site.

#### PART 16: UTILITIES

16.1 Utilities such as sewer, water and electric lines encountered during the work shall be preserved from

damage and retained in service pending the relocation or replacement as mandated by this Contract or by others as the case may be, or abandoned as might be essential for the appropriate construction and use of the new work.

#### PART 17: ADJUSTMENT OF DISCREPANCIES

17.1 In any instance of inconsistencies among the dimensions and details presented by drawings or between the drawings and these specifications, the more costly construction shall be estimated preceding construction and the issue shall be presented to the Engineer for explanation. Without such a judgment, the inconsistencies shall be modified by the Contractor at his own risk and Contractor shall bear all additional costs incurred due to the resolution of complications from such modification.

#### PART 18: MEASUREMENT AND PAYMENT

18.1 Measurement and payment shall be rendered for the units presented on the Bid Schedule and at the lump sum contract prices shown on the Bid Schedule. Direct payment shall only be rendered for those items or work specifically shown in the proposal. Cost incurred due to any additional work must be encompassed in the contract price for the appropriate items to which it is associated.

# PART 19: MODIFICATION OF QUANTITIES

- 19.1 The itemized quantities shall be deemed the quantities essential for the accomplishment of the Work by Contractor for the determination of the bid. In any instance that the quantities necessary for the construction of the work are greater or less than the quantities displayed on the items, an amount equal to the difference in quantities will be added or deducted from the contract price at the unit prices for such item.
- 19.2 If itemized quantities are not identified in the Proposal, the work displayed on the plans or indicated shall be believed by the Contractor to be incorporated in his contract for the lump sum price bid.

#### PART 20: SAFETY AND HEALTH REGULATIONS

20.1 Contractor shall conform with the Department of Labor Safety and Health Regulations for Construction declared under the Occupational Safety and Health Act of 1970 (PL 91-596) and under Section 107 of the contract Work and Safety Standards Act (PL 91-54). The regulations are controlled by the Department of Labor and admission to the project by Department of Labor Safety and Health shall be permitted by Contractor.

#### PART 21: SANITARY CONVENIENCES

21.1 Satisfactory sanitary conveniences shall be furnished by Contractor for use by anyone employed for the work and their use shall be rigorously enforced. Such accommodations shall be accessible at the time the first employees reach the site and shall be withdrawn following the exit of the last employees from the site.

051.18.2.22

## PART 22: UTILITY SERVICES

22.1 Water and electrical service will be determined by and at the expense of the Contractor.

# PART 23: ENVIRONMENTAL IMPACT

- 23.1 All activities by the Contractor shall be organized to reduce, to the utmost degree achievable, any unfavorable environmental impact.
  - A. Noise: All equipment and machinery shall be equipped with exhaust mufflers that are maintained in good working order to result in minimized operational noise.
  - B. Dust/Smoke: All equipment activities shall be characterized by a minimum of dust. Traveled surfaces and earthwork shall be kept in a moist condition to minimize the creation of dust or the airborne migration of particulate material under all prevalent atmospheric conditions.
  - C. Traffic: Trucks shall be moved along roads in such a way that minimizes the influence on traffic and annoyance to the public. All material shall be loaded in a way which will prevent the loss of any part of the load during transport.
  - D. Sedimentation: All locations of intense runoff due to rainfall shall be visually examined to verify that no material eroded from the construction site is being transferred offsite. If such transfer is transpiring, actions, including the establishment of detention basins, shall rapidly be employed to abolish such transfer. Soil Erosion and sediment control efforts shall contain all temporary and permanent techniques of protection and trapping soils of the construction site throughout land disturbing actions. Activity incorporated in this contract is exempt from the Georgia Erosion and Sediment Control Act, and NPDES General Permit for Construction Activity.

# PART 24: CONSTRUCTION STAKEOUT

- 24.1 Construction stakeout shall be the obligation of the Contractor.
- 24.2 Contractor shall execute the layout of the work by following the dimensions and benchmarks presented on the plans and shall be accountable for all measurements that could be essential during the performance of the work prescribed in the specifications or on the Drawings, conditional to such adjustments which might be necessary to accommodate altered conditions or resulting from required revisions to the Work.
- 24.3 All stakes, spikes, steel pins, templates, platforms, equipment, instruments, tools and materials and all labor mandatory for laying out any portion of the Work from baselines and benchmarks shall be supplied by and at the expense of the Contractor.
- 24.4 Contractor shall be accountable for maintaining and preserving all benchmarks displayed on the plans.
- 24.5 All survey records shall be documented in conformance with standard and approved methods. All field notes, sketches, records and computations generated by the Contractor in the lay out of the work shall be accessible for analysis by the Engineer or his duly approved representative at all times throughout the execution of the work.

- 24.6 Such surveys and computations shall be performed by Contractor as needed for the determination of the amount of work accomplished or placed throughout each period for which a progress payment is to be prepared.
- 24.7 The Engineer is authorized to perform checks as the work proceeds to verify lines and grades identified by the Contractor and to confirm the completed work is in compliance with the specifications of the Contract Documents and Drawings. Such examination by the Engineer or Engineer's representative shall not dismiss the obligation of the Contractor to complete all work in conformance with the Contract Documents and Drawings and the lines and grades specified within them. If the location marks identified by the Contractor are confirmed to be inaccurate or inadequate, work shall be postponed until amendments have been made to the erroneous location marks.
- 24.8 No distinct payment will be made to Contractor for costs associated with the survey work, layout work or staking. All such costs will be deemed incidental to the Work.

# PART 25: INSTRUCTION OF OWNER'S EMPLOYEES

- 25.1 Contractor shall furnish capable personnel who thoroughly comprehend the operation of the equipment to inform the Owner's employees of the operation and maintenance of each item and system. Such training shall occur at a time or times suitable to Owner and preceding acceptance of the installation by the Owner. The Contractor's bid price for this Contract shall contain cost incurred due to this training. Training shall be of the on-the-job type and shall contain all operation and equipment maintenance.
- 25.2 Arrangements for the training of Owner's employees will be jointly established by Owner and Contractor.

# PART 26: RECORD DATA AND DRAWINGS

- 26.1 Contractor shall maintain precise, understandable documentation of the locations, types and sizes of sanitary lines, service laterals, manholes, cleanouts, water lines, fittings, valves, hydrants, drainage pipes, drainage structures and other related work completed during the execution of this project. On a set of project prints furnished by the Owner, Contractor shall develop a set of "record" drawings displaying the data specified above. The horizontal positions of all portions of items used on this project shall be precisely linked to features that are physical and visible, including property corner markers and/or permanent structures. Invert elevations of all manholes, storm sewers and structures, sanitary sewers and lift stations shall be visibly shown. These "as-built" drawings shall be maintained clean and dry and sustained in an up-to-date state throughout the performance of the work. Such drawings shall be available to Owner or Engineer, within twenty-four (24) hours following their request, at any time during the performance of the work.
- 26.2 Contractor shall furnish Owner with one (1) set of red-lined prints showing record drawing information.

#### PART 27: PROPERTY CORNERS

27.1 Any property corners or monuments disturbed by Contractor during the performance of the work shall

be restored by a professional surveyor registered in the State of Georgia at the expense of the Contractor.

## **PART 28: RESTORATION**

- 28.1 Contractor shall manage his operations in a way that repairs roadways, driveways, curb and gutter, ditches and easements as the pipe laying advances.
- 28.2 Reasonable care shall be taken by Contractor throughout the performance of the work to prevent damage to vegetation. When applicable, ornamental shrubbery and tree branches shall be restrained in such a way to decrease possible damage. Trees sustaining damage to branches shall have damaged branches pruned in such a way to enhance the appearance of the tree. Tree trunks sustaining damage during the performance of the work shall have tree dressing administered to the damaged areas.

#### PART 29: MAINTENANCE DURING CONSTRUCTION

- 29.1 Work shall be maintained by Contractor from the start of construction operations until final acceptance. Such maintenance shall incorporate continuous and effective work conducted day by day with satisfactory equipment and require that the site and structures at the project are retained in acceptable condition at all times, in addition to acceptable signing or marking as applicable and control of traffic wherever necessary by use of traffic control devices as mandated by the State of Georgia.
- 29.2 Contractor shall extract all construction signs and barriers after the conclusion of the Work.
- 29.3 Contractor shall keep roads accessible to traffic throughout the performance of the Work. Contractor shall maintain sections of the site accessed by public traffic in such a way to accommodate traffic. All cost incurred due to mandatory signs and markings and other maintenance work throughout construction and before the Work is accepted will be the responsibility of the Contractor. Contractor shall also endure the cost associated with maintenance of approaches, crossings, intersections and other features as may be needed.

# PART 30: BARRICADES, DANGER, WARNING AND DETOUR SIGNS

- 30.1 All required barricades, lighting, danger signals, signs and other traffic control devices shall be provided, erected and maintained by Contractor. Contractor shall also take all applicable provisions to insure the protection of the work and the safety of the public. All highways and streets restricted from traffic shall be protected by adequate barricades and all obstructions shall be lighted at any time of darkness. Contractor shall furnish appropriate warning signs for the control and direction of traffic.
- 30.2 All mandatory barricades, warning signs and other protection devices shall be furnished, installed and maintained, in accordance with the State requirements, by Contractor. Temporary signs can be reused only if such signs are in acceptable condition and legible. All protective devices shall be maintained in good, legible condition throughout their use.
- 30.3 Temporary barricades and signs shall be removed as soon as the construction progresses to the point that such signs are no longer necessary.
- 30.4 The cost of furnishing, erecting, maintaining and removing protective devices will not be paid for as a

distinct Bid Item. The cost incurred by Contractor due to the performance of these functions shall be incorporated into the overall Bid. Contractor shall maintain ownership of all temporary warning devices.

## PART 31: HIGH VOLTAGE ACT

31.1 Contractor accepts the obligations of the High Voltage Act of the General Assembly of Georgia by execution of this Contract.

#### PART 32: BUY AMERICAN

32.1 By submitting this bid, Contractor guarantees that preference will be given to domestic construction material by the Contractor, sub-contractors, material men and suppliers during the performance of this Contract.

## **END OF SECTION 00800**

## SECTION 011000 SUMMARY

## **PART 1 - GENERAL**

#### 1.1 SUMMARY

#### A. Section Includes:

- 1. Contract description.
- 2. Work by Owner or other Work at the Site.
- 3. Owner-furnished products.
- 4. Contractor's use of Site and premises.
- 5. Future work.
- 6. Work sequence.
- 7. Owner occupancy.
- 8. Permits.
- 9. Specification conventions.

#### 1.2 CONTRACT DESCRIPTION

The work to be performed and provided under this Contract shall consist of furnishing all labor, materials, tools, equipment, safety equipment, all necessary traffic control, taxes, profit, overhead, and other costs and effort required for the Construction of a Roundabout and all necessary Utility relocations and Utility Construction as shown on the drawings. The time allotted for final completion of work is **240 consecutive calendar days.** 

- A. Perform Work of the Contract under fixed cost Contract with Owner according to Conditions of Contract.
- B. Work of the Contract is identified in the following:
  - 1. Contract Documents, Technical Specifications, and Plans

## 1.3 WORK BY OWNER OR OTHERS

- A. Owner will award a contract for installation of the Work beginning on receipt of Notice to Proceed.
- B. If Owner-awarded contracts interfere with each other due to work being performed at the same time or at the same Site, Owner will determine the sequence of work under all contracts according to "Work Sequence" and "Contractor's Use of Site" Articles in this Section.

051.18.2.22

C. Coordinate Work with utilities of Owner and public or private agencies.

## 1.4 OWNER-FURNISHED PRODUCTS

## A. Owner's Responsibilities:

- 1. Arrange for and deliver Owner-reviewed Shop Drawings, Product Data, and Samples to Contractor.
- 2. Upon delivery, inspect products jointly with Contractor.
- 3. Submit claims for transportation damage and replace damaged, defective, or deficient items

## B. Contractor's Responsibilities:

- 1. Review Owner-reviewed Shop Drawings, Product Data, and Samples.
- 2. Receive and unload products at Site; inspect for completeness or damage jointly with Owner.
- 3. Handle, store, install, and finish products.
- 4. Repair or replace items damaged after receipt.
- 5. Arrange for manufacturers' warranties, inspections, and service.

## 1.5 WORK SEQUENCE/SCHEDULE

A. The Contract time for this project to be complete and placed into service prior to the completion date listed in the Notice to Proceed.

#### 1.6 PARTIAL OWNER OCCUPANCY

- A. The Owner will occupy existing facilities to which these improvements are being made and will continue operation during the period of construction.
- B. There shall not be any interruption in service that hasn't been approved by the Engineer. The Contractor is responsible for fines assessed due to his activities.
- C. Whenever, in the opinion of the Engineer, any section or portion of the work or any structure is in suitable condition, it may be put into use upon the written order of the Engineer and such usage will not be held in any way as an acceptance of said work or structure, or any part thereof, or as a waiver of any of the provisions of these Specifications and the Contract. Pending final completion and acceptance of the Work, all necessary repairs and replacements, due to defective materials or workmanship or operations of the Contractor, for any section of the work so put into use shall be performed by the Contractor at Contractor's own expense.

# 1.7 **PERMITS** – (**NOT USED**)

## 1.8 SPECIFICATION CONVENTIONS

A. These Specifications are written in imperative mood and streamlined form. This imperative language is directed to Contractor unless specifically noted otherwise. The words "shall be" are included by inference where a colon (:) is used within sentences or phrases.

PART 2 - PRODUCTS - (NOT USED)

PART 3 - EXECUTION - (NOT USED)

**END OF SECTION 011000** 

# SECTION 012000 PRICE AND PAYMENT PROCEDURES

## **PART 1 - GENERAL**

#### 1. SECTION INCLUDES

- A. Cash allowances.
- B. Testing and inspection allowances.
- C. Schedule of Values.
- D. Application for Payment.
- E. Change procedures.
- F. Defect assessment.
- G. Unit prices.
- H. Alternates.

### 2. SCHEDULE OF VALUES

- A. Submit schedule for Engineer review. Contractor's standard form or electronic media printout will be considered for this use. Type schedule on 8-1/2 x 11-inch white paper.
- B. Identify schedule with:
  - 1. Title of Project and Location
  - 2. Engineer
  - 3. Name and address of Contractor
  - 4. Contract designation
  - 5. Date of Submission
- C. Submit Schedule of Values within 20 days after date of Owner-Contractor Agreement.
- D. Format: Each line item of the schedule on the bid schedule shall be used for the pay items in the schedule of values.
- E. Include in each line item amount of allowances as specified in this Section. For unit cost allowances, identify quantities taken from Contract Documents multiplied by unit cost to achieve total for each item.

051.18.2.22

F. Revise schedule to list approved Change Orders with each Application for Payment.

## 3. APPLICATION FOR PAYMENT

- A. Submit schedule on a form of Application for Payment that is satisfactory to the Engineer.
- B. Content and Format: Use Schedule of Values for listing items in Application for Payment.
- C. Submit updated construction schedule with each Application for Payment..
- D. Payment Period: Submit at intervals stipulated in the Agreement.
- E. Submit submittals with transmittal letter as specified in Section 013300 Submittal Procedures.
- F. Submit three copies of waivers requested by Owner.
- G. Substantiating Data: When Engineer requires substantiating information, submit data justifying dollar amounts in question. Include the following with Application for Payment:
  - 1. Current construction photographs specified in Section 013300 Submittal Procedures.
  - 2. Partial release of liens from major Subcontractors and vendors.
  - 3. Record Documents as specified in Section 017000 Execution and Closeout Requirement, for review by Owner, which will be returned to Contractor.
  - 4. Affidavits attesting to off-Site stored products.
  - 5. Construction Progress Schedule revised and current as specified in Section 013300 Submittal Procedures.
  - 6. Delay days from current pay period

## 4. CHANGE PROCEDURES

- A. Submittals: Submit name of individual who is authorized to receive change documents and is responsible for informing others in Contractor's employ or Subcontractors of changes to the Work.
- B. Carefully study and compare Contract Documents before proceeding with fabrication and installation of Work. Promptly advise Engineer of any error, inconsistency, omission, or apparent discrepancy.
- C. Requests for Interpretation (RFI) and Clarifications: Allot time in construction scheduling for liaison with Engineer; establish procedures for handling queries and clarifications.
  - 1. Submit Request for information on a form that is satisfactory to the Engineer.
  - 2. Engineer may respond with a direct answer on the Request for Interpretation form.
- D. Engineer will advise of minor changes in the Work not involving adjustment to Contract Sum/Price or Contract Time by issuing supplemental instructions.
- E. Engineer may issue Proposal Request including a detailed description of proposed change with supplementary or revised Drawings and Specifications, a change in Contract Time for executing the change with stipulation of overtime work required and with the period of time during which the requested price will be considered valid. Contractor will prepare and submit estimate within 5 days.

- F. Contractor may propose changes by submitting a request for change to Engineer, describing proposed change and its full effect on the Work. Include a statement describing reason for the change and the effect on Contract Sum/Price and Contract Time with full documentation and a statement describing effect on the Work by separate or other Contractors.
- G. Document requested substitutions according to Section 012500 Substitution Procedures.
- H. Stipulated Sum/Price Change Order: Based on Notice of Change and Contractor's fixed price quotation or Contractor's request for Change Order as approved by Engineer.
- I. Unit Price Change Order: For Contract unit prices and quantities, the Change Order will be executed on a fixed unit price basis. For unit costs or quantities of units of that which are not predetermined, execute Work under Work Directive Change. Changes in Contract Sum/Price or Contract Time will be computed as specified for Time and Material Change Order.
- J. Work Directive Change: Engineer may issue directive, signed by Owner, instructing Contractor to proceed with change in the Work, for subsequent inclusion in a Change Order. Document will describe changes in the Work and designate method of determining any change in Contract Sum/Price or Contract Time. Promptly execute change.
- K. Time and Material Change Order: Submit itemized account and supporting data after completion of change, within time limits indicated in Conditions of the Contract. Engineer will determine change allowable in Contract Sum/Price and Contract Time as provided in Contract Documents.
- L. Maintain detailed records of Work done on time and material basis. Provide full information required for evaluation of proposed changes and to substantiate costs for changes in the Work.
- M. Document each quotation for change in Project Cost or Time with sufficient data to allow evaluation of quotation.
- N. Change Order Forms: Form approved for use by Engineer.
- O. Execution of Change Orders: Engineer will issue Change Orders for signatures of parties as provided in Conditions of the Contract.
- P. Correlation of Contractor Submittals:
  - 1. Promptly revise Schedule of Values and Application for Payment forms to record each authorized Change Order as separate line item and adjust Contract Sum/Price.
  - 2. Promptly revise Progress Schedules to reflect change in Contract Time, revise subschedules to adjust times for other items of Work affected by the change, and resubmit.
  - 3. Promptly enter changes in Record Documents.

#### 5. **DEFECT ASSESSMENT**

A. Replace the Work, or portions of the Work, not conforming to specified requirements.

- B. If, in the opinion of Engineer, it is not practical to remove and replace the Work, Engineer will direct appropriate remedy or adjust payment.
- C. The defective Work may remain, but unit sum/price will be adjusted to new sum/price reduced 50 percent at discretion of Engineer.
- D. Defective Work will be partially repaired according to instructions of Engineer, and unit sum/price will be adjusted to new sum/price reduced 50 percent reduced at discretion of Engineer.
- E. Individual Specification Sections may modify these options or may identify specific formula or percentage sum/price reduction.
- F. Authority of Engineer to assess defects and identify payment adjustments is final.
- G. Nonpayment for Rejected Products: Payment will not be made for rejected products for any of the following reasons:
  - 1. Products wasted or disposed of in a manner that is not acceptable.
  - 2. Products determined as unacceptable before or after placement.
  - 3. Products not completely unloaded from transporting vehicle.
  - 4. Products placed beyond lines and levels of the required Work.
  - 5. Products remaining on hand after completion of the Work.
  - 6. Loading, hauling, and disposing of rejected products.

## 6. UNIT PRICES

- A. Authority: Measurement methods are delineated in individual Specification Sections.
- B. Measurement methods delineated in individual Specification Sections complement criteria of this Section. In event of conflict, requirements of individual Specification Section govern.
- C. Take measurements and compute quantities. Engineer will verify measurements and quantities.
- D. Unit Quantities: Quantities and measurements indicated on Bid Form are for Contract purposes only. Quantities and measurements supplied or placed in the Work shall determine payment. Actual quantities provided shall determine payment.
  - 1. When actual Work requires more or fewer quantities than those quantities indicated, provide required quantities at contracted unit sum/prices.
  - 2. When actual Work requires 25 percent or greater change in quantity than those quantities indicated, Owner or Contractor may claim a Contract Price adjustment.
- E. Payment Includes: Full compensation for required labor, products, tools, equipment, plant and facilities, transportation, services and incidentals; erection, application, or installation of item of the Work; overhead and profit.
- F. Final payment for Work governed by unit prices will be made on basis of actual measurements and quantities accepted by Engineer multiplied by unit sum/price for Work incorporated in or made necessary by the Work.

## G. Measurement of Quantities:

- 1. Weigh Scales: Inspected, tested, and certified by applicable State weights and measures department within past year.
- 2. Platform Scales: Of sufficient size and capacity to accommodate conveying vehicle.
- 3. Metering Devices: Inspected, tested, and certified by applicable State department within past year.
- 4. Measurement by Weight: Concrete reinforcing steel, rolled or formed steel, or other metal shapes will be measured by handbook weights. Welded assemblies will be measured by handbook or scale weight.
- 5. Measurement by Volume: Measured by cubic dimension using mean length, width, and height or thickness.
- 6. Measurement by Area: Measured by square dimension using mean length and width or radius.
- 7. Linear Measurement: Measured by linear dimension, at item centerline or mean chord.
- 8. Stipulated Sum/Price Measurement: Items measured by weight, volume, area, or linear means or combination, as appropriate, as completed item or unit of the Work.

Unit Price Schedule (table is for reference only):

	GDOT					
Item	Item	Description	Quantity	Unit	Unit Price	Total
	Code					
1.	150	Traffic Control	1	LS		
2.	151	Mobilization	1	LS		
3.	163	Haybale Check Dam (Cd-Hb)	36	EA		
4.	163	Tree Protection (Tr)	21	EΑ		
5.	163	Concrete Washdown	1	EA		
6.	163	Outlet Protection (St)	38	SY		
7.	163	Slope Stabilization (Ss)	2,600	SY		
8.	163	Silt Fence (Sd1-NS)	2,510	LF		
9.	163	Temporary Grassing and Mulching (Ds1 & Ds2)	2	AC		
10.	163	Permanent Grassing (Ds3)	2	AC		
11.	163	Construction Exit (Co)	5	EΑ		
12.	210	Grading Complete	1	LS		
13.	310	Graded Aggregate Base 10 inch Incl. Matl.	4,000	TN		
14.	402	Recycled 2 in. Asphalt Concrete 12.5mm Superpave, GP 2 only, Incl. Bi- tum. Matl. and H Lime (220 LBS/SY)	660	ΙΝ		
15.	402	Recycled 2 in. Asphalt Concrete 19mm Superpave GP 1 or 2, Incl. Bitum. Matl. and H Lime (220 LBS/SY)	650	TN		
16.	402	Recycled 4 in. Asphalt Concrete 25mm Superpave, GP 1 or 2, Incl. Bitum Matl. And H Lime (440 LBS/SY)	1,300	TN		
17.	429	Rumble Strips 20 ft Long	12	EA		

			1			
18.	441	Concrete Integral Median, Type 7, 6 inch	600	SY		
19.	441	Concrete Truck Apron Stamped	350	SY		
20.	441	Concrete Flume	14	EA		
21.	441	Concrete Header Curb, Type 2, 6 inch	250	LF		
22.	441	Concrete Header Curb, Type 7, 6 inch	190	LF		
23.	441	Concrete Header Curb, Type 9, 4 inch	300	LF		
24.	441	Concrete Curb and Gutter, Type 2, 30 inch	1,700	LF		
25.	444	Sawed Joints	100	LF		
26.	550	15 inch Storm Drain Pipe, H1-10	24	LF		
27.	550	18 inch Storm Drain Pipe, H1-10	40	LF		
28.	550	15 inch Flared End Section	2	EA		
29.	550	18 inch Flared End Section	2	EA		
30.	600	Controlled Low Strength Flowable Fill	30	CY		
31.	636	Highway Signs and Posts	1	LS		
32.	652	Pavement Striping and Markings	1	LS		
33.	-	Remove Sanitary Sewer Pipe - 8 inch	50	LF		
34.	-	Remove Sanitary Sewer Manhole	1	EA		
35.	-	Sanitary Sewer Manhole	1	EA		
36.	-	Adjust Watervalve Box to Grade	1	EA		
37.	-	Water Main Pipe - 6 inch	150	LF		
38.	-	Water Main 45 Deg Bend - 6 inch	1	EA		
39.	-	Connect to Existing Water Main	2	EA		
40.	-	Water Main 22.5 Deg Bend - 6 inch	1	EA		
41.	-	PVC Carrier Pipes – 3 inch	200	LF		
42.	-	Water Valve - 6 inch	2	EA		
43.	-	Fire Hydrant - 6 inch	1	EA		
44.	-	Lighting and Electrical Instrumentation and Controls Installation Complete	1	LS		
TOTAL BID						

## 7. ALTERNATES-Not Used

#### PART 2 - PRODUCTS - Not Used

#### **PART 3 - SPECIAL PROVISIONS**

<u>01-PROJECT SPECIFICATIONS</u>: Unless otherwise specified, all work including methods of measurement and payment shall meet all requirements of the current Georgia Department of Transportation (GDOT) Standard Specification for the Construction of Roads and Bridges including all applicable supplemental additions and revisions thereto which by reference are hereby made a part of this agreement. In addition, any utility relocation as part of this project shall follow the City of Byron (City) Standard Specifications.

**<u>02-EXISTING UTILITIES</u>:** It shall be the responsibility of the Contractor to coordinate the relocation of utilities with his schedule. He shall also locate and protect all existing utilities during the work. The Contractor at no cost to the City shall repair any damages done during construction in accordance with the repair standards of the utility owner. Utilities as shown on the plans are for representation purposes only and no claim is made as to the accuracy of the utility locations as drawn. Call Before You Dig – (800) 282-7411, 72 hours advanced notice required! All City waterline and sewerline utilities in conflict with the proposed roadway improvements must be relocated and reestablished by the Contractor at no expense to the City. All work must be coordinated through the City inspection staff and operations department. A by-pass plan must be submitted to the City by the Contractor for review and approval prior to any relocation of sewerlines.

<u>03-SUBMITTALS</u>: All submittals shall be reviewed and approved by the Engineer prior to use. Specific submittals required, but not limited to, include traffic control plans, storm drain pipe, storm structures, filter fabric, concrete, asphalt mix designs, graded aggregate base, electrical controls and equipment, utility materials, and type II backfill. All utility submittals shall be provided to the Engineer for review and approval.

<u>04- CONTRACT TIME</u>: Time is of the essence and the Contractor shall diligently pursue the work to completion. Extraordinary delays beyond the control of the Contractor may be considered for time extension.

**<u>05-LAYOUT OF WORK:</u>** The Contractor shall be responsible for setting his own grades and make sure that his construction fits existing field conditions.

**<u>06-QUALITY CONTROL:</u>** All work shall be subject to inspection and testing. The Contractor shall provide all required testing, as determined by the Engineer or City.

<u>07-TRAFFIC CONTROL AND CONSTRUCTION SIGNAGE</u>: A traffic control plan shall be submitted and approved prior to any mobilization upon the project. This item shall include all the costs associated with providing construction signage and traffic control in accordance with the MUTCD, and GDOT requirements, as well as all other costs which are not covered by any of the other bid items. No night work shall be approved. Traffic is to be maintained with no more than one lane closed.

**08-EROSION & SEDIMENTATION CONTROL:** The Contractor shall use Best Management Practices (BMP's) throughout construction. The work covered by this item shall be in accordance with Section 163 and 171 of the GDOT Specifications except as noted in the special provisions. Payment for Erosion and Sedimentation Control bid items shall include all labor, equipment, and materials required to accomplish this work as shown on the plans as well as any additional measures that may be required, as directed by the Engineer. All disturbed areas shall be dressed and stabilized as required by these provisions and the GDOT Worksite Erosion Control Manual. The Contractor shall remove silt from temporary erosion control devices, as a minimum, when one half of the capacity, by height, depth, or volume has been reached. There is no direct payment for maintenance of erosion control devices and any anticipated cost shall be included in the bid price and shall be full payment for any and all resources required to adequately remove all silt necessary to re-establish full potential volume (0% filled). Any required testing, sampling, and fees related to the Notice of Intent (NOI) or Land Disturbance Permit (LDA) for NPDES permit purposes shall be provided by the Contractor and included in the No separate payment will be the iob. provided to the

**09-FOUNDATION BACKFILL MATERIAL TYPE II:** Foundation backfill material Type II as needed for wet conditions to provide a suitable foundation shall be used if authorized by the Engineer. A minimum of 12 inches undercut and Type II backfill shall be installed. Stone bedding materials placed under the storm sewer and structures shall be isolated from the surrounding subgrade by an approved filter fabric to prevent infiltration of the backfill into the type II bedding due to fluctuating water tables. Filter fabric shall be placed to the satisfaction of the Engineer to prevent both Horizontal and Vertical migration. Type II material shall not be paid for under this item to satisfy dewatering requirements. Payment for this item shall be included in the cost of the storm sewer construction.

**10-STORM SEWER CONSTRUCTION:** The pipe material for this project shall be reinforced concrete pipe, unless otherwise noted. Each pipe joint shall be wrapped with filter fabric. This filter fabric shall be a minimum of three (3) feet in width and shall be a product specifically designed for this purpose (4 -ounce non-woven, needle punched geotextile). The product must be pre-approved by the Engineer. Payment for wrapping the joints shall be incidental to the storm pipe item.

Any hand-built drainage structures shall receive a mortar coat on both the inside and outside walls. Concrete fillets shall be incorporated into all storm drain structures and all dewatering devices shall remain in place until the non-shrink grout used to secure the pipe to the drainage structure has had adequate time to set. If used, precast drainage structures shall be ordered to a top finished elevation of twelve (12) inches below proposed finish grade to provide adequate room for final field adjustment. Additionally, the City reserves the flexibility of hand built structures in regards to vertical, horizontal, previously unknown tie-ins or other adjustments as may be necessary during construction to meet required conditions. No additional compensation shall come due to modify precast structures to meet necessary field conditions.

11-GRADING COMPLETE: The work covered by this item shall be in accordance with Section 210, GDOT Standard Specifications unless separate pay items have been established to otherwise cover a particular part of the work. Unless specific payment items have been established in the bid tab items, this work shall also include, but is not limited to, all grading, all clearing and grubbing, tree removal, asphalt removal, existing structure removal, removal of excess materials, topsoiling, dust control, and borrow as may be necessary. In all areas, slopes shall be graded to achieve a minimal slope while still providing positive drainage to proposed drainage structures.

<u>12-FINISHING, DRESSING, AND TURF ESTABLISHMENT</u>: It shall be the responsibility of the Contractor to finish and dress all areas disturbed by construction. All loose concrete, granite stone, base materials, roots, construction debris, etc., shall be removed at the time of fine grading before in-

stallation of any permanent vegetation. All disturbed area shall be mulched to prevent erosion as soon as possible after completion of grading activities.

13-DEWATERING: This item shall be full compensation for equipment, labor, and materials required for dewatering during installation of culvert additions and storm sewers in this contract. The dewatering method shall be submitted to and approved by the City prior to any construction activities by the Contractor. Payment is on a lump sum basis and this price includes the work for all activities and shall cove the additional option if chosen.

**14-DRIVEWAY ACCESS:** The Contractor shall maintain driveway and personal property access at all times during the project. Suitable means of access will be determined by the Engineer. Payment for this item shall be incidental to the cost of the job.

**15-PAVEMENT STRIPING AND HIGHWAY SIGNAGE:** The Contractor shall install all pavement striping and highway signage in accordance with the project plans, or as directed by the Engineer. Any necessary labor, equipment, and materials needed to install the pavement striping and highway signage, including temporary striping, shall be incidental to the bid items.

**16-LIGHTING** AND ELECTRICAL INSTRUMENTATION AND CONTROLS INSTALLATION COMPLETE: The Contractor shall install all lighting and electrical instrumentation and controls in accordance with the project plans, or as directed by the Engineer. Any necessary labor, equipment, and materials needed to install all lighting and electrical instrumentation and controls, shall be incidental to the bid items.

<u>17-CONTROLLED LOW STRENGTH FLOWABLE FILL</u>: This item shall include all safety measures, excavation, backfill, flowable fill, brick, compaction, testing, materials, and all other incidental appurtenances required for a complete installation as required in accordance with the drawings and specifications. Payment shall be made at the Contract Price per CY installed.

**18-REMOVE SANITARY SEWER PIPE-8 INCH:** This item shall include all excavation, removal of existing piping, loading, hauling, and disposal costs. Payment shall be made at the contract unit price per linear foot for the removal and proper disposal of the total length of pipe.

**19-REMOVE SANITARY SEWER MAHOLE:** This item shall include all necessary labor, equipment, and materials for demolition of an existing sewer manhole as described. Specifically, this item shall include removal of the cone or top section, hauling, legal and lawful disposal, plugging of sewer mains, cracking of bottom, and filling the remainder of manhole structure with sand. Payment shall be made at the Contract unit price for each manhole that is abandoned in this manner.

**20-SANITARY SEWER MANHOLE:** This item shall include all excavation, dewatering, subgrade preparation, bedding, installation of base, barrels, cone, and manhole steps, pipe connections, invert construction, grouting, backfill, compaction, clean-up, testing, and all necessary labor, equipment, and materials for installation of complete 4' diameter structure as shown and described in the project plans and technical specifications. Payment shall be made at the Contract unit price per each for structures installed, tested, and placed into service.

**21-WATER MAIN PIPE:** 6" C-900, DR 18 PVC Water Main- This item shall include all excavation, dewatering, clean-up, hauling, pressure testing, disinfection, and all necessary labor, equipment, and materials for installation of the 8" PVC water main piping as shown and described in the project plans and technical specifications. Payment shall be made at the Contract unit price per linear foot installed as measured along the centerline of the pipeline.

**22-WATER MAIN 45 DEGREE BEND:** This item shall include all excavation, dewatering, preparation of foundation, compact fittings, polyethylene encasement, glands, gaskets, thrust blocking, disinfection, pressure testing, and all necessary labor, equipment, and materials required for installation of fittings as shown and described in the project plans and technical specifications. Payment shall be made at the Contract unit price per pound for the total weight of fittings installed based on compact fitting weight. Fittings installed for contractor's convenience will not be counted for payment.

23-WATER MAIN 22.5 DEGREE BEND: This item shall include all excavation, dewatering, preparation of foundation, compact fittings, polyethylene encasement, glands, gaskets, thrust blocking, disinfection, pressure testing, and all necessary labor, equipment, and materials required for installation of fittings as shown and described in the project plans and technical specifications. Payment shall be made at the Contract unit price per pound for the total weight of fittings installed based on compact fitting weight. Fittings installed for contractor's convenience will not be counted for payment.

**24-ADJUST WATER VALVE BOX TO GRADE:** This item shall include all excavation, dewatering, preparation of foundation, valve box extensions as required, and all necessary labor, equipment, and materials required for adjusting existing water valve box to grade as shown and described in the project plans and technical specifications. Payment shall be made at the Contract unit price for each valve box successfully adjusted to grade.

**25-CONNECT TO EXISTING WATER MAIN:** This item shall include all excavation, dewatering, location of existing pipeline, sleeves, glands, couplings, fittings and all necessary labor, equipment, and materials required for connection to an existing water main as shown and described in the project plans and technical specifications. Payment shall be made at the Contract unit price for each connection to an existing water main.

**26-WATER VALVE WITH BOX:** This item shall include all excavation, dewatering, preparation of foundation, valve box, concrete collar, stem extensions as required, gear actuators as required, disinfection, testing, and all necessary labor, equipment, and materials required for setting of 6" gate valve as shown and described in the project plans and technical specifications. Payment shall be made at the Contract unit price for each 6" gate valve successfully placed into service.

**27-FIRE HYDRANT:** This item shall include all excavation, dewatering, preparation of foundation, placement of stone, blocking, barrel extensions as required, disinfection, testing and all necessary labor, equipment, and materials for setting of fire hydrants as shown and described in the project plans and technical specifications. Payment shall be made at the Contract unit price for each fire hydrant successfully placed into service.

#### **PART 4-OWNERS ALLOWANCE:**

This item shall only be authorized for use at the written discretion of the City. Under no circumstance shall the Contractor assume any allowance is available for use in the project unless authorized by the City for unknowns associated with construction activities.

#### **END OF SECTION 012000**

# SECTION 013000 ADMINISTRATIVE REQUIREMENTS

#### **PART 1 - GENERAL**

#### 1.1 SECTION INCLUDES

- A. Coordination and Project conditions.
- B. Preconstruction meeting.
- C. Site mobilization meeting.
- D. Progress meetings.
- E. Preinstallation meetings.
- F. Closeout meeting.
- G. Alteration procedures.

#### 1.2 COORDINATION AND PROJECT CONDITIONS

- A. Coordinate scheduling, submittals, and Work of various Sections of Project Manual to ensure efficient and orderly sequence of installation of interdependent construction elements, with provisions for accommodating items installed later.
- B. Verify that utility requirements and characteristics of operating equipment are compatible with building utilities. Coordinate Work of various Sections having interdependent responsibilities for installing, connecting to, and placing operating equipment in service.
- C. Coordinate space requirements, supports, and installation of mechanical and electrical Work indicated diagrammatically on Drawings. Follow routing shown for pipes, ducts, and conduit as closely as practical; place runs parallel with lines of building. Use spaces efficiently to maximize accessibility for other installations, for maintenance, and for repairs.
  - 1. Coordination Drawings: Prepare as required to coordinate all portions of Work. Show relationship and integration of different construction elements that require coordination during fabrication or installation to fit in space provided or to function as intended. Indicate locations where space is limited for installation and access and where sequencing and coordination of installations are important.
- D. Coordination Meetings: In addition to other meetings specified in this Section, hold coordination meetings with personnel and Subcontractors to ensure coordination of Work.
- E. In finished areas except as otherwise indicated, conceal pipes, ducts, and wiring within construction. Coordinate locations of fixtures and outlets with finish elements.

- F. Coordinate completion and clean-up of Work of separate Sections in preparation for Substantial Completion and for portions of Work designated for Owner's partial occupancy and for portions of Work designated for Owner's occupancy.
- G. After Owner's occupancy of premises, coordinate access to Site for correction of defective Work and Work not complying with Contract Documents, to minimize disruption of Owner's activities.

## 1.3 PRECONSTRUCTION MEETING

- A. Engineer will schedule and preside over meeting after Notice of Award.
- B. Attendance Required: Engineer, Owner, major Subcontractors, and Contractor.
- C. Minimum Agenda:
  - 1. Execution of Owner-Contractor Agreement.
  - 2. Submission of executed bonds and insurance certificates.
  - 3. Distribution of Contract Documents.
  - 4. Submission of list of Subcontractors, list of products, schedule of values, and Progress Schedule.
  - 5. Designation of personnel representing parties in Contract.
  - 6. Communication procedures.
  - 7. Procedures and processing of requests for interpretations, field decisions, field orders, submittals, substitutions, Applications for Payments, proposal request, Change Orders, and Contract closeout procedures.
  - 8. Scheduling.
  - 9. Critical Work sequencing.
  - 10. Scheduling activities of Geotechnical Engineer.
- D. Engineer: Record minutes and distribute copies to participants within five days after meeting, to Contractor, Owner, and those affected by decisions made.

# 1.4 SITE MOBILIZATION MEETING

- A. Engineer will schedule and preside over meeting at Project Site prior to Contractor occupancy.
- B. Attendance Required: Engineer, Owner, Contractor, Contractor's superintendent, special consultants and major Subcontractors.

- C. Minimum Agenda:
  - 1. Use of premises by Owner and Contractor.
  - 2. Owner's requirements and partial occupancy.
  - 3. Construction facilities and controls provided by Owner.
  - 4. Temporary utilities provided by Owner.
  - 5. Survey and building layout.
  - 6. Security and housekeeping procedures.
  - 7. Schedules.

- 8. Procedures for testing.
- 9. Procedures for maintaining record documents.
- 10. Requirements for startup of equipment.
- 11. Inspection and acceptance of equipment put into service during construction period.
- D. Engineer: Record minutes and distribute copies to participants within five days after meeting, to Contractor, and those affected by decisions made.

#### 1.5 PROGRESS MEETINGS

- A. Schedule and administer meetings throughout progress of the Work at maximum bi-monthly intervals.
- B. Engineer will make arrangements for meetings, prepare agenda with copies for participants, and preside over meetings.
- C. Attendance Required: Job superintendent, major Subcontractors, Contractors and suppliers, and Engineer, Owner, as appropriate to agenda topics for each meeting.
- D. Minimum Agenda:
  - 1. Review minutes of previous meetings.
  - 2. Review of Work progress.
  - 3. Field observations, problems, and decisions.
  - 4. Identification of problems impeding planned progress.
  - 5. Review of submittal schedule and status of submittals.
  - 6. Review of off-Site fabrication and delivery schedules.
  - 7. Maintenance of Progress Schedule.
  - 8. Corrective measures to regain projected schedules.
  - 9. Planned progress during succeeding work period.
  - 10. Coordination of projected progress.
  - 11. Maintenance of quality and work standards.
  - 12. Effect of proposed changes on Progress Schedule and coordination.
  - 13. Other business relating to Work.
- E. Engineer: Record minutes and distribute copies to participants within five days after meeting, to Engineer, Owner, and those affected by decisions made.

## 1.6 PREINSTALLATION MEETINGS

- A. When required in individual Specification Sections, convene preinstallation meetings at Project Site before starting Work of specific Section.
- B. Require attendance of parties directly affecting, or affected by, Work of specific Section.
- C. Notify Engineer five days in advance of meeting date.
- D. Prepare agenda and preside over meeting:
  - 1. Review conditions of installation, preparation, and installation procedures.

- 2. Review coordination with related Work.
- E. Engineer: Record minutes and distribute copies to participants within five days after meeting, to Engineer, Owner, and those affected by decisions made.

#### 1.7 CLOSEOUT MEETING

- A. Schedule Project closeout meeting with sufficient time to prepare for requesting Substantial Completion. Preside over meeting and be responsible for minutes.
- B. Attendance Required: Contractor Construction Manager, major Contractors, major Subcontractors, Engineer, Owner, and others appropriate to agenda.
- C. Notify Engineer five days in advance of meeting date.
- D. Minimum Agenda:
  - 1. Start-up of facilities and systems.
  - 2. Operations and maintenance manuals.
  - 3. Testing, adjusting, and balancing.
  - 4. System demonstration and observation.
  - 5. Operation and maintenance instructions for Owner's personnel.
  - 6. Temporary indoor-air-quality plan and procedures.
  - 7. Contractor's inspection of Work.
  - 8. Contractor's preparation of an initial "punch list."
  - 9. Procedure to request Engineer inspection to determine date of Substantial Completion.
  - 10. Completion time for correcting deficiencies.
  - 11. Inspections by authorities having jurisdiction.
  - 12. Certificate of Occupancy and transfer of insurance responsibilities.
  - 13. Partial release of retainage.
  - 14. Final cleaning.
  - 15. Preparation for final inspection.
  - 16. Closeout Submittals:
    - a. Project record documents.
    - b. Operating and maintenance documents.
    - c. Operating and maintenance materials.
    - d. Affidavits.
  - 17. Final Application for Payment.
  - 18. Contractor's demobilization of Site.
  - 19. Maintenance.
- E. Engineer: Record minutes and distribute copies to participants within five days after meeting, to Engineer, Owner, and those affected by decisions made.

### PART 2 - PRODUCTS - (NOT USED)

#### **PART 3 - EXECUTION**

#### 3.1 ALTERATION PROCEDURES

- A. Designated areas of existing facilities will be occupied for normal operations during progress of construction. Cooperate with Owner in scheduling operations to minimize conflict and to permit continuous usage.
  - 1. Perform Work not to interfere with operations of occupied areas.
  - 2. Keep utility and service outages to a minimum and perform only after written approval of Owner.
  - 3. Clean Owner-occupied areas daily. Clean spillage, overspray, and heavy collection of dust in Owner-occupied areas immediately.
- B. Materials: As specified in product Sections; match existing products with new and salvaged products for patching and extending Work.
- C. Employ skilled and experienced installer to perform alteration and renovation Work.
- D. Cut, move, or remove items as necessary for access to alterations and renovation Work. Replace and restore at completion. Comply with Section 017000 Execution and Closeout Requirements
- E. Remove unsuitable material not marked for salvage, including rotted wood, corroded metals, and deteriorated masonry and concrete. Replace materials as specified for finished Work.
- F. Remove debris and abandoned items from area and from concealed spaces.
- G. Prepare surface and remove surface finishes to permit installation of new Work and finishes.
- H. Close openings in exterior surfaces to protect existing Work from weather and extremes of temperature and humidity.
- I. Remove, cut, and patch Work to minimize damage and to permit restoring products and finishes to original or specified condition.
- J. Refinish existing visible surfaces to remain in renovated rooms and spaces, to specified renewed condition for each material, with neat transition to adjacent finishes.
- K. Where new Work abuts or aligns with existing Work, provide smooth and even transition. Patch Work to match existing adjacent Work in texture and appearance.
- L. When finished surfaces are cut so that smooth transition with new Work is not possible, terminate existing surface along straight line at natural line of division and submit recommendation to Engineer for review.
- M. Where change of plane of 1/4 inch or more occurs, submit recommendation for providing smooth transition to Engineer for review request instructions from Engineer.

- N. Trim existing doors to clear new floor finish. Refinish trim to original or specified condition.
- O. Patch or replace portions of existing surfaces that are damaged, lifted, discolored, or showing other imperfections.
- P. Finish surfaces as specified in individual product Sections.

**END OF SECTION 013000** 

# SECTION 013216 CONSTRUCTION PROGRESS SCHEDULE

## **PART 1 - GENERAL**

#### 1.1 SECTION INCLUDES

- A. Submittals.
- B. Quality assurance.
- C. Format for network analysis schedules.
- D. Network analysis schedules.
- E. Bar chart schedules.
- F. Review and evaluation.
- G. Updating schedules.
- H. Distribution.

## 1.2 SUBMITTALS

- A. Within (10) calendar days after the Notice to Proceed, submit proposed preliminary network diagram defining planned operations for all project task.
- B. Participate in review of preliminary and complete network diagrams jointly with Engineer.
- C. Within (20) days after joint review of proposed preliminary network diagram, submit draft of proposed complete network diagram for review. Include written certification that major Subcontractors have reviewed and accepted proposed schedule.
- D. Within (10) days after joint review, submit complete network analysis consisting of network diagrams and mathematical analyses.
- E. Submit updated network schedules with each Application for Payment.
- F. Submit number of opaque reproductions Contractor requires, plus two copies Engineer will retain.
- G. Submit network schedules under transmittal letter form specified in Section 013300 Submittal Procedures.

- H. Schedule Updates:
  - 1. Overall percent complete, projected and actual.

- 2. Completion progress by listed activity and sub activity, to within five working days prior to submittal.
- 3. Changes in Work scope and activities modified since submittal.
- 4. Delays in submittals or resubmittals, deliveries, or Work.
- 5. Adjusted or modified sequences of Work.
- 6. Other identifiable changes.
- 7. Revised projections of progress and completion.

## I. Narrative Progress Report:

- 1. Submit with each submission of Progress Schedule.
- 2. Summary of Work completed during the past period between reports.
- 3. Work planned during the next period.
- 4. Explanation of differences between summary of Work completed and Work planned in previously submitted report.
- 5. Current and anticipated delaying factors and estimated impact on other activities and completion milestones.
- 6. Corrective action taken or proposed.

## 1.3 QUALITY ASSURANCE

- A. Scheduler: Contractor's personnel specializing in CPM scheduling with two years' minimum experience in scheduling construction work of complexity comparable to the Project, and having use of computer facilities capable of delivering detailed graphic printout within 48 hours of request.
- B. Contractor's Administrative Personnel: Two years' minimum experience in using and monitoring CPM schedules on comparable Projects.

#### 1.4 FORMAT FOR NETWORK ANALYSIS SCHEDULE

- A. Listings: Reading from left to right, in ascending order for each activity. Identify each activity with applicable Specification Section number.
- B. Diagram Sheet Size: 11 inches high x 17 inches wide required.
- C. Scale and Spacing: To allow for notations and revisions.

#### 1.5 NETWORK ANALYSIS SCHEDULES

- A. Prepare network analysis diagrams and supporting mathematical analyses using critical path method.
- B. Illustrate order and interdependence of activities and sequence of Work; how start of given activity depends on completion of preceding activities, and how completion of activity may restrain start of subsequent activities.
- C. Illustrate complete sequence of construction by activity, identifying Work of separate stages. Indicate dates for submittals and return of submittals; dates for procurement and delivery of

critical products; and dates for installation and provision for testing. Include legend for symbols and abbreviations used.

- D. Mathematical Analysis: Tabulate each activity of detailed network diagrams using calendar dates, and identify for each activity:
  - 1. Preceding and following event numbers.
  - 2. Activity description.
  - 3. Estimated duration of activity, in maximum 15-day intervals. Status of critical activities.
  - 4. Earliest start date.
  - 5. Earliest finish date.
  - 6. Actual start date.
  - 7. Actual finish date.
  - 8. Latest start date.
  - 9. Latest finish date.
  - 10. Total and free float; accrue float time to Owner and to Owner's benefit.
  - 11. Monetary value of activity, keyed to Schedule of Values.
  - 12. Percentage of activity completed.
  - 13. Responsibility.
- E. Analysis Program: Capable of compiling monetary value of completed and partially completed activities, of accepting revised completion dates, and of recomputing of scheduled dates and float.
- F. Required Sorts: List activities in sorts or groups:
  - 1. By preceding Work item or event number from lowest to highest.
  - 2. By longest float, then in order of early start.
  - 3. By responsibility in order of earliest possible start date.
  - 4. In order of latest allowable start dates.
  - 5. In order of latest allowable finish dates.
  - 6. Contractor's periodic payment request sorted by Schedule of Values list.
  - 7. List of basic input data-generating report.
  - 8. List of activities on critical path.
- G. Prepare subschedules for each stage of Work and Sequencing of Construction Plan identified in Section 011000 Summary.
- H. Coordinate contents with Schedule of Values in Section 013300 Submittal Procedures.

#### 1.6 BAR CHART SCHEDULES

- A. Format: Bar chart Schedule, to include at least:
  - 1. Identification and listing in chronological order of those activities reasonably required to complete the Work, including:
    - a. Subcontract Work.
    - b. Major equipment design, fabrication, factory testing, and delivery dates including required lead times.

- c. Move-in and other preliminary activities.
- d. Equipment and equipment system test and startup activities.
- e. Project closeout and cleanup.
- f. Work sequences, constraints, and milestones.
- 2. Listings identified by Specification Section number.
- 3. Identification of the following:
  - a. Horizontal time frame by year, month, and week.
  - b. Duration, early start, and completion for each activity and subactivity.
  - c. Critical activities and Project float.
  - d. Subschedules to further define critical portions of Work.

## 1.7 REVIEW AND EVALUATION

- A. Participate in joint review and evaluation of schedules with Engineer at each submittal.
- B. Evaluate Project status to determine Work behind schedule and Work ahead of schedule.
- C. After review, revise schedules incorporating results of review, and resubmit within 5 days.

## 1.8 UPDATING SCHEDULES

- A. Maintain schedules to record actual start and finish dates of completed activities.
- B. Indicate progress of each activity to date of revision, with projected completion date of each activity. Update schedules to depict current status of Work.
- C. Identify activities modified since previous submittal, major changes in Work, and other identifiable changes.
- D. Upon approval of a Change Order, include the change in the next schedule submittal.
- E. Indicate changes required to maintain Date of Substantial and Total Completion.
- F. Submit sorts as required to support recommended changes.
- G. Prepare narrative report to define problem areas, anticipated delays, and impact on schedule. Report corrective action taken or proposed and its effect including effects of changes on schedules of separate Contractors.

#### 1.9 DISTRIBUTION

- A. Following joint review, distribute copies of updated schedules to Contractor's Project site file, to Subcontractors, suppliers, Engineer, Owner, and other concerned parties.
- B. Instruct recipients to promptly report, in writing, problems anticipated by projections shown in schedules.

PART 2 - PRODUCTS - (NOT USED)

PART 3 - EXECUTION - (NOT USED)

**END OF SECTION 013216** 

# SECTION 013300 SUBMITTAL PROCEDURES

# PART 1 - GENERAL

# 1.1 SECTION INCLUDES

A.	Definitions.								
B.	Submittal procedures.								
C.	Construction progress schedules.								
D.	Proposed product list.								
E.	Product data.								
F.	Use of electronic CAD files of Project Drawings.								
G.	Shop Drawings.								
H.	Samples.								
I.	Other submittals.								
J.	Design data.								
K.	Test reports.								
L.	Certificates.								
M.	Manufacturer's instructions.								
N.	Manufacturer's field reports.								
O.	Erection Drawings.								
P.	Construction photographs.								
Q.	Contractor review.								
R.	Engineer review.								
1.2	DEFINITIONS								
A.	Action Submittals: Written and graphic information and physical samples that require								

051.18.2.22

Engineer's responsive action.

B. Informational Submittals: Written and graphic information and physical Samples that do not require Engineer's responsive action. Submittals may be rejected for not complying with requirements.

#### 1.3 SUBMITTAL PROCEDURES

- A. Transmit each submittal with Submittal Transmittal Engineer-accepted form.
- B. Sequentially number transmittal forms. Mark revised submittals with original number and sequential alphabetic suffix.
- C. Identify: Project, Contractor, Subcontractor and supplier, pertinent Drawing and detail number, and Specification Section number appropriate to submittal.
- D. Apply Contractor's stamp, signed or initialed, certifying that review, approval, verification of products required, field dimensions, adjacent construction Work, and coordination of information is according to requirements of the Work and Contract Documents.
- E. Schedule submittals to expedite Project, and deliver to Engineer at business address. Coordinate submission of related items.
- F. For each submittal for review, allow 15 days excluding delivery time to and from Contractor.
- G. Identify variations in Contract Documents and product or system limitations that may be detrimental to successful performance of completed Work.
- H. Allow space on submittals for Contractor and Engineer review stamps.
- I. When revised for resubmission, identify changes made since previous submission.
- J. Distribute copies of reviewed submittals as appropriate. Instruct parties to promptly report inability to comply with requirements.
- K. Submittals not requested will not be recognized nor processed.
- L. Incomplete Submittals: Engineer will not review. Complete submittals for each item are required. Delays resulting from incomplete submittals are not the responsibility of Engineer.

### 1.4 CONSTRUCTION PROGRESS SCHEDULES

A. Comply with Section 013216 - Construction Progress Schedule

## 1.5 PROPOSED PRODUCT LIST

- A. Within 15 days after date of Notice to Proceed, submit list of major products proposed for use, with name of manufacturer, trade name, and model number of each product.
- B. For products specified only by reference standards, indicate manufacturer, trade name, model or catalog designation, and reference standards.

## 1.6 PRODUCT DATA

- A. Product Data: Action Submittal: Submit to Engineer for review for assessing conformance with information given and design concept expressed in Contract Documents.
- B. Submit number of copies Contractor requires, plus four copies Engineer will retain.
- C. Mark each copy to identify applicable products, models, options, and other data. Supplement manufacturers' standard data to provide information specific to this Project.
- D. Indicate product utility and electrical characteristics, utility connection requirements, and location of utility outlets for service for functional equipment and appliances.
- E. After review, produce copies and distribute according to "Submittal Procedures" Article and for record documents described in Section 017000 Execution and Closeout Requirements.

#### 1.7 ELECTRONIC CAD FILES OF PROJECT DRAWINGS

- A. Electronic CAD Files of Project Drawings: May only be used to expedite production of Shop Drawings for the Project. Use for other Projects or purposes is not allowed.
- B. Electronic CAD Files of Project Drawings: Distributed only under the following conditions:
  - 1. Use of files is solely at receiver's risk. Engineer does not warrant accuracy of files. Receiving files in electronic form does not relieve receiver of responsibilities for measurements, dimensions, and quantities set forth in Contract Documents. In the event of ambiguity, discrepancy, or conflict between information on electronic media and that in Contract Documents, notify Engineer of discrepancy and use information in hard-copy Drawings and Specifications.
  - 2. CAD files do not necessarily represent the latest Contract Documents, existing conditions, and as-built conditions. Receiver is responsible for determining and complying with these conditions and for incorporating addenda and modifications.
  - 3. User is responsible for removing information not normally provided on Shop Drawings and removing references to Contract Documents. Shop Drawings submitted with information associated with other trades or with references to Contract Documents will not be reviewed and will be immediately returned.
  - 4. Receiver shall not hold Engineer responsible for data or file clean-up required to make files usable, nor for error or malfunction in translation, interpretation, or use of this electronic information.
  - 5. Receiver shall understand that even though Engineer has computer virus scanning software to detect presence of computer viruses, there is no guarantee that computer viruses are not present in files or in electronic media.
  - 6. Receiver shall not hold Engineer responsible for such viruses or their consequences, and shall hold Engineer harmless against costs, losses, or damage caused by presence of computer virus in files or media.

## 1.8 SHOP DRAWINGS

- A. Shop Drawings: Action Submittal: Submit to Engineer for assessing conformance with information given and design concept expressed in Contract Documents.
- B. Indicate special utility and electrical characteristics, utility connection requirements, and location of utility outlets for service for functional equipment and appliances.
- C. When required by individual Specification Sections, provide Shop Drawings signed and sealed by a professional Engineer responsible for designing components shown on Shop Drawings.
  - 1. Include signed and sealed calculations to support design.
  - 2. Submit Shop Drawings and calculations in form suitable for submission to and approval by authorities having jurisdiction.
  - 3. Make revisions and provide additional information when required by authorities having jurisdiction.
- D. Submit number of opaque reproductions Contractor requires, plus four copies Engineer will retain.
- E. After review, produce copies and distribute according to "Submittal Procedures" Article and for record documents described in Section 017000 Execution and Closeout Requirements.

#### 1.9 SAMPLES

- A. Samples: Action Submittal: Submit to Engineer for assessing conformance with information given and design concept expressed in Contract Documents.
- B. Samples for Selection as Specified in Product Sections:
  - 1. Submit to Engineer for aesthetic, color, and finish selection.
  - 2. Submit Samples of finishes, textures, and patterns for Engineer selection.
- C. Submit Samples to illustrate functional and aesthetic characteristics of products, with integral parts and attachment devices. Coordinate Sample submittals for interfacing work.
- D. Include identification on each Sample, with full Project information.
- E. Submit number of Samples specified in individual Specification Sections; Engineer will retain one Sample.
- F. Reviewed Samples that may be used in the Work are indicated in individual Specification Sections.
- G. Samples will not be used for testing purposes unless specifically stated in Specification Section.
- H. After review, produce copies and distribute according to "Submittal Procedures" Article and for record documents described in Section 017000 Execution and Closeout Requirements.

## 1.10 OTHER SUBMITTALS

- A. Closeout Submittals: Comply with Section 017000 Execution and Closeout Requirements.
- B. LEED Submittals: Comply with Section 018113 Sustainable Design Requirements. Permits: Within 15 days after date established in Notice to Proceed, submit a list of permits and licenses to be obtained, identifying the granting agency and the required date of permit submittal.

#### 1.11 DESIGN DATA

- A. Informational Submittal: Submit data for Engineer's knowledge as Contract administrator or for Owner.
- B. Submit information for assessing conformance with information given and design concept expressed in Contract Documents.

#### 1.12 TEST REPORTS

- A. Informational Submittal: Submit reports for Engineer's knowledge as Contract administrator or for Owner.
- B. Submit test reports for information for assessing conformance with information given and design concept expressed in Contract Documents.

#### 1.13 CERTIFICATES

- A. Informational Submittal: Submit certification by manufacturer, installation/application Subcontractor, or Contractor to Engineer, in quantities specified for Product Data.
- B. Indicate material or product conforms to or exceeds specified requirements. Submit supporting reference data, affidavits, and certifications as appropriate.
- C. Certificates may be recent or previous test results on material or product but must be acceptable to Engineer.

## 1.14 MANUFACTURER'S INSTRUCTIONS

- A. Informational Submittal: Submit manufacturer's installation instructions for Engineer's knowledge as Contract administrator or for Owner.
- B. Submit printed instructions for delivery, storage, assembly, installation, startup, adjusting, and finishing, to Engineer in quantities specified for Product Data.
- C. Indicate special procedures, perimeter conditions requiring special attention, and special environmental criteria required for application or installation.

## 1.15 MANUFACTURER'S FIELD REPORTS

- A. Informational Submittal: Submit reports for Engineer's knowledge as Contract administrator or for Owner.
- B. Submit report in duplicate within 5 days of observation to Engineer for information.
- C. Submit reports for information for assessing conformance with information given and design concept expressed in Contract Documents.

## 1.16 ERECTION DRAWINGS

- A. Informational Submittal: Submit Drawings for Engineer's knowledge as Contract administrator or for Owner.
- B. Submit Drawings for information assessing conformance with information given and design concept expressed in Contract Documents.
- C. Data indicating inappropriate or unacceptable Work may be subject to action by Engineer or Owner.

## 1.17 CONSTRUCTION VIDEOS AND PHOTOGRAPHS

- A. The Contractor shall furnish all equipment and labor materials required to provide the Owner with digital construction videos and photographs of the Project.
- B. Prior to the beginning of any work, the Contractor shall take videos and photographs of the work area to record existing conditions.
- C. Photo and video files shall become the property of the Owner and none of the videos or photographs shall be published without express permission of the Owner.
- D. Following completion of the work, another set of videos and photographs shall be made showing the same areas and features as in the pre-construction videos and photographs.
- E. All conditions which might later be subject to disagreement shall be shown in sufficient detail to provide a basis for decisions.
- F. Photographs shall be in "jpg" format.
- G. Videos shall be in a format viewable by Microsoft Windows Media Player or Apple QuickTime Player. Audio narration is desirable.
- H. Files shall be named such that what is being viewed is self-evident.
- I. Files shall be submitted on a flash drive, compact disk (CD) or a digital video disk (DVD). If submitted on DVD, disk shall be recorded in "Minus R" format.
- J. The pre-construction videos and photographs shall be submitted to the Engineer within 25 calendar days after the date of receipt by the Contractor of Notice to Proceed.

- K. Each month submit photographs with Application for Payment.
- L. Post-construction videos and photographs shall be provided prior to final acceptance of the Project.

#### 1.18 CONTRACTOR REVIEW

- A. Review for compliance with Contract Documents and approve submittals before transmitting to Engineer.
- B. Contractor: Responsible for:
  - 1. Determination and verification of materials including manufacturer's catalog numbers.
  - 2. Determination and verification of field measurements and field construction criteria.
  - 3. Checking and coordinating information in submittal with requirements of Work and of Contract Documents.
  - 4. Determination of accuracy and completeness of dimensions and quantities.
  - 5. Confirmation and coordination of dimensions and field conditions at Site.
  - 6. Construction means, techniques, sequences, and procedures.
  - 7. Safety precautions.
  - 8. Coordination and performance of Work of all trades.
- C. Stamp, sign or initial, and date each submittal to certify compliance with requirements of Contract Documents.
- D. Do not fabricate products or begin Work for which submittals are required until approved submittals have been received from Engineer.

## 1.19 ENGINEER REVIEW

- A. Do not make "mass submittals" to Engineer. "Mass submittals" are defined as six or more submittals or items in one day or 15 or more submittals or items in one week. If "mass submittals" are received, Engineer's review time stated above will be extended as necessary to perform proper review. Engineer will review "mass submittals" based on priority determined by Engineer after consultation with Owner and Contractor.
- B. Informational submittals and other similar data are for Engineer's information, do not require Engineer's responsive action, and will not be reviewed or returned with comment.
- C. Submittals made by Contractor that are not required by Contract Documents may be returned without action.
- D. Submittal approval does not authorize changes to Contract requirements unless accompanied by Change Order, Field Order, or Construction Change Directive.
- E. Owner may withhold monies due to Contractor to cover additional costs beyond the second submittal review.

PART 2 - PRODUCTS - (NOT USED)

PART 3 - EXECUTION - (NOT USED)

**END OF SECTION 013300** 

# SECTION 014000 QUALITY REQUIREMENTS

## **PART 1 - GENERAL**

#### 1.1 SECTION INCLUDES

- A. Quality control.
- B. Tolerances.
- C. References.
- D. Labeling.
- E. Mockup requirements.
- F. Testing and inspection services.
- G. Manufacturers' field services.

## 1.2 QUALITY CONTROL

- A. Monitor quality control over suppliers, manufacturers, products, services, Site conditions, and workmanship, to produce Work of specified quality.
- B. Comply with specified standards as the minimum quality for the Work except where more stringent tolerances, codes, or specified requirements indicate higher standards or more precise workmanship.
- C. Perform Work using persons qualified to produce required and specified quality.
- D. Products, materials, and equipment may be subject to inspection by Engineer at place of manufacture or fabrication. Such inspections shall not relieve Contractor of complying with requirements of Contract Documents.
- E. Supervise performance of Work in such manner and by such means to ensure that Work, whether completed or in progress, will not be subjected to harmful, dangerous, damaging, or otherwise deleterious exposure during construction period.

## 1.3 TOLERANCES

- A. Monitor fabrication and installation tolerance control of products to produce acceptable Work. Do not permit tolerances to accumulate.
- B. Comply with manufacturers' recommended tolerances and tolerance requirements in reference standards. When such tolerances conflict with Contract Documents, request clarification from Engineer before proceeding.

C. Adjust products to appropriate dimensions; position before securing products in place.

## 1.4 REFERENCES

- A. For products or workmanship specified by association, trade, or other consensus standards, comply with requirements of standard except when more rigid requirements are specified or are required by applicable codes.
- B. Conform to reference standard by date of issue current as of date of Contract Documents except where specific date is established by code.
- C. Obtain copies of standards and maintain on Site when required by product Specification Sections.
- D. When requirements of indicated reference standards conflict with Contract Documents, request clarification from Engineer before proceeding.
- E. Neither contractual relationships, duties, or responsibilities of parties in Contract nor those of Engineer shall be altered from Contract Documents by mention or inference in reference documents.

## 1.5 LABELING

- A. Attach label from agency approved by authorities having jurisdiction for products, assemblies, and systems required to be labeled by applicable code.
- B. Label Information: Include manufacturer's or fabricator's identification, approved agency identification, and the following information, as applicable, on each label:
  - 1. Model number.
  - 2. Serial number.
  - 3. Performance characteristics.
- C. Manufacturer's Nameplates, Trademarks, Logos, and Other Identifying Marks on Products: Not allowed on surfaces exposed to view in public areas, interior or exterior.

## 1.6 MOCK-UP REQUIREMENTS – (NOT USED)

- A. Tests will be performed under provisions identified in this Section and identified in individual product Specification Sections.
- B. Assemble and erect specified or indicated items with specified or indicated attachment and anchorage devices, flashings, seals, and finishes.
- C. Accepted mockups shall be comparison standard for remaining Work.
- D. Where mockup has been accepted by Engineer and is specified in product Specification Sections to be removed, remove mockup and clear area when directed to do so by Engineer.

# 1.7 TESTING AND INSPECTION SERVICES

- A. Owner will provide a representative to perform testing and inspection.
- B. The representative will perform tests, inspections, and other services specified in individual Specification Sections and as required by Engineer and authorities having jurisdiction.
  - 1. Laboratory: Authorized to operate at Project location in State of Georgia.
  - 2. Laboratory Staff: Maintain full-time specialist on staff to review services.
  - 3. Testing Equipment: Calibrated at reasonable intervals with devices of an accuracy traceable to National Bureau of Standards or accepted values of natural physical constants.
- C. Testing, inspections, and source quality control may occur on or off Project Site. Perform off-Site testing as required by Engineer or Owner.
- D. Reports shall be submitted by representative to Engineer, Contractor, and authorities having jurisdiction, in duplicate, indicating observations and results of tests and compliance or noncompliance with Contract Documents.
  - 1. Submit final report indicating correction of Work previously reported as noncompliant.
- E. Cooperate with representative; furnish samples of materials, design mix, equipment, tools, storage, safe access, and assistance by incidental labor as requested.
  - 1. Notify Engineer and representative 24 hours before expected time for operations requiring services.
  - 2. Make arrangements with representative and pay for additional Samples and tests required for Contractor's use.
- F. Employment of testing agency or laboratory shall not relieve Contractor of obligation to perform Work according to requirements of Contract Documents.
- G. Retesting or re-inspection required because of nonconformance with specified or indicated requirements shall be performed by same independent firm on instructions from Engineer. Payment for retesting or re-inspection will be charged to Contractor by deducting testing charges from Contract Sum/Price.
- H. Agency Responsibilities:
  - 1. Test Samples of mixes submitted by Contractor.
  - 2. Provide qualified personnel at Site. Cooperate with Engineer and Contractor in performance of services.
  - 3. Perform indicated sampling and testing of products according to specified standards.

- 4. Ascertain compliance of materials and mixes with requirements of Contract Documents.
- 5. Promptly notify Engineer and Contractor of observed irregularities or nonconformance of Work or products.
- 6. Perform additional tests required by Engineer.
- 7. Attend preconstruction meetings and progress meetings.

- I. Agency Reports: After each test, promptly submit two copies of report to Engineer, Contractor, and authorities having jurisdiction. When requested by Engineer, provide interpretation of test results. Include the following:
  - 1. Date issued.
  - 2. Project title and number.
  - 3. Name of inspector.
  - 4. Date and time of sampling or inspection.
  - 5. Identification of product and Specification Section.
  - 6. Location in Project.
  - 7. Type of inspection or test.
  - 8. Date of test.
  - 9. Results of tests.
  - 10. Conformance with Contract Documents.
- J. Limits on Testing Authority:
  - 1. Agency or laboratory may not release, revoke, alter, or enlarge on requirements of Contract Documents.
  - 2. Agency or laboratory may not approve or accept any portion of the Work.
  - 3. Agency or laboratory may not assume duties of Contractor.
  - 4. Agency or laboratory has no authority to stop the Work.

## 1.8 MANUFACTURER'S FIELD SERVICES

- A. When specified in individual Specification Sections, require material or product suppliers or manufacturers to provide qualified staff personnel to observe Site conditions, conditions of surfaces and installation, quality of workmanship, startup of equipment, testing, adjusting, and balancing of equipment, commissioning as applicable, and to initiate instructions when necessary.
- B. Submit qualifications of observer to Engineer 30 days in advance of required observations. Observer is subject to approval of Engineer. Observer is subject to approval of Owner.
- C. Report observations and Site decisions or instructions given to applicators or installers that are supplemental or contrary to manufacturer's written instructions.
- D. Refer to Section 013300 Submittal Procedures, "Manufacturer's Field Reports" Article.

# PART 2 - PRODUCTS - (NOT USED)

# PART 3 - EXECUTION - (NOT USED)

#### **END OF SECTION 014000**

# **SECTION 014529 - TESTING LABORATORY SERVICES**

## **PART 1 - GENERAL**

## 1.1 SCOPE

- A. This Section includes testing which the Owner may require, beyond that testing required of the manufacturer, to determine if materials provided for the Project meet the requirements of these Specifications.
- B. This work also includes all testing required by the Owner to verify work performed by the Contractor is in accordance with the requirements of these Specifications, i.e., concrete strength and slump testing, soil compaction, etc.
- C. This work does not include materials testing required in various sections of these Specifications to be performed by the manufacturer, e.g., testing of pipe.
- D. The testing laboratory or laboratories will be selected by the Contractor and approved by the Owner. The testing laboratory or laboratories will work for the Contractor and payment for the testing required in these specifications will be the responsibility of the Contractor.

# 1.2 PAYMENT FOR TESTING SERVICES

- A. The cost of testing services required by the Contract to be provided by the Contractor shall be paid for by the Contractor. No special payment will be made for the testing required by these specifications, as the testing will be considered a part of the bid items to which they pertain.
- B. The cost of additional testing services not specifically required in the Specifications, but requested by the Owner or Engineer, shall be paid for by the Owner.
- C. The cost of material testing described in various sections of these Specifications or as required in referenced standards to be provided by a material manufacturer, shall be included in the price bid for that item and shall not be paid for by the Owner.
- D. The cost of retesting any item that fails to meet the requirements of these Specifications shall be paid for by the Contractor. Retesting shall be performed by the testing laboratory working for the Owner.

## 1.3 LABORATORY DUTIES

- A. Cooperate with the Owner, Engineer and Contractor.
- B. Provide qualified personnel promptly on notice.
- C. Perform specified inspections, sampling and testing of materials.
  - 1. Comply with specified standards, ASTM, other recognized authorities, and as specified.

051.18.2.22

2. Ascertain compliance with requirements of the Contract Documents.

- D. Promptly notify the Engineer and Contractor of irregularities or deficiencies of work which are observed during performance of services.
- E. Promptly submit three copies (two copies to the Engineer and one copy to the Contractor) of report of inspections and tests in addition to those additional copies required by the Contractor with the following information included:
  - 1. Date issued
  - 2. Project title and number
  - 3. Testing laboratory name and address
  - 4. Name and signature of inspector
  - 5. Date of inspection or sampling
  - 6. Record of temperature and weather
  - 7. Date of test
  - 8. Identification of product and Specification section
  - 9. Location of Project
  - 10. Type of inspection or test
  - 11. Results of test
  - 12. Observations regarding compliance with the Contract Documents
- F. Perform additional services as required.
- G. The laboratory is not authorized to release, revoke, alter or enlarge on requirements of the Contract Documents, or approve or accept any portion of the work.

## 1.4 CONTRACTOR RESPONSIBILITIES

- A. Cooperate with laboratory personnel, provide access to work and/or comply with manufacturer's requirements.
- B. Provide to the laboratory, representative samples, in required quantities, of materials to be tested.
- C. Furnish copies of mill test reports.
- D. Furnish required labor and facilities to:
  - 1. Provide access to work to be tested;

- 2. Obtain and handle samples at the site;
- 3. Facilitate inspections and tests;
- 4. Build or furnish a holding box for concrete cylinders or other samples as required by the laboratory.
- E. Notify the laboratory sufficiently in advance of operation to allow for the assignment of personnel and schedules of tests.
- F. Laboratory Tests: Where such inspection and testing are to be conducted by an independent laboratory agency, the sample(s) shall be selected by such laboratory or agency, or the Engineer, and shipped to the laboratory by the Contractor at Contractor's expense.
- G. Copies of all correspondence between the Contractor and testing agencies shall be provided to the Engineer.

# 1.5 QUALITY ASSURANCE

Testing shall be in accordance with all pertinent codes and regulations and with procedures and requirements of the American Society for Testing and Materials (ASTM).

## 1.6 PRODUCT HANDLING

Promptly process and distribute all required copies of test reports and related instructions to insure all necessary retesting or replacement of materials with the least possible delay in the progress of the work.

## 1.7 FURNISHING MATERIALS

The Contractor shall be responsible for furnishing all materials necessary for testing.

## 1.8 CODE COMPLIANCE TESTING

Inspections and tests required by codes or ordinances or by a plan approval authority, and made by a legally constituted authority, shall be the responsibility of, and shall be paid for by the Contractor, unless otherwise provided in the Contract Documents.

## 1.9 CONTRACTOR'S CONVENIENCE TESTING

Inspection or testing performed exclusively for the Contractor's convenience shall be the sole responsibility of the Contractor.

051.18.2.22

## 1.10 SCHEDULES FOR TESTING

A. Establishing Schedule

- 1. The Contractor shall, by advance discussion with the testing laboratory selected by the Owner, determine the time required for the laboratory to perform its tests and to issue each of its findings, and make all arrangements for the testing laboratory to be on site to provide the required testing.
- 2. Provide all required time within the construction schedule.
- B. When changes of construction schedule are necessary during construction, coordinate all such changes of schedule with the testing laboratory as required.
- C. When the testing laboratory is ready to test according to the determined schedule, but is prevented from testing or taking specimens due to incompleteness of the work, all extra costs for testing attributable to the delay will be back-charged to the Contractor and shall not be borne by the Owner.

## 1.11 TAKING SPECIMENS

Unless otherwise provided in the Contract Documents, all specimens and samples for tests will be taken by the testing laboratory or the Engineer.

## 1.12 TRANSPORTING SAMPLES

The Contractor shall be responsible for transporting all samples, except those taken by testing laboratory personnel, to the testing laboratory.

PART 2 - PRODUCTS - Not Used

**PART 3 - EXECUTION - Not Used** 

**END OF SECTION 014529** 

# SECTION 015000 TEMPORARY FACILITIES AND CONTROLS

051.18.2.22

# **PART 1 - GENERAL**

# 1.1 SECTION INCLUDES

- A. Temporary facilities under Construction Management Agreement.
- B. Temporary Utilities:
  - 1. Temporary electricity.
  - 2. Temporary lighting for construction purposes.
  - 3. Temporary heating.
  - 4. Temporary cooling.
  - 5. Temporary ventilation.
  - 6. Communication services.
  - 7. Temporary water service.
  - 8. Temporary sanitary facilities.

## C. Construction Facilities:

- 1. Field offices and sheds.-Not Required
- 2. Vehicular access.
- 3. Parking.
- 4. Progress cleaning and waste removal.
- 5. Project identification.
- 6. Traffic regulation.
- 7. Fire-prevention facilities.

# D. Temporary Controls:

- 1. Barriers.
- 2. Enclosures and fencing.
- 3. Security.
- 4. Water control.
- 5. Dust control.
- 6. Erosion and sediment control.
- 7. Noise control.
- 8. Pest and rodent control.
- 9. Pollution control.
- E. Removal of utilities, facilities, and controls.

## 1.2 REFERENCES

A. ASTM International:

- 1. ASTM E 84 Standard Test Method for Surface Burning Characteristics of Building Materials.
- 2. ASTM E 90 Standard Test Method for Laboratory Measurement of Airborne Sound Transmission Loss of Building Partitions and Elements.
- 3. ASTM E 119 Standard Test Methods for Fire Tests of Building Construction and Materials.

# 1.3 TEMPORARY FACILITIES UNDER CONSTRUCTION MANAGEMENT AGREEMENT

- A. Temporary Provisions Provided by Construction Manager:
  - 1. Temporary barriers, barricades, covered walkways, fencing, exterior closures, and interior closures.
  - 2. Temporary field offices.
  - 3. Cleaning during construction.
  - 4. Access roads and approaches.
  - 5. Temporary elevator.
  - 6. Temporary sanitary facilities.
  - 7. Temporary heating and ventilating after enclosure.
  - 8. Temporary electrical service and distribution system for power and lighting.
  - 9. Temporary telephone and internet service.
- B. Each Contractor: Coordinate provisions with Construction Manager and provide the following items as necessary for execution of the Work including associated costs:
  - 1. Construction aids.
  - 2. Temporary fire protection, dust control, erosion and sediment control, water control, noise control, and other necessary temporary controls.
  - 3. Temporary barriers, barricades, and similar devices as necessary for safety and protection of construction personnel and public.
  - 4. On Construction Manager's approval, may provide temporary field office including electrical service and temporary telephone.
  - 5. Temporary tree and plant protection.
  - 6. Temporary heating before building enclosure.
  - 7. Electrical service required in addition to temporary service and distribution provided by Construction Manager.
  - 8. Temporary provisions for protection of installed Work.

#### 1.4 TEMPORARY ELECTRICITY

- A. Provide and pay for power service required from utility source as needed for construction operation. Use Owner's existing power service, provide separate metering, and reimburse Owner for cost of energy used.
- B. Provide temporary electric feeder from existing electrical service as directed by Engineer. Do not disrupt Owner's use of service.

- C. Complement existing power service capacity and characteristics as required for construction operations.
- D. Provide power outlets with branch wiring and distribution boxes located as required for construction operations. Provide suitable, flexible power cords as required for portable construction tools and equipment.
- E. Provide meter at source distribution equipment.
- F. Permanent convenience receptacles may be used during construction.
- G. Provide distribution equipment, wiring, and outlets for single-phase branch circuits for power and lighting.
  - 1. Provide 20-ampere duplex outlets, single-phase circuits for power tools for active work
  - 2. Provide 20-ampere, single-phase branch circuits for lighting.

# 1.5 TEMPORARY LIGHTING FOR CONSTRUCTION PURPOSES

- A. Provide and maintain incandescent lighting for construction operations.
- B. Provide and maintain after dark for security purposes.
- C. Provide and maintain HID lighting to interior work areas after dark for security purposes.
- D. Provide branch wiring from power source to distribution boxes with lighting conductors, pigtails, lamps, and the like, for specified lighting levels.
- E. Maintain lighting and provide routine repairs.
- F. Permanent building lighting may be used during construction.

## 1.6 TEMPORARY HEATING

- A. Provide heating devices and heat as needed to maintain specified conditions for construction operations.
- B. Enclose building before activating temporary heat according to "Enclosures and Fencing" Article in this Section.
- C. Before operating permanent equipment for temporary heating purposes, verify installation is approved for operation, equipment is lubricated, and filters are in place. Provide and pay for operation, maintenance, and regular replacement of filters and worn or consumed parts. Replace filters at Substantial Completion.
- D. Maintain minimum ambient temperature in areas where construction is in progress unless indicated otherwise in individual product Sections.

# 1.7 TEMPORARY COOLING-N/A

- A. Provide cooling devices and cooling as needed to maintain specified conditions for construction operations.
- B. Enclose building before activating temporary cooling according to ""Enclosures and Fencing"" Article in this Section.
- C. Before operating permanent equipment for temporary cooling purposes, verify installation is approved for operation, equipment is lubricated, and filters are in place. Provide and pay for operation, maintenance, and regular replacement of filters and worn or consumed parts. Replace filters at Substantial Completion.
- D. Maintain maximum ambient temperature in areas where construction is in progress unless indicated otherwise in individual product Sections.

## 1.8 TEMPORARY VENTILATION-N/A

- A. Ventilate enclosed areas to achieve curing of materials, to dissipate humidity, and to prevent accumulation of dust, fumes, vapors, or gases.
- B. Use existing ventilation equipment. Extend and supplement equipment with temporary fan units as required to maintain clean air for construction operations.

# 1.9 COMMUNICATION SERVICES-N/A

- A. Telephone Service: Provide, maintain, and pay for telephone service to field office and Engineer's field office at time of Project mobilization and until completion of Work.
- B. Facsimile Service: Provide, maintain, and pay for facsimile service including dedicated telephone line to field office and Engineer's field office at time of Project mobilization and until completion of Work.
- C. Internet Service: Provide, maintain, and pay for broadband Internet service to field office and Engineer's field office at time of Project mobilization. Provide desktop computer with Microsoft operating system and appropriate office function software, modem, and printer.

# 1.10 TEMPORARY WATER SERVICE

- A. Provide suitable quality water service as needed to maintain specified conditions for construction operations.
- B. Extend branch piping with outlets located so that water is available by hoses with threaded connections. Provide temporary pipe insulation and heat tape to prevent freezing.

# 1.11 TEMPORARY SANITARY FACILITIES

- A. Provide and maintain required facilities and enclosures. Existing facility use is not permitted. Provide facilities at time of Project mobilization.
- B. At end of construction, return existing facilities used for construction operations to same or better condition as original condition.

## 1.12 FIELD OFFICES AND SHEDS-Not Required

- A. Designated existing spaces may be used for field offices,
- B. Provide space for Project meetings, with table and chairs.
- C. Provide separate private office similarly equipped and furnished, for use of Engineer.
- D. Locate field offices and sheds a minimum distance of 30 feet from existing and new structures.
- E. Do not use permanent facilities for field offices or for storage.
- F. Construction: Portable or mobile buildings, or buildings constructed with floors raised aboveground, securely fixed to foundations with steps and landings at entrance doors.
  - 1. Construction: Structurally sound, secure, weathertight enclosures for office and storage spaces. Maintain during progress of Work; remove enclosures at completion of Work.
  - 2. Thermal Resistance of Floors, Walls, and Ceilings: Compatible with occupancy and storage requirements.
  - 3. Exterior Materials: Weather-resistant, finished in one color acceptable to Engineer.
  - 4. Interior Materials in Field Offices: Sheet-type materials for walls and ceilings, prefinished or painted; resilient floors and bases.
  - 5. Lighting for Field Offices: 50 ft-C at desktop height; exterior lighting at entrance doors.
  - 6. Interior Materials in Storage Sheds: As required to provide specified conditions for storage of products.

## G. Environmental Control:

- 1. Heating, Cooling, and Ventilating for Offices: Automatic equipment to maintain comfort conditions.
- 2. Storage Spaces: Heating and ventilating as needed to maintain products according to Contract Documents; lighting for maintenance and inspection of products.

# H. Engineer Field Office:

- 1. Separate space for use of Engineer only, with separate entrance door with new lock and two keys.
- 2. Area: Minimum 150 sq ft, minimum dimension 8 ft.
- 3. Windows: Minimum of three with a minimum total area of 10 percent of floor area, with operable sash and insect screens. Locate windows to provide views of construction area.
- 4. Electrical Distribution Panel: Two circuits minimum, 110-volt, 60 Hz service.

051.18.2.22

5. Minimum of four 110-volt duplex convenience outlets, one on each wall.

- 6. Communication Services: As specified in this Section.
- 7. Sanitary Facilities: Convenient access to private lavatory facilities.
- 8. Drinking Fountain: Convenient access by workers.
- 9. Engineer Field Office Furnishings:
  - a. One desk 54 x 30 inches, with three drawers.
  - b. One drafting table 36 x 72 inches, with one equipment drawer and full-width parallel straight edge.
  - c. One metal, double-door storage cabinet under table.
  - d. Plan rack to hold working Drawings, Shop Drawings, and Record Documents.
  - e. One standard four-drawer letter-size metal filling cabinet with locks and two keys for each lock.
  - f. Six linear ft of metal bookshelves.
  - g. Two swivel armchairs.
  - h. Two straight chairs.
  - i. One drafting table stool.
  - j. One tackboard 36 x 30 inches.
  - k. One wastebasket for each desk and table.
- I. Storage Areas and Sheds: Size to storage requirements for products of individual Sections, allowing for access and orderly provision for maintenance and inspection of products to suit requirements in Section 016000 Product Requirements.
- J. Preparation: Fill and grade Sites for temporary structures sloped for drainage away from buildings.

# K. Installation:

- 1. Install field office spaces ready for occupancy 15 days after date established by Notice to Proceed.
- 2. Employee Residential Occupancy: Not allowed on Owner's property.

# L. Maintenance and Cleaning:

- 1. Weekly janitorial services for field offices; periodic cleaning and maintenance for sheds and storage areas.
- 2. Maintain walks free of mud, water, snow, and the like.
- M. Removal: At completion of Work remove buildings, foundations, utility services, and debris. Restore areas to same or better condition as original condition.

## 1.13 VEHICULAR ACCESS

- A. Construct temporary all-weather access roads from public thoroughfares to serve construction area, of width and load-bearing capacity to accommodate unimpeded traffic for construction purposes.
- B. Construct temporary bridges and culverts to span low areas and allow unimpeded drainage.

- C. Extend and relocate vehicular access as Work progress requires and provide detours as necessary for unimpeded traffic flow.
- D. Locate as approved by Engineer.
- E. Provide unimpeded access for emergency vehicles. Maintain 20-foot-wide driveways with turning space between and around combustible materials.
- F. Provide and maintain access to fire hydrants and control valves free of obstructions.
- G. Provide means of removing mud from vehicle wheels before entering streets.
- H. Use designated existing on-Site roads for construction traffic.

# 1.14 PARKING

- A. Provide temporary gravel surface parking areas to accommodate construction personnel.
- B. Locate as approved by Engineer.
- C. If Site space is not adequate, provide additional off-Site parking.
- D. Use of designated areas of existing on-Site streets and driveways used for construction traffic is permitted. Tracked vehicles are not allowed on paved areas.
- E. Use of designated areas of existing parking facilities used by construction personnel is [not] permitted.
- F. Do not allow heavy vehicles or construction equipment in parking areas.
- G. Do not allow vehicle parking on existing pavement.
- H. Designate one parking spaces for Engineer.
- I. Permanent Pavements and Parking Facilities:
  - 1. Bases for permanent roads and parking areas may be used for construction traffic.
  - 2. Avoid traffic loading beyond paving design capacity. Tracked vehicles are not allowed.
  - 3. Use of permanent parking structures is not permitted.

#### J. Maintenance:

- 1. Maintain traffic and parking areas in sound condition free of excavated material, construction equipment, products, mud, snow, ice, and the like.
- 2. Maintain existing and permanent paved areas used for construction; promptly repair breaks, potholes, low areas, standing water, and other deficiencies, to maintain paving and drainage in original condition.

051.18.2.22

K. Removal, Repair:

- 1. Remove temporary materials and construction when permanent paving is usable at Substantial Completion.
- 2. Remove underground Work and compacted materials to depth of 2 feet; fill and grade Site as indicated.
- 3. Repair existing facilities damaged by use, to original condition.
- L. Mud from Site vehicles: Provide means of removing mud from vehicle wheels before entering streets.

## 1.15 PROGRESS CLEANING AND WASTE REMOVAL

- A. Maintain areas free of waste materials, debris, and rubbish. Maintain Site in clean and orderly condition.
- B. Remove debris and rubbish from pipe chases, plenums, attics, crawl spaces, and other closed or remote spaces, before enclosing spaces.
- C. Broom and vacuum clean interior areas before starting surface finishing, and continue cleaning to eliminate dust.
- D. Collect and remove waste materials, debris, and rubbish from Site weekly and dispose of off-Site. Comply with Section 017419 Construction Waste Management and Disposal.
- E. Open free-fall chutes are not permitted. Terminate closed chutes into appropriate containers with lids.

# 1.16 PROJECT IDENTIFICATION

- A. Project Identification Sign:
  - 1. One painted sign of construction, design, and content shown on Drawings, location designated.
  - 2. Content:
    - a. Project number, title, logo, and name of Owner.
    - b. Names and titles of authorities.
    - c. Names and titles of Engineer and Consultants.
    - d. Name of Prime Contractor and major Subcontractors.
  - 3. Lettering: Series C of Standard Alphabet for Traffic Control Devices, "Manual on Uniform Traffic Control Devices for Streets and Highways," Federal Highway Administration.
- B. Project Informational Signs:
  - 1. Painted informational signs of same colors and lettering as Project identification sign or standard products; size lettering for legibility at 100-foot distance.

- 2. Provide sign at each field office and storage shed, and provide directional signs to direct traffic into and within Site. Relocate as Work progress requires.
- 3. Provide state traffic agency directional traffic signs to and within Site.
- 4. No other signs are allowed without Owner's permission except those required by law.
- C. Design sign and structure to withstand 60-mph wind velocity.
- D. Sign Painter: Experienced as professional sign painter for minimum of three years.
- E. Finishes, Painting: Adequate to withstand weathering, fading, and chipping for duration of construction.
- F. Show content, layout, lettering, color, foundation, structure, sizes, and grades of members.

# G. Sign Materials:

- 1. Structure and Framing: New and structurally adequate.
- 2. Sign Surfaces: Exterior grade plywood with medium-density overlay, minimum of 3/4 inches thick, standard large sizes to minimize joints.
- 3. Rough Hardware: Galvanized aluminum or brass.
- 4. Paint and Primers: Exterior quality, two coats; sign background of color as selected.
- 5. Lettering: Precut vinyl self-adhesive products or quality exterior paint, contrasting in colors as selected.

#### H. Installation:

- 1. Install Project identification sign within 15 days after date established by Notice to Proceed.
- 2. Erect at designated location.
- 3. Erect supports and framing on secure foundation, rigidly braced and framed to resist wind loadings.
- 4. Install sign surface plumb and level, with butt joints. Anchor securely.
- 5. Paint exposed surfaces of sign, supports, and framing.
- I. Maintenance: Maintain clean signs and supports; repair deterioration and damage.
- J. Removal: Remove signs, framing, supports, and foundations at completion of Project and restore area.

# 1.17 TRAFFIC REGULATION

- A. Signs, Signals, and Devices:
  - 1. Post-Mounted and Wall-Mounted Traffic Control and Informational Signs: As approved by authorities having jurisdiction.
  - 2. Automatic Traffic Control Signals: As approved by local jurisdictions.
  - 3. Traffic Cones, Drums, Flares, and Lights: As approved by authorities having jurisdiction.
  - 4. Flag Person Equipment: As required by authorities having jurisdiction.

- B. Flag Persons: Provide trained and equipped flag persons to regulate traffic when construction operations or traffic encroach on public traffic lanes.
- C. Flares and Lights: Use flares and lights during hours of low visibility to delineate traffic lanes and to guide traffic.

#### D. Haul Routes:

- 1. Consult with authorities having jurisdiction and establish public thoroughfares to be used for haul routes and Site access.
- 2. Confine construction traffic to designated haul routes.
- 3. Provide traffic control at critical areas of haul routes to regulate traffic and to minimize interference with public traffic.

# E. Traffic Signs and Signals:

- 1. Provide signs at approaches to Site and on Site, at crossroads, detours, parking areas, and elsewhere as needed to direct construction and affected public traffic.
- 2. Provide, operate, and maintain automatic traffic control signals to direct and maintain orderly flow of traffic in areas under Contractor's control and areas affected by Contractor's operations.
- 3. Relocate signs and signals as Work progresses, to maintain effective traffic control.

#### F. Removal:

- 1. Remove equipment and devices at Substantial Completion.
- 2. Repair damage caused by installation.
- 3. Remove post settings to depth of 2 feet.

## 1.18 FIRE-PREVENTION FACILITIES

- A. Prohibit smoking within buildings under construction and demolition. Designate area on Site where smoking is permitted. Provide approved ashtrays in designated smoking areas.
- B. Establish fire watch for cutting, welding, and other hazardous operations capable of starting fires. Maintain fire watch before, during, and after hazardous operations until threat of fire does not exist.
- C. Standpipes: Maintain existing standpipes in usable condition to height within one floor of floor being demolished.
- D. Portable Fire Extinguishers: NFPA 10; 10-pound capacity, 4A-60B: C UL rating.
  - 1. Provide one fire extinguisher at each stairway on each floor of buildings under construction and demolition.
  - 2. Provide minimum of one fire extinguisher in every construction trailer and storage shed.

051.18.2.22

3. Provide minimum of one fire extinguisher on roof during roofing operations using heat-producing equipment.

# 1.19 BARRIERS

- A. Provide barriers to prevent unauthorized entry to construction areas and to protect existing facilities and adjacent properties from damage from construction operations and demolition.
- B. Provide barricades and covered walkways required by authorities having jurisdiction for public rights-of-way and for public access to existing building.
  - 1. Barricade Construction: As indicated on Drawings.
  - 2. Covered Walkway Construction: As indicated on Drawings.
- C. Tree and Plant Protection: Preserve and protect existing trees and plants designated to remain.
  - 1. Protect areas within drip lines from traffic, parking, storage, dumping, chemically injurious materials and liquids, ponding, and continuous running water.
  - 2. Provide 6-foot high barriers around drip line, with access for maintenance.
  - 3. Replace trees and plants damaged by construction operations.
- D. Protect non-owned vehicular traffic, stored materials, Site, and structures from damage.

## 1.20 ENCLOSURES AND FENCING

- A. Construction: Contractor's option.
- B. Provide 6-foot high fence around construction Site; equip with vehicular and pedestrian gates with locks.
- C. Exterior Enclosures:
  - 1. Provide temporary insulated weathertight closure of exterior openings to accommodate acceptable working conditions and protection for products, to allow for temporary heating and maintenance of required ambient temperatures identified in individual Specification Sections, and to prevent entry of unauthorized persons. Provide access doors with self-closing hardware and locks.
  - 2. Provide temporary roofing as specified.

## D. Interior Enclosures:

- 1. Provide temporary partitions and ceilings as indicated on Drawings to separate work areas from Owner-occupied areas, to prevent penetration of dust and moisture into Owner-occupied areas, and to prevent damage to existing materials and equipment.
- 2. Construction: Framing and reinforced polyethylene, plywood or gypsum board materials with closed joints and sealed edges at intersections with existing surfaces.
  - a. Insulated to R-19.
  - b. STC rating of 35 according to ASTM E 90.
  - c. Surface-Burning Characteristics: Maximum flame-spread/smoke-developed index when tested according to ASTM E 84.
  - d. Fire-Rated Wall, Floor or Roof Construction: Rating as indicated on Drawings.
    - 1) Tested Rating: Determined according to ASTM E 119.

3. Paint surfaces exposed to view from Owner-occupied areas.

# 1.21 SECURITY

# A. Security Program:

- 1. Protect Work on existing premises and Owner's operations from theft, vandalism, and unauthorized entry.
- 2. Initiate program in coordination with Owner's existing security system at Project mobilization.
- 3. Maintain program throughout construction period until directed by Engineer.

## B. Entry Control:

- 1. Restrict entrance of persons and vehicles to Project Site and existing facilities.
- 2. Allow entrance only to authorized persons with proper identification.
- 3. Maintain log of workers and visitors and make available to Owner on request.
- 4. Coordinate access of Owner's personnel to Site in coordination with Owner's security forces.

#### C. Personnel Identification:

- 1. Provide identification badge for each person authorized to enter premises.
- 2. Badge to Include: Personal photograph, name, assigned number, expiration date, and employer.
- 3. Maintain list of accredited persons and submit copy to Owner on request.
- 4. Require return of badges at expiration of employment on the Work.

# D. Security Service:

1. Employ uniformed guard or patrol service to provide guards on Site twenty-four hours a day, seven days a week.

## E. Restrictions:

- 1. Do not allow cameras on Site or photographs taken except by written approval of Owner.
- 2. Do no work on days indicated in Owner-Contractor Agreement.

## 1.22 WATER CONTROL

- A. Grade Site to drain. Maintain excavations free of water. Provide, operate, and maintain necessary pumping equipment.
- B. Protect Site from puddles or running water.

# 1.23 DUST CONTROL

A. Execute Work by methods that minimize raising dust from construction operations.

B. Provide positive means to prevent airborne dust from dispersing into atmosphere and into Owner-occupied areas.

# 1.24 EROSION AND SEDIMENT CONTROL

- A. Plan and execute construction by methods to control surface drainage from cuts and fills from borrow and waste disposal areas. Prevent erosion and sedimentation.
- B. Minimize surface area of bare soil exposed at one time.
- C. Provide temporary measures including berms, dikes, drains, and other devices to prevent water flow.
- D. Construct fill and waste areas by selective placement to avoid erosive surface silts and clays.
- E. Periodically inspect earthwork to detect evidence of erosion and sedimentation. Promptly apply corrective measures.
- F. Comply with sediment and erosion control plan indicated on Drawings.

## 1.25 NOISE CONTROL

A. Provide methods, means, and facilities to minimize noise produced by construction operations.

## 1.26 PEST AND RODENT CONTROL

- A. Provide methods, means, and facilities to prevent pests and insects from damaging the Work or entering facility.
- B. Provide methods, means, and facilities to prevent rodents from accessing or invading premises.

# 1.27 POLLUTION CONTROL

- A. Provide methods, means, and facilities to prevent contamination of soil, water, and atmosphere from discharge of noxious, toxic substances and pollutants produced by construction operations.
- B. Comply with pollution and environmental control requirements of authorities having jurisdiction.

# 1.28 REMOVAL OF UTILITIES, FACILITIES, AND CONTROLS

- A. Remove temporary utilities, equipment, facilities, and materials before Final Application for Payment.
- B. Remove underground installations to minimum depth of 2 feet. Grade Site as indicated on Drawings.

- C. Clean and repair damage caused by installation or use of temporary Work.
- D. Restore existing and permanent facilities used during construction to original condition. Restore permanent facilities used during construction to specified condition.

PART 2 - PRODUCTS - (NOT USED)

PART 3 - EXECUTION - (NOT USED)

**END OF SECTION 015000** 

# SECTION 015633 JOB SITE SECURITY

# **PART 1 - GENERAL**

## 1.1 SECTION INCLUDES

- A. Temporary Barricades.
- B. Temporary Fencing.

# 1.2 BARRICADES, LIGHTS AND SIGNALS

- A. The Contractor shall furnish and erect such barricades, fences, lights and danger signals and shall provide such other precautionary measures for the protection of persons or property and of the work as necessary. Barricades shall be painted in a color that will be visible at night. From sunset to sunrise, the Contractor shall furnish and maintain at least one light at each barricade and sufficient numbers of barricades shall be erected to keep vehicles from being driven on or into any work under construction.
- B. The Contractor will be held responsible for all damage to the work due to failure of barricades, signs and lights and whenever evidence is found of such damage, the Contractor shall immediately remove the damaged portion and replace it at Contractor's cost and expense. The Contractor's responsibility for the maintenance of barricades, signs and lights shall not cease until the Project has been accepted by the Owner.
- C. Traffic control devices shall comply with the latest edition of the Manual on Uniform Traffic Control Devices for Streets and Highways.

## 1.3 TEMPORARY FENCING-Contractor's Choice

A. Prior to the start of work, enclose those areas at the construction site which are not within the existing fencing with a temporary safety fence, including gates and warning signs, to protect the public from construction activities. Remove the fence from the work site upon completion of the contract.

PART 2 - PRODUCTS - (NOT USED)

PART 3 - EXECUTION - (NOT USED)

## **END OF SECTION 015633**

# SECTION 017000 EXECUTION AND CLOSEOUT REQUIREMENTS

# **PART 1 - GENERAL**

# 1.1 SECTION INCLUDES

- A. Field engineering.
- B. Closeout procedures.
- C. Starting of systems.
- D. Demonstration and instructions.
- E. Testing, adjusting, and balancing.
- F. Project record documents.
- G. Operation and maintenance data.
- H. Manual for materials and finishes.
- I. Manual for equipment and systems.
- J. Spare parts and maintenance products.
- K. Product warranties and product bonds.
- L. Maintenance service.
- M. Examination.
- N. Preparation.
- O. Execution.
- P. Cutting and patching.
- Q. Protecting installed construction.
- R. Final cleaning.

# 1.2 FIELD ENGINEERING

- A. Employ land surveyor registered in State of Georgia and acceptable to Engineer.
- B. Contractor shall locate and protect survey control and reference points. Promptly notify Engineer of discrepancies discovered.

- C. Control datum for survey is established by Owner-provided survey.
- D. Prior to beginning Work, verify and establish floor elevations of existing facilities to ensure that new Work will meet existing elevations in smooth and level alignment except where specifically detailed or indicated otherwise.
- E. Verify setbacks and easements; confirm Drawing dimensions and elevations.
- F. Provide field engineering services. Establish elevations, lines, and levels using recognized engineering survey practices.
- G. Submit copy of Site drawing and certificate signed by land surveyor certifying elevations and locations of the Work are in conformance with Contract Documents.
- H. Maintain complete and accurate log of control and survey Work as Work progresses.
- I. On completion of foundation walls and major Site improvements, prepare certified survey illustrating dimensions, locations, angles, elevations of construction and Site Work.
- J. Protect survey control points prior to starting Site Work; preserve permanent reference points during construction.
- K. Promptly report to Engineer loss or destruction of reference point or relocation required because of changes in grades or other reasons.
- L. Replace dislocated survey control points based on original survey control. Make no changes without prior written notice to /Engineer.
- M. Final Property Survey: Prior to Substantial Completion, prepare final property survey illustrating locations, dimensions, angles, and elevations of buildings and Site Work that have resulted from construction indicating their relationship to permanent bench marks and property lines.
  - 1. Show significant features (real property) for Project.
  - 2. Include certification on survey, signed by surveyor, that principal metes, bounds, lines, levels, and elevations of Project are accurately shown.

# 1.3 CLOSEOUT PROCEDURES

- A. Prerequisites to Substantial Completion: Complete following items before requesting Certification of Substantial Completion, either for entire Work or for portions of Work:
  - 1. Submit maintenance manuals, Project record documents, digital images of construction photographs, and other similar final record data in compliance with this Section.
  - 2. Complete facility startup, testing, adjusting, balancing of systems and equipment, demonstrations, and instructions to Owner's operating and maintenance personnel as specified in compliance with this Section.
  - 3. Conduct inspection to establish basis for request that Work is substantially complete. Create comprehensive list (initial punch list) indicating items to be completed or corrected, value of incomplete or nonconforming Work, reason for being incomplete, and date of

- anticipated completion for each item. Include copy of list with request for Certificate of Substantial Completion.
- 4. Obtain and submit releases enabling Owner's full, unrestricted use of Project and access to services and utilities. Include certificate of occupancy, operating certificates, and similar releases from authorities having jurisdiction and utility companies.
- 5. Deliver tools, spare parts, extra stocks of material, and similar physical items to Owner.
- 6. Make final change-over of locks and transmit keys directly to Owner. Advise Owner's personnel of change-over in security provisions.
- 7. Discontinue or change over and remove temporary facilities and services from Project Site, along with construction tools, mockups, and similar elements.
- 8. Perform final cleaning according to this Section.

# B. Substantial Completion Inspection:

- 1. When Contractor considers Work to be substantially complete, submit to Engineer:
  - a. Written certificate that Work, or designated portion, is substantially complete.
  - b. List of items to be completed or corrected (initial punch list).
- 2. Within 7 days after receipt of request for Substantial Completion, Engineer will make inspection to determine whether Work or designated portion is substantially complete.
- 3. Should Engineer determine that Work is not substantially complete:
  - a. Engineer will promptly notify Contractor in writing, stating reasons for its opinion.
  - b. Contractor shall remedy deficiencies in Work and send second written request for Substantial Completion to Engineer.
  - c. Engineer will reinspect Work.
  - d. Redo and Inspection of Deficient Work: Repeated until Work passes Engineer inspection.
- 4. When Engineer finds that Work is substantially complete, Engineer will:
  - a. Prepare Certificate of Substantial Completion, accompanied by Contractor's list of items to be completed or corrected as verified and amended by Engineer and Owner (final punch list).
  - b. Submit Certificate to Owner and Contractor for their written acceptance of responsibilities assigned to them in Certificate.
- 5. After Work is substantially complete, Contractor shall:
  - a. Allow Owner occupancy of Project under provisions stated in Certificate of Substantial Completion.
  - b. Complete Work listed for completion or correction within time period stipulated.
- 6. Owner will occupy all portions of building as specified in Section 011000 Summary.
- C. Prerequisites for Final Completion: Complete following items before requesting final acceptance and final payment.
  - 1. When Contractor considers Work to be complete, submit written certification that:

- Contract Documents have been reviewed.
- b. Work has been examined for compliance with Contract Documents.
- c. Work has been completed according to Contract Documents.
- d. Work is completed and ready for final inspection.

# 2. Submittals: Submit following:

- a. Final punch list indicating all items have been completed or corrected.
- b. Final payment request with final releases and supporting documentation not previously submitted and accepted. Include certificates of insurance for products and completed operations where required.
- c. Specified warranties, workmanship/maintenance bonds, maintenance agreements, and other similar documents.
- d. Accounting statement for final changes to Contract Sum.
- e. Contractor's affidavit of payment of debts and claims.
- f. Contractor affidavit of release of liens.
- g. Consent of surety to final payment.
- 3. Perform final cleaning for Contractor-soiled areas according to this Section.

# D. Final Completion Inspection:

- 1. Within 7 days after receipt of request for final inspection, Engineer will make inspection to determine whether Work or designated portion is complete.
- 2. Should Engineer consider Work to be incomplete or defective:
  - a. Engineer will promptly notify Contractor in writing, listing incomplete or defective Work.
  - b. Contractor shall remedy stated deficiencies and send second written request to Engineer that Work is complete.
  - c. Engineer will reinspect Work.
  - d. Redo and Inspection of Deficient Work: Repeated until Work passes Engineer's inspection.

# 1.4 STARTING OF SYSTEMS

- A. Coordinate schedule for startup of various equipment and systems.
- B. Notify Engineer 7 days prior to startup of each item.
- C. Verify that each piece of equipment or system has been checked for proper lubrication, drive rotation, belt tension, control sequence, and for conditions which may cause damage.
- D. Verify that tests, meter readings, and electrical characteristics agree with those required by equipment or system manufacturer.
- E. Verify that wiring and support components for equipment are complete and tested.
- F. Execute startup under supervision of manufacturer's representative or Contractors' personnel according to manufacturer's instructions.

- G. When specified in individual Specification Sections, require manufacturer to provide authorized representative who will be present at Site to inspect, check, and approve equipment or system installation prior to startup and will supervise placing equipment or system in operation.
- H. Submit a written report according to Section 013300 Submittal Procedures that equipment or system has been properly installed and is functioning correctly.

## 1.5 DEMONSTRATION AND INSTRUCTIONS

- A. Demonstrate operation and maintenance of products to Engineer when the Work is 80 percent complete.
- B. Demonstrate Project equipment and instruct in classroom environment by authorized manufacturer's representative who is knowledgeable about the Project.
- C. Video Recordings: Provide high-quality color video recordings of demonstration and instructional sessions. Engage commercial videographer to record sessions. Include classroom instructions, demonstrations, board diagrams, and other visual aids. Include menu navigation.
- D. For equipment or systems requiring seasonal operation, perform demonstration for other season within three months.
- E. Use operation and maintenance manuals as basis for instruction. Review contents of manual with Owner's personnel in detail to explain all aspects of operation and maintenance.
- F. Demonstrate startup, operation, control, adjustment, troubleshooting, servicing, maintenance, and shutdown of each item of equipment at agreed time and location.
- G. Prepare and insert additional data in operations and maintenance manuals when need for additional data becomes apparent during instruction.
- H. Required instruction time for each item of equipment and system is specified in individual Specification Sections.

# 1.6 TESTING, ADJUSTING, AND BALANCING

- A. Owner will appoint, employ, and pay for services of independent firm to perform testing, adjusting, and balancing.
- B. Independent firm will perform services specified.
- C. Reports will be submitted by independent firm to Engineer indicating observations and results of tests and indicating compliance or noncompliance with requirements of Contract Documents.

#### 1.7 PROJECT RECORD DOCUMENTS

A. Maintain on Site one set of the following record documents; record actual revisions to the Work:

- 1. Drawings.
- 2. Specifications.
- 3. Addenda.
- 4. Change Orders and other modifications to the Contract.
- 5. Reviewed Shop Drawings, product data, and Samples.
- 6. Manufacturer's instruction for assembly, installation, and adjusting.
- B. Ensure entries are complete and accurate, enabling future reference by Owner.
- C. Store record documents separate from documents used for construction.
- D. Record information concurrent with construction progress, not less than weekly.
- E. Specifications: Legibly mark and record, at each product Section, description of actual products installed, including the following:
  - 1. Manufacturer's name and product model and number.
  - 2. Product substitutions or alternates used.
  - 3. Changes made by Addenda and modifications.
- F. Record Drawings and Shop Drawings: Legibly mark each item to record actual construction as follows:
  - 1. Include Contract modifications such as Addenda, supplementary instructions, change directives, field orders, minor changes in the Work, and change orders.
  - 2. Include locations of concealed elements of the Work.
  - 3. Identify depth of buried utility lines and provide dimensions showing distances from permanent facility components that are parallel to utilities.
  - 4. Dimension ends, corners, and junctions of buried utilities to permanent facility components using triangulation.
  - 5. Identify and locate existing buried or concealed items encountered during Project.
  - 6. Measured depths of foundations in relation to finish first floor datum.
  - 7. Measured horizontal and vertical locations of underground utilities and appurtenances, referenced to permanent surface improvements.
  - 8. Measured locations of internal utilities and appurtenances concealed in construction, referenced to visible and accessible features of the Work.
  - 9. Field changes of dimension and detail.
  - 10. Details not on original Drawings.
- G. Submit marked-up paper copy documents to Engineer with claim for final Application for Payment.
- H. Submit PDF electronic files of marked-up documents to Engineer with claim for final Application for Payment.

051.18.2.22

## 1.8 OPERATION AND MAINTENANCE DATA

A. Submit in PDF composite electronic indexed file.

- B. Submit data bound in 8-1/2 x 11-inch text pages, three D side ring capacity expansion binders with durable plastic covers.
- C. Prepare binder cover with printed title "OPERATION AND MAINTENANCE INSTRUCTIONS," title of Project, and subject matter of binder when multiple binders are required.
- D. Internally subdivide binder contents with permanent page dividers, logically organized as described below; with tab titling clearly printed under reinforced laminated plastic tabs.
- E. Drawings: Provide with reinforced punched binder tab. Bind in with text; fold larger drawings to size of text pages.
- F. Contents: Prepare table of contents for each volume, with each product or system description identified, typed on white paper, in three parts as follows:
  - 1. Part 1: Directory, listing names, addresses, and telephone numbers of Engineer, Contractor, Subcontractors, and major equipment suppliers.
  - 2. Part 2: Operation and maintenance instructions, arranged by system and subdivided by Specification Section. For each category, identify names, addresses, and telephone numbers of Subcontractors and suppliers. Include the following:
    - a. Significant design criteria.
    - b. List of equipment.
    - c. Parts list for each component.
    - d. Operating instructions.
    - e. Maintenance instructions for equipment and systems.
    - f. Maintenance instructions for finishes, including recommended cleaning methods and materials, and special precautions identifying detrimental agents.
    - g. Safety precautions to be taken when operating and maintaining or working near equipment.
  - 3. Part 3: Project documents and certificates, including the following:
    - a. Shop Drawings and product data.
    - b. Air and water balance reports.
    - c. Certificates.
    - d. Originals of warranties and bonds.

# 1.9 MANUAL FOR MATERIALS AND FINISHES

- A. Submit two copies of preliminary draft or proposed formats and outlines of contents before start of Work. Engineer will review draft and return one copy with comments.
- B. For equipment or component parts of equipment put into service during construction and operated by Owner, submit documents within ten days after acceptance.
- C. Submit one copy of completed volumes before Substantial Completion. Draft copy be reviewed and returned after Substantial Completion, with Engineer comments. Revise content of document sets as required prior to final submission.

- D. Submit two sets of revised final volumes within ten days after final inspection.
- E. Submit in PDF composite electronic indexed file of final manual within ten days after final inspection.
- F. Building Products, Applied Materials, and Finishes: Include product data, with catalog number, size, composition, and color and texture designations. Include information for re-ordering custom-manufactured products.
- G. Instructions for Care and Maintenance: Include manufacturer's recommendations for cleaning agents and methods, precautions against detrimental agents and methods, and recommended schedule for cleaning and maintenance.
- H. Moisture Protection and Weather Exposed Products: Include product data listing applicable reference standards, chemical composition, and details of installation. Include recommendations for inspections, maintenance, and repair.
- I. Additional Requirements: As specified in individual product Specification Sections.
- J. Include listing in table of contents for design data, with tabbed fly sheet and space for insertion of data.

# 1.10 MANUAL FOR EQUIPMENT AND SYSTEMS

- A. Submit two copies of preliminary draft or proposed formats and outlines of contents before start of Work. Engineer will review draft and return one copy with comments.
- B. For equipment, or component parts of equipment put into service during construction and operated by Owner, submit documents within ten days after acceptance.
- C. Submit one copy of completed volumes before Substantial Completion Draft copy will be reviewed and returned after Substantial Completion, with Engineer comments. Revise content of document sets as required prior to final submission.
- D. Submit two sets of revised final volumes within ten days after final inspection.
- E. Submit in PDF composite electronic indexed file of final manual within ten days after final inspection.
- F. Each Item of Equipment and Each System: Include description of unit or system and component parts. Identify function, normal operating characteristics, and limiting conditions. Include performance curves, with engineering data and tests, and complete nomenclature and model number of replaceable parts.
- G. Panelboard Circuit Directories: Provide electrical service characteristics, controls, and communications.

051.18.2.22

H. Include color-coded wiring diagrams as installed.

- I. Operating Procedures: Include startup, break-in, and routine normal operating instructions and sequences. Include regulation, control, stopping, shutdown, and emergency instructions. Include summer, winter, and special operating instructions.
- J. Maintenance Requirements: Include routine procedures and guide for preventative maintenance and troubleshooting; disassembly, repair, and reassembly instructions; and alignment, adjusting, balancing, and checking instructions.
- K. Include servicing and lubrication schedule and list of lubricants required.
- L. Include manufacturer's printed operation and maintenance instructions.
- M. Include sequence of operation by controls manufacturer.
- N. Include original manufacturer's parts list, illustrations, assembly drawings, and diagrams required for maintenance.
- O. Include control diagrams by controls manufacturer as installed.
- P. Include Contractor's coordination drawings with color-coded piping diagrams as installed.
- Q. Include charts of valve tag numbers, with location and function of each valve, keyed to flow and control diagrams.
- R. Include list of original manufacturer's spare parts, current prices, and recommended quantities to be maintained in storage.
- S. Include test and balancing reports as specified in Section 014000 Quality Requirements.
- T. Additional Requirements: As specified in individual product Specification Sections.
- U. Include listing in table of contents for design data with tabbed dividers and space for insertion of data.

# 1.11 SPARE PARTS AND MAINTENANCE PRODUCTS

- A. Furnish spare parts, maintenance, and extra products in quantities specified in individual Specification Sections.
- B. Deliver to place in location as directed by Engineer; obtain receipt prior to final payment.

# 1.12 PRODUCT WARRANTIES AND PRODUCT BONDS

- A. Obtain warranties and bonds executed in duplicate by responsible Subcontractors, suppliers, and manufacturers within ten days after completion of applicable item of Work.
- B. Execute and assemble transferable warranty documents and bonds from Subcontractors, suppliers, and manufacturers.

051.18.2.22

C. Verify documents are in proper form, contain full information, and are notarized.

- D. Co-execute submittals when required.
- E. Include table of contents and assemble in three D side ring binder with durable plastic cover.
- F. Submit prior to final Application for Payment.
- G. Time of Submittals:
  - 1. For equipment or component parts of equipment put into service during construction with Owner's permission, submit documents within ten days after acceptance.
  - 2. Make other submittals within ten days after date of Substantial Completion, prior to final Application for Payment.
  - 3. For items of Work for which acceptance is delayed beyond Substantial Completion, submit within ten days after acceptance, listing date of acceptance as beginning of warranty or bond period.

#### 1.13 MAINTENANCE SERVICE

- A. Furnish service and maintenance of components indicated in Specification Sections during warranty period.
- B. Examine system components at frequency consistent with reliable operation. Clean, adjust, and lubricate as required.
- C. Include systematic examination, adjustment, and lubrication of components. Repair or replace parts whenever required. Use parts produced by manufacturer of original component.
- D. Do not assign or transfer maintenance service to agent or Subcontractor without prior written consent of Owner.

# PART 2 - PRODUCTS - (NOT USED)

## **PART 3 - EXECUTION**

## 3.1 EXAMINATION

- A. Verify that existing Site conditions and substrate surfaces are acceptable for subsequent Work. Beginning new Work means acceptance of existing conditions.
- B. Verify that existing substrate is capable of structural support or attachment of new Work being applied or attached.
- C. Examine and verify specific conditions described in individual Specification Sections.
- D. Verify that utility services are available with correct characteristics and in correct locations.

# 3.2 PREPARATION

- A. Clean substrate surfaces prior to applying next material or substance according to manufacturer's instructions.
- B. Seal cracks or openings of substrate prior to applying next material or substance.
- C. Apply manufacturer-required or -recommended substrate primer, sealer, or conditioner prior to applying new material or substance in contact or bond.

# 3.3 EXECUTION

- A. Comply with manufacturer's installation instructions, performing each step in sequence. Maintain one set of manufacturer's installation instructions at Project Site during installation and until completion of construction.
- B. When manufacturer's installation instructions conflict with Contract Documents, request clarification from Engineer before proceeding.
- C. Verify that field measurements are as indicated on approved Shop Drawings or as instructed by manufacturer.
- D. Secure products in place with positive anchorage devices designed and sized to withstand stresses, vibration, physical distortion, or disfigurement.
  - 1. Secure Work true to line and level and within specified tolerances, or if not specified, industry-recognized tolerances.
  - 2. Physically separate products in place, provide electrical insulation, or provide protective coatings to prevent galvanic action or corrosion between dissimilar metals.
  - 3. Exposed Joints: Provide uniform joint width and arrange to obtain best visual effect. Refer questionable visual-effect choices to Engineer for final decision.
- E. Allow for expansion of materials and building movement.
- F. Climatic Conditions and Project Status: Install each unit of Work under conditions to ensure best possible results in coordination with entire Project.
  - 1. Isolate each unit of Work from incompatible Work as necessary to prevent deterioration.
  - 2. Coordinate enclosure of Work with required inspections and tests to minimize necessity of uncovering Work for those purposes.
- G. Mounting Heights: Where not indicated, mount individual units of Work at industry recognized standard mounting heights for particular application indicated.
  - 1. Refer questionable mounting heights choices to Engineer for final decision.
  - 2. Elements Identified as Accessible to Handicapped: Comply with applicable codes and regulations.
- H. Adjust operating products and equipment to ensure smooth and unhindered operation.

I. Clean and perform maintenance on installed Work as frequently as necessary through remainder of construction period. Lubricate operable components as recommended by manufacturer.

# 3.4 CUTTING AND PATCHING

- A. Employ skilled and experienced installers to perform cutting and patching.
- B. Submit written request in advance of cutting or altering elements affecting:
  - 1. Structural integrity of element.
  - 2. Integrity of weather-exposed or moisture-resistant elements.
  - 3. Efficiency, maintenance, or safety of element.
  - 4. Visual qualities of sight-exposed elements.
  - 5. Work of Owner or separate contractor.
- C. Execute cutting, fitting, and patching including excavation and fill to complete Work and to:
  - 1. Fit the several parts together, to integrate with other Work.
  - 2. Uncover Work to install or correct ill-timed Work.
  - 3. Remove and replace defective and nonconforming Work.
  - 4. Remove samples of installed Work for testing.
  - 5. Provide openings in elements of Work for penetrations of mechanical and electrical Work.
- D. Execute Work by methods to avoid damage to other Work and to provide proper surfaces to receive patching and finishing.
- E. Cut masonry and concrete materials using masonry saw or core drill.
- F. Restore Work with new products according to requirements of Contract Documents.
- G. Fit Work tight to pipes, sleeves, ducts, conduits, and other penetrations through surfaces.
- H. Maintain integrity of wall, ceiling, or floor construction; completely seal voids.
- I. At penetrations of fire-rated walls, partitions, ceiling, or floor construction, completely seal voids with fire-rated material to full thickness of penetrated element.
- J. Refinish surfaces to match adjacent finishes. For continuous surfaces, refinish to nearest intersection; for assembly, refinish entire unit.
- K. Identify hazardous substances or conditions exposed during the Work to Engineer for decision or remedy.

## 3.5 PROTECTING INSTALLED CONSTRUCTION

- A. Protect installed Work and provide special protection where specified in individual Specification Sections.
- B. Provide temporary and removable protection for installed products. Control activity in immediate Work area to prevent damage.

- C. Provide protective coverings at walls, projections, jambs, sills, and soffits of openings.
- D. Use durable sheet materials to protect finished floors, stairs, and other surfaces from traffic, dirt, wear, damage, or movement of heavy objects.
- E. Prohibit traffic or storage upon waterproofed or roofed surfaces. When traffic or activity is necessary, obtain recommendations for protection from waterproofing or roofing material manufacturer.
- F. Prohibit traffic from landscaped areas.

# 3.6 FINAL CLEANING

- A. Execute final cleaning prior to final Project assessment.
  - 1. Employ experienced personnel or professional cleaning firm.
- B. Clean interior and exterior glass and surfaces exposed to view; remove temporary labels, stains, and foreign substances; polish transparent and glossy surfaces.
- C. Clean equipment and fixtures to sanitary condition with appropriate cleaning materials.
- D. Replace filters of operating equipment.
- E. Clean debris from roofs, gutters, downspouts, and drainage systems.
- F. Clean Site; sweep paved areas, rake clean landscaped surfaces.
- G. Remove waste and surplus materials, rubbish, and construction facilities from Site.

# **END OF SECTION 017000**

# SECTION 017123 CONSTRUCTION LAYOUT

# **PART 1 - GENERAL**

#### 1.1 SCOPE

- A. Construction staking shall include all of the surveying work required to layout the work and control the location of the finished Project. The Contractor shall have the full responsibility for constructing the Project to the correct horizontal and vertical alignment, as shown on the Drawings, as specified, or as ordered by the Engineer. The Contractor shall assume all costs associated with rectifying work constructed in the wrong location.
- B. From the information shown on the Drawings and the information to be provided as indicated under Project Conditions below, the Contractor shall:
  - 1. Be responsible for setting reference points and/or offsets, establishment of baselines, and all other layout, staking, and all other surveying required for the construction of the Project.
  - 2. Safeguard all reference points, stakes, grade marks, horizontal and vertical control points, and shall bear the cost of re-establishing same if disturbed.
  - 3. Stake out the permanent and temporary easements or the limits of construction to ensure that the work is not deviating from the indicated limits.
  - 4. Be responsible for all damage done to reference points, baselines, center lines and temporary bench marks, and shall be responsible for the cost of re-establishment of reference points, baselines, center lines and temporary bench marks as a result of the operations.
- C. Baselines shall be defined as the line to which the location of the work is referenced, i.e., edge of pavement, road centerline, property line, right-of-way or survey line.
- D. Record Drawing surveys shall be performed in accordance with Section 017000 of these Specifications.

# 1.2 PROJECT CONDITIONS

- A. The Drawings provide the location and/or coordinates of principal components of the Project. The alignment of some components of the Project may be indicated in the Specifications. The Engineer may order changes to the location of some of the components of the Project or provide clarification to questions regarding the correct alignment.
- B. The survey points, control points, and baseline to be provided to the Contractor shall be limited to only that information which can be found on the Project site by the Contractor.

# 1.3 QUALITY ASSURANCE

- A. The Contractor shall furnish documentation, prepared by a surveyor currently registered in the State in which the Project is located, confirming that staking is being done to the horizontal and vertical alignment shown in the Contract Documents. This requires that the Contractor hire, at the Contractor's own expense, a currently registered surveyor, acceptable to the Owner, to provide ongoing construction staking or confirmation of such.
- B. Any deviations from the Drawings shall be confirmed by the Engineer prior to construction of that portion of the Project.

## 1.4 PRESSURE MAINS AND ACCESSORIES

A. Staking Precision: The precision of construction staking required shall be that which the correct location of the main can be established for construction and verified by the Engineer. Where the location of components of the main, e.g. fittings, valves, road crossings and are not dimensioned, the establishment of the location of these components shall be based upon scaling these locations from the Drawings with relation to readily identifiable land marks, e.g., survey reference points, power poles, manholes, etc.

## B. Reference Points

- 1. Reference points shall be placed, at or no more than three feet, from the outside of the construction easement or right-of-way. The location of the reference points shall be recorded in a log with a copy provided to the Engineer for use, prior to verifying reference point locations. Distances shall be accurately measured to 0.01 foot.
- 2. The Contractor shall give the Engineer reasonable notice that reference points are set. The reference point locations must be verified by the Engineer prior to commencing clearing and grubbing operations.

**END OF SECTION 017123** 

# SECTION 312333 TRENCHING AND BACKFILLING

### **PART 1 - GENERAL**

### 1.1 SECTION INCLUDES

- A. Clearing and grubbing
- B. Excavation and disposal of all wet and dry materials (including rock) encountered that must be removed for construction purposes
- C. Dewatering of trenches and other excavations
- D. Pipe bedding
- E. Backfilling and tamping of trenches, foundations, and other structures

# 1.2 RELATED SECTIONS

A. Section 331413 – Public Water Utility Distribution Piping

# 1.3 **DEFINITIONS**

- A. Degree of Compaction: Degree of compaction is expressed as a percentage of the maximum density obtained by the test procedure presented in ASTM D698, for general soil types, abbreviated as percent laboratory maximum density.
- B. Hard Materials: Weathered rock, dense consolidated deposits, or conglomerate materials which are not included in the definition of "rock" but which usually require the use of heavy excavation equipment, ripper teeth, or jack hammers for removal.
- C. Rock: Solid homogeneous interlocking crystalline material with firmly cemented, laminated, or foliated masses or conglomerate deposits, neither of which can be removed without systematic drilling and blasting, drilling and the use of expansion jacks or feather wedges, or the use of backhoe-mounted pneumatic hole punchers or rock breakers; also large boulders, buried masonry, or concrete other than pavement.

### 1.4 SUBMITTALS

A. The following shall be submitted in accordance with Section 013300 - Submittal Procedures:

- B. Test Reports Submit copies of all laboratory and field test reports within 24 hours of the completion of the test.
  - 1. Borrow Site Testing: Fill and backfill test

- 2. Select material test
- 3. Porous fill test for capillary water barrier
- 4. Density tests

### 1.5 DELIVERY, STORAGE, AND HANDLING

A. Perform in a manner to prevent contamination or segregation of materials.

# 1.6 REQUIREMENTS FOR OFF SITE SOIL

Soils brought in from off site for use as backfill shall be tested for petroleum hydrocarbons, A. BTEX, PCBs and HW characteristics (including toxicity, ignitability, corrosivity, and reactivity). Backfill shall not contain concentrations of these analytes above the appropriate State and/or EPA criteria, and shall pass the tests for HW characteristics. Determine petroleum hydrocarbon concentrations by using appropriate State protocols. Determine BTEX concentrations by using EPA SW- 846.3-3 Method 5035/82608. Perform complete TCLP in accordance with EPA SW-846.3-3 Method 1311. Perform HW characteristic tests for ignitability, corrosivity, and reactivity in accordance with accepted standard methods. Perform PCB testing in accordance with accepted standard methods for sampling and analysis of bulk solid samples. Provide borrow site testing for petroleum hydrocarbons and BTEX from a grab sample of material from the area most likely to be contaminated at the borrow site (as indicated by visual or olfactory evidence), with at least one test from each borrow site. For each borrow site, provide borrow site testing for HW characteristics from a composite sample of material, collected in accordance with standard soil sampling techniques. Do not bring material onsite until tests results have been received and approved by the Owner.

### 1.7 FIELD MEASUREMENTS

A. Verify that survey bench mark and intended elevations for the Work are as shown on the drawings.

### 1.8 COORDINATION

A. Verify work associated with lower elevation utilities is complete before placing higher elevation utilities.

### 1.9 QUALITY ASSURANCE

A. Utilities: Movement of construction machinery and equipment over pipes and utilities during construction shall be at the Contractor's risk Excavation made with power- driven equipment is not permitted within two feet of known utility or subsurface construction. For work immediately adjacent to or for excavations exposing a utility or other buried obstruction, excavate by hand. Start hand excavation on each side of the indicated obstruction and continue until the obstruction is uncovered or until clearance for the new grade is assured. Support uncovered lines or other

existing work affected by the contract excavation until approval for backfill is granted by the Engineer. Report damage to utility lines or subsurface construction immediately to the Engineer.

### **PART 2 - PRODUCTS**

### 2.1 SOIL MATERIALS

- A. Satisfactory Materials: Any materials classified by ASTM D2487 as GW, GP, GM, GP-GM, GW-GM, GC, GP-GC, GM-GC, SW, or SP, free of debris, roots, wood, scrap material, vegetation, refuse, soft unsound particles, and frozen, deleterious, or objectionable materials. Unless specified otherwise, the maximum particle diameter shall be one-half the lift thickness at the intended location.
- B. Unsatisfactory Materials: Materials which do not comply with the requirements for satisfactory materials. Unsatisfactory materials also include man-made fills, trash, refuse, or backfills from previous construction. Unsatisfactory material also includes material classified as satisfactory which contains root and other organic matter, frozen material, and stones larger than 3 inches. The Engineer shall be notified of any contaminated materials.
- C. Backfill and Fill Material: ASTM D2487, classification GW, GP, GM, SW, SP, or SM, non-plastic, and a maximum of 25 percent by weight passing ASTM D1140, No. 200 sieve.

#### 2.2 BORROW

A. Obtain borrow materials required in excess of those furnished from excavations from sources outside of Owner's property.

### **PART 3 - EXECUTION**

### 3.1 PROTECTION

# A. Shoring and Sheeting

- 1. Take special care to avoid damage wherever excavation is being done. Sufficiently sheet, shore, and brace the sides of all excavations to prevent slides, cave-ins, settlement, or movement of the banks and to maintain the specified trench widths. Use solid sheets in wet, saturated, or flowing ground. All sheeting, shoring, and bracing shall have enough strength and rigidity to withstand the pressures exerted, to keep the walls of the excavation properly in place, and to protect all persons and property from injury or damage. Separate payment will not be made for sheeting, shoring, and bracing, which are considered an incidental part of the excavation work.
- 2. Wherever employees may be exposed to moving ground or cave-ins, shore and lay back exposed earth excavation surfaces more than 5 feet high to a stable slope, or else provide some equivalent means of protection. Effectively protect trenches less than 5 feet deep when examination of the ground indicates hazardous ground movement may be expected.

Guard the walls and faces of all excavations in which employees are exposed to danger from moving ground by a shoring system, sloping of the ground, or some equivalent protection.

- 3. Trench excavation safety protection shall be accomplished as required by the most recent provisions of Part 1926, Subpart P Excavations, Trenching, and Shoring of the Occupational Safety and Health Administration (OSHA) Standards and Interpretations, as may be amended. Comply with all OSHA standards in determining where and in what manner sheeting, shoring, and bracing are to be done. The sheeting, shoring, and bracing system shall be designed by a professional engineer licensed in the State of Georgia and shall be subject to approval by the Engineer. However, such approval does not relieve the Contractor of the sole responsibility for the safety of all employees, the effectiveness of the system, and any damages or injuries resulting from the lack or inadequacy of sheeting, shoring, and bracing.
- 4. Where excavations are made adjacent to existing buildings or structures or in paved streets or alleys, take particular care to sheet, shore, and brace the sides of the excavation so as to prevent any undermining of or settlement beneath such structures or pavement. Underpin adjacent structures wherever necessary, with the approval of the Engineer.
- 5. Do not leave sheeting, shoring, or bracing materials in place unless this is called for by the Drawings, ordered by the Engineer, or deemed necessary or advisable for the safety or protection of the new or existing work or features. Remove these materials in such a manner that the new structure or any existing structures or property, whether public or private, will not be endangered or damaged and that cave-ins and slides are avoided.
- 6. Fill and compact all holes and voids left in the work by the removal of sheeting, shoring, or bracing as specified herein.
- 7. The Contractor may use a trench box, which is a prefabricated movable trench shield composed of steel plates welded to a heavy steel frame. The trench box shall be designed to provide protection equal to or greater than that of an appropriate shoring system.
- 8. A "Qualified Person", as defined by OSHA regulations, shall be on-site at all times during activities requiring trench safety provisions.

# B. Drainage and Dewatering

- 1. Provide for the collection and disposal of surface and subsurface water encountered during construction.
- 2. Drainage: So that construction operations progress successfully, completely drain construction site during periods of construction to keep soil materials sufficiently dry.: Where applicable, the Contractor shall establish/construct storm drainage features (ponds/basins) at the earliest stages of site development, and throughout construction grade the construction area to provide positive surface water runoff away from the construction activity and/or provide temporary ditches, swales, and other drainage features and equipment as required to maintain dry soils. When unsuitable working platforms for equipment operation and unsuitable soil support for subsequent construction features develop, remove unsuitable material and provide new soil material as specified herein. It is the responsibility of the Contractor to assess the soil and ground water conditions

presented by the plans and specifications and to employ necessary measures to permit construction to proceed. Excavated slopes and backfill surfaces shall be protected to prevent erosion and sloughing. Excavation shall be performed so that the site, the area immediately surrounding the site, and the area affecting operations at the site shall be continually and effectively drained.

# 3. Dewatering:

- a. Dewatering Trenches: Where quicksand is encountered, dewatering shall be done by well pointing. If soil conditions are not suitable for well points, trench drains or crushed stone or gravel shall be constructed to carry the water to sumps and the water removed by pumps or bailing.
- b. Groundwater flowing toward or into excavations shall be controlled to prevent sloughing of excavation slopes and walls, boils, uplift and heave in the excavation and to eliminate interference with orderly progress of construction. French drains, sumps, ditches or trenches will not be permitted within 3 feet of the foundation of any structure, except with specific written approval, and after specific contractual provisions for restoration of the foundation area have been made. Control measures shall be taken by the time the excavation reaches the water level in order to maintain the integrity of the in situ material. While the excavation is open, the water level shall be maintained continuously, at least 2 feet below the working level.
- c. Operate dewatering system continuously until construction work below existing water levels is complete. Submit performance records weekly.

# C. Underground Utilities

- 1. Location of the existing utilities indicated is approximate. The Contractor shall physically verify the location and elevation of all existing utilities prior to starting construction. The Contractor shall contact the State One-Call Service affected utilities for assistance in locating existing utilities.
- D. Machinery and Equipment: Movement of construction machinery and equipment over pipes during construction shall be at the Contractor's risk. Repair, or remove and provide new pipe for existing or newly installed pipe that has been displaced or damaged.

#### 3.2 SURFACE PREPARATION

- A. Clear and grub project area in accordance with Section 311000 Site Clearing.
- B. Identify required lines, levels, contours, and datum.
- C. Protect plant life, lawns, and other features remaining as part of final landscaping.

051.18.2.22

D. Maintain and protect above and below grade utilities which are to remain.

### 3.3 EXCAVATION

- A. Excavate to contours, elevation, and dimensions indicated. Reuse excavated materials that meet the specified requirements for the material type required at the intended location. Keep excavations free from water. Excavate soil disturbed or weakened by Contractor's operations, soils softened or made unsuitable for subsequent construction due to exposure to weather. Excavations below indicated depths will not be permitted except to remove unsatisfactory material.
- B. Wherever muck, quicksand, soft clay, swampy ground, or other material unsuitable for foundations, subgrade, or backfilling is encountered, remove it and continue excavation until suitable material is encountered. The material removed shall be disposed of in the manner described below. Then refill the areas excavated for this reason with 1 inch to 2 inch sized crushed stone up to the level of the lines, grades, and/or cross sections shown on the Drawings. The top 6 inches of this refill shall meet the requirements for bedding material.
- C. Unless specified otherwise, refill excavations cut below indicated depth with backfill and fill material and compact to 95 percent of ASTM D698 maximum density. Satisfactory material removed below the depths indicated, without specific direction of the Engineer, shall be replaced with satisfactory materials to the indicated excavation grade. Determination of elevations and measurements of approved overdepth excavation of unsatisfactory material below grades indicated shall be done under the direction of the Engineer.

### D. Pipe Trenches:

- 1. Unless the construction of lines by tunneling, jacking, or boring is called for by the Drawings or specifically authorized by the Engineer, make excavation for pipelines in open cut and true to the lines and grades shown on the Drawings or established by the Engineer on the ground. Cut the banks of trenches between vertical parallel planes equidistant from the pipe centerline. The trench shall be dug to the alignment and grade required. The trench shall be braced if necessary and drained in order that workmen therein may work safely and efficiently. It is essential that the discharge from any pumps be led to natural drainage channels or to drains. Minimum width of trench shall be six inches (6") from the outside of barrel of pipe on each side of pipe. Maximum width of trench shall be nine inches (9") from the outside of barrel of pipe on each side of pipe. Sides of trench shall be dug and maintained substantially vertical except Contractor shall adhere to all applicable requirements of OSHA for trench safety.
- 2. Grade bottom of trenches to provide uniform support for each section of pipe after pipe bedding placement. Tamp if necessary to provide a firm pipe bed. Recesses shall be excavated to accommodate bells and joints so that pipe will be uniformly supported for the entire length. Rock, where encountered, shall be excavated to a depth of at least 6 inches below the bottom of the pipe. Any part of the trench excavated below grade shall be corrected with approved material thoroughly compacted. Wherever the material at the bottom of the trench is unsuitable for the proper installation of the pipe, the City will direct the removal and replacement of the unsuitable material. When so directed, undercut the trench and backfill with No. 57 stone meeting the requirements of Georgia Department of Transportation Specification 800.01. Place and compact this material to bring the trench to the required grade.

- 3. Excavate bell holes for bell and spigot pipe at proper intervals so that the barrel of the pipe will rest for its entire length upon the bottom of the trench. Bell holes shall be large enough to permit proper jointing of the pipe. Do not excavate bell holes more than 2 joints ahead of pipe laying.
- 4. For all pipe other than ductile iron pipe, provide minimum of four inches of "Bedding Material", unless otherwise shown or specified.
- 5. Do not excavate pipe trenches more than 200 feet ahead of the pipe laying, and perform all work so as to cause the least possible inconvenience to the public. Construct temporary bridges or crossings when and where the Engineer deems necessary to maintain vehicular or pedestrian traffic.
- 6. In all cases where materials are deposited along open trenches, place them so that in the event of rain no damage will result to the work and/or to adjacent property.

### E. Hard Material and Rock

- 1. Any material that is encountered within the limits of the required excavation that cannot be removed except by drilling and/or blasting, including rock, boulders, masonry, hard pan, chert, shale, street and sidewalk pavements, and/or similar materials, shall be considered as unclassified excavation, and no separate payment will be made therefore.
- 2. See Section 312316.26 Rock Removal for excavation of rock.

#### F. Excavated Materials

 Satisfactory excavated material required for fill or backfill shall be placed in the proper section of the permanent work required or shall be separately stockpiled if it cannot be readily placed. Satisfactory material in excess of that required for the permanent work and all unsatisfactory material shall be disposed of as specified in Paragraph "DISPOSITION OF SURPLUS MATERIAL."

### 3.4 FILLING AND BACKFILLING

- A. Fill and backfill to contours, elevations, and dimensions indicated. Compact each lift before placing overlaying lift.
- B. Backfill and Fill Material Placement For Utilities
  - 1. Erosion Control: Plan excavation work to prevent erosion and the washing of soil into adjacent streams. Limit the amount of open excavation at any one time. Place spoil in the proper place and keep natural water routes open. All erosion control procedures must comply with the Georgia Erosion and Sedimentation Control Act.
  - 2. Time of Backfilling: As soon as practicable after the completion of laying and jointing of the pipe, the trench shall be backfilled and at no time shall the completed backfilled trench be more than 200 feet behind the pipe laying.

- 3. Backfill Procedure at Pipe Zone: Haunching material shall be deposited in the trench simultaneously on both sides of the pipe for the full width of the trench to an elevation as shown on the detail for type pipe selected. Initial backfill material shall be moistened if necessary, tamped in thin six inch (6") layers and thoroughly compacted on each side of the pipe as shown on the Typical Sewer Pipe Bedding Detail. Walking or working on the completed pipe line, except as may be necessary in tamping or backfilling, will not be permitted until the trench has been compacted or a height of at least two-and-one-half feet (2 ½") over the barrel of the pipe for PVC pipe and two feet (2") over the barrel of the pipe for ductile iron (DI) pipe.
- 4. Compaction Requirements: All backfill for trenches under paved or roadway areas shall be compacted to 98 percent of the Standard Proctor maximum density (ASTM D-698). Backfill for trenches other than under pavement areas shall be compacted to 95 percent of the Standard Proctor maximum density (ASTM D-698) to eighteen inches (18") above the crown of the pipe. The remainder shall be compacted to 93 percent of the Standard Proctor maximum density (ASTM D-698). The Contractor shall furnish the City proof of compaction at a point eighteen inches (18") above the crown of the sewer pipe at all pavement, driveway or roadway crossings. Additional compaction tests are required at each crossing eight inches (8") below the surface of the pavement or on the pavement subgrade before pavement placement. Contractor shall incur all costs for compaction tests performed by a testing laboratory approved by the City and selected by the Contractor.
- 5. Rock and Rock Fragment Exclusion: No rock or rock fragments shall be used in the backfill for at least eighteen inches (18") above the top of the pipe and no stone larger than eight inches (8") in its greatest dimension shall be used in the backfill.
- 6. Deficiency of Backfill: Any deficiency in the quantity of material for backfilling the trenches or for filling depressions caused by settlement shall be supplied by the Contractor.

#### 3.5 BORROW

- A. Whenever the backfill of excavated areas or the placement of embankments requires more material than is available from authorized excavations, or whenever the backfill material from such excavations is unsuitable, then obtain additional material from other sources. This may require the opening of borrow pits at points accessible to the work. In such cases, make suitable arrangements with the property owner and pay all incidental costs, including any royalties, for the use of the borrowed material. Before a borrow pit is opened, the quality and suitability of its material shall be approved by the Engineer.
- B. Excavate borrow pits in such a way that the remaining surfaces and slopes are reasonably smooth and that adequate drainage is provided over the entire area. Construct drainage ditches wherever necessary to provide outlets for water to the nearest natural channel, thus preventing the formation of pools in the pit area. Leave the sides of borrow pit cuts at a maximum slope of 2:1 unless otherwise directed by the Engineer.
- C. Properly clear and grub borrow pits and remove all objectionable matter from the borrow pit material before placing it in the backfill.

D. The taking of materials from borrow pits for use in the construction of backfill, fills, or embankments shall be considered an incidental part of the work; no separate payment shall be made for this.

#### 3.6 FINISH OPERATIONS

- A. Grading: Finish grades as indicated within one-tenth of one foot. Grade areas to drain water away from structures. Maintain areas free of trash and debris. For existing grades that will remain but which were disturbed by Contractor's operations, grade as directed.
- B. Protection of Surfaces: Protect newly backfilled, graded, and topsoiled areas from traffic, erosion, and settlements that may occur. Repair or reestablish damaged grades, elevations, or slopes.
- C. Restoration of Surfaces: The Contractor shall replace all curbing, sidewalks, pavements, gutters, shrubbery, fences, sod and other surfaces disturbed to a condition equal to that before the work began, furnishing all labor and materials incidental thereto.

### 3.7 DISPOSITION OF SURPLUS MATERIAL

- A. Whenever practicable, all materials removed by excavation that are suitable for backfilling pipe trenches or for other purposes shown on the Drawings or directed by the Engineer shall be used for these purposes. Any materials not so used shall be considered waste materials and disposed of by the Contractor as specified below.
- B. Once any part of the work is completed, properly dispose of all surplus or unused materials (including waste materials) left within the construction limits of that work. The Contractor shall dispose of these surplus and waste materials off-site in an appropriate manner in conformity with pertinent codes and ordinances. Leave the surface of the work in a neat and workmanlike condition, as described below.
- C. The disposal of waste materials shall be considered an integral part of the excavation work and one for which no separate payment shall be allowed.

### 3.8 FIELD QUALITY CONTROL

- A. Sampling: Take the number and size of samples required to perform the following tests.
- B. Testing: Perform one of each of the following tests for each material used. Provide additional tests for each source change.
  - Bedding Material and Fill and Backfill Material Testing: Test fill and backfill material in accordance with ASTM C136 for conformance to ASTM D2487 gradation limits; ASTM D1140 for material finer than the No. 200 sieve; ASTM D4318 for liquid limit and for plastic limit; ASTM D698 or ASTM D1557 for moisture density relations, as applicable.
  - Density Tests: Test density in accordance with ASTM D1556, or ASTM D6938. When ASTM D6938 density tests are used, verify density test results by performing an ASTM D1556 density test at a location already ASTM D6938 tested as specified herein. Perform

an ASTM D1556 density test at the start of the job, and for every 10 ASTM D6938 density tests thereafter. Test each lift at randomly selected locations with one test per 400 linear feet in each lift.

# **END OF SECTION 312323**

# SECTION 330110 DISINFECTION OF WATER UTILITY PIPING SYSTEMS

#### PART 1 - GENERAL

### 1.1 SUMMARY

- A. Section Includes:
  - 1. Disinfection of potable water distribution and transmission system.
  - 2. Testing and reporting of results.
- B. Related Requirements:
  - 1. Section 331413 Public Water Utility Distribution Piping: Product and execution requirements for installation and testing of public domestic water distribution piping.

### 1.2 UNIT PRICE - MEASUREMENT AND PAYMENT

A. Section 012000 – Price and Payment Procedures: Contract Sum Price/Modification Procedures

### 1.3 REFERENCE STANDARDS

- A. American Water Works Association:
  - 1. AWWA B300 Hypochlorites.
  - 2. AWWA B302 Ammonium Sulfate.
  - 3. AWWA B303 Sodium Chlorite.
  - 4. AWWA C651 Disinfecting Water Mains.

### 1.4 SUBMITTALS

- A. Section 013300 Submittal Procedures: Requirements for submittals.
- B. Before disinfection is performed, the Contractor shall submit a written procedure for approval before being permitted to proceed with the disinfection. This plan shall also include the steps to be taken for the neutralization of the chlorinated water.
- C. Disinfection Procedure:
  - 1. Submit description of procedure, including type of disinfectant and calculations indicating quantities of disinfectants required to produce specified chlorine concentration.
- D. Product Data: Submit manufacturer information for proposed chemicals and treatment doses.

051.18.2.22

E. Manufacturer's Certificate: Certify that products meet or exceed specified requirements.

- F. Certify that final water complies with disinfectant quality standards of authority having jurisdiction.
- G. Test and Evaluation Reports: Indicate testing results comparative to specified requirements.
- H. Field Quality-Control Submittals: Indicate results of Contractor-furnished tests and inspections.
- I. Qualifications Statements:
  - 1. Submit qualifications for manufacturer and applicator.

### 1.5 CLOSEOUT SUBMITTALS

- A. Section 017000 Execution and Closeout Requirements: Requirements for submittals.
- B. Disinfection Report:
  - 1. Type and form of disinfectant used.
  - 2. Date and time of disinfectant injection start and completion.
  - 3. Test locations.
  - 4. Name of person collecting samples.
  - 5. Initial and 24-hour disinfectant residuals in treated water in ppm for each outlet tested.
  - 6. Date and time of flushing start and completion.
  - 7. Disinfectant residual after flushing in ppm for each outlet tested.

# 1.6 QUALITY ASSURANCE

- A. Perform Work according to AWWA C651.
- B. Maintain one copy of each standard affecting Work of this Section on Site.

### 1.7 QUALIFICATIONS

- A. Manufacturer: Company specializing in manufacturing products specified in this Section with minimum three years' documented experience.
- B. Applicator: Disinfection shall be performed by an approved specialty contractor with minimum three years' experience.

051.18.2.22

#### **PART 2 - PRODUCTS**

# 2.1 DISINFECTION CHEMICALS

- A. Chemicals:
  - 1. Chlorine: Comply with AWWA Standards.

### **PART 3 - EXECUTION**

# 3.1 EXAMINATION

- A. Section 017000 Execution and Closeout Requirements: Requirements for installation examination.
- B. Verify that piping system has been cleaned, inspected, and pressure tested.
- C. Verify that access fittings have been installed under Section 331416 Site Water Utility Distribution Piping.
- D. Perform scheduling and disinfecting activity with startup, water pressure testing, adjusting and balancing, and demonstration procedures, including coordination with related systems.

#### 3.2 INSTALLATION

- A. Apply chlorine solution to achieve a concentration of at least 25 milligrams per liter free chlorine in new line. Retain chlorinated water for 24 hours. Chlorine concentration shall be recorded at every outlet along the line at the beginning and end of the 24 hour period. After 24 hours, all samples of water shall contain at least 10 milligrams per liter free chlorine. Rechlorinate if required results are not obtained on all samples.
- B. Disposal of Chlorinated Water: Reduce chlorine residual of disinfection water to that of existing distribution system prior to disposal. Treat water with sulfur dioxide or other reducing chemicals to neutralize chlorine residual. Flush all lines until residual is equal to existing system.

# 3.3 FIELD QUALITY CONTROL

- A. Section 014000 Quality Requirements: Requirements for inspecting and testing.
- B. Section 017000 Execution and Closeout Requirements: Requirements for testing, adjusting, and balancing.
- C. Bacteriological Testing: After final flushing and before the water along main is placed in service, the Contractor shall collect samples from the line and have tested for bacteriological quality in accordance with the rules of the Georgia Department of Natural Resources, Environmental Protection Division. Testing shall be performed by a laboratory certified by the State of Georgia. Re-chlorinate lines until required results are obtained.

### **END OF SECTION 330110**

# SECTION 330505 HYDROSTATIC TESTING

### **PART 1 - GENERAL**

# 1.1 SUMMARY

- A. Section Includes: Hydrostatic testing of pressure piping.
- B. Related Requirements:
  - 1. Section 331413 Public Water Utility Distribution Piping: Pipe materials and accessories normally encountered with pressurized water distribution systems.

### 1.2 REFERENCE STANDARDS

- A. American Water Works Association:
  - 1. AWWA C600 Installation of Ductile-Iron Water Mains and Their Appurtenances.

### 1.3 UNIT PRICE - MEASUREMENT AND PAYMENT

A. Section 012000 – Price and Payment Procedures: Contract Sum Price/Modification Procedures

# 1.4 SUBMITTALS

- A. Section 013300 Submittal Procedures: Requirements for submittals.
- B. Submit following items prior to start of testing:
  - 1. Testing procedures.
  - 2. List of test equipment.
  - 3. Testing sequence schedule.
  - 4. Provisions for disposal of flushing and test water.
  - 5. Certification of test gage calibration.
- C. Test and Evaluation Reports: Indicate results of piping tests.
- D. Qualifications Statement:
  - 1. Submit qualifications for applicator.

# 1.5 QUALITY ASSURANCE

A. Perform Work according to owner and current industry standards.

B. Maintain one copy of each standard affecting Work of this Section on Site.

# 1.6 QUALIFICATIONS

A. Applicator: Company specializing in performing Work of this Section with minimum three years' documented experience.

### **PART 2 - PRODUCTS**

# 2.1 HYDROSTATIC TESTING

- A. Equipment:
  - 1. Pressure pump.
  - 2. Pressure hose.
  - 3. Water meter.
  - 4. Test connections.
  - 5. Pressure relief valve.
  - 6. Pressure Gage: Calibrated to 0.1 psi.

### **PART 3 - EXECUTION**

### 3.1 EXAMINATION

- A. Section 017000 Execution and Closeout Requirements: Requirements for installation examination.
- B. Verify that piping is ready for testing.
- C. Verify that trenches are backfilled.
- D. Verify that pressure piping thrust restraints have been installed.

# 3.2 FIELD QUALITY CONTROL

- A. Section 014000 Quality Requirements: Requirements for inspecting and testing.
- B. Section 017000 Execution and Closeout Requirements: Requirements for testing, adjusting, and balancing.

- C. Testing of Pressure Piping:
  - 1. Test system according to AWWA C600 and following:

- a. Hydrostatically test each portion of pressure piping, including valved section, at 1.5 times working pressure of piping, based on elevation of lowest point in piping corrected to elevation of test gage.
- b. Conduct hydrostatic testing for at least two hours.
- c. Slowly fill with water portion of piping to be tested, expelling air from piping at high points.
- d. Install corporation cocks at high points.
- e. Close air vents and corporation cocks after air is expelled.
- f. Raise pressure to specified test pressure.
- g. Observe joints, fittings, and valves undergoing testing.
- h. Remove and renew cracked pipes, joints, fittings, and valves that show visible leakage.
- i. Retest.
- j. Correct visible deficiencies and continue testing at same test pressure for additional two hours to determine leakage rate.
- k. Maintain pressure within plus or minus 5.0 psi of test pressure.
- 1. Leakage is defined as quantity of water supplied to piping necessary to maintain test pressure during period of testing.
- m. Compute maximum allowable leakage using following formula:
  - 1)  $L = [SD \times sqrt(P)]/C$ .
  - 2) L = testing allowance, gph.
  - S = length of pipe tested, feet.
  - 4) D = nominal diameter of pipe, inches.
  - 5) P = average test pressure during hydrostatic testing, psig.
  - 6) C = 148,000.
  - 7) If pipe undergoing testing contains sections of various diameters, calculate allowable leakage from sum of computed leakage for each pipe size.
- 2. If testing of piping indicates leakage greater than that allowed, locate source of leakage, make corrections, and retest until leakage is within acceptable limits.
- 3. Correct visible leaks regardless of quantity of leakage.

### **END OF SECTION 330505**

# SECTION 330509 THRUST RESTRAINT FOR UTILITY PIPING

#### PART 1 - GENERAL

### 1.1 SUMMARY

- A. Section Includes:
  - 1. Tied joint restraint system.
- B. Related Requirements:
  - 1. Section 312333 Trenching and Backfilling: Trenching and backfilling requirements for Site utilities.
  - 2. Section 331413 Public Water Utility Distribution Piping: Requirements for piping Work as required by this Section.

### 1.2 UNIT PRICE - MEASUREMENT AND PAYMENT

A. Section 012000 – Price and Payment Procedures: Contract Sum Price/Modification Procedures

# 1.3 REFERENCE STANDARDS

- A. American Water Works Association:
  - 1. AWWA C600 Installation of Ductile-Iron Water Mains and Their Appurtenances.
- B. ASME International:
  - 1. ASME B1.1 Unified Inch Screw Threads, UN and UNR Thread Form.
- C. ASTM International:
  - 1. ASTM A36 Standard Specification for Carbon Structural Steel.
  - 2. ASTM A123 Standard Specification for Zinc (Hot-Dip Galvanized) Coatings on Iron and Steel Products.
  - 3. ASTM A153 Standard Specification for Zinc Coating (Hot-Dip) on Iron and Steel Hardware.
  - 4. ASTM A307 Standard Specification for Carbon Steel Bolts, Studs, and Threaded Rod 60 000 PSI Tensile Strength.
  - 5. ASTM A325 Standard Specification for Structural Bolts, Steel, Heat Treated, 120/105 ksi Minimum Tensile Strength.
  - 6. ASTM A563 Standard Specification for Carbon and Alloy Steel Nuts.
  - 7. ASTM A588 Standard Specification for High-Strength Low-Alloy Structural Steel, up to 50 ksi Minimum Yield Point, with Atmospheric Corrosion Resistance.

- 8. ASTM B633 Standard Specification for Electrodeposited Coatings of Zinc on Iron and Steel.
- 9. ASTM F436 Standard Specification for Hardened Steel Washers.

# 1.4 COORDINATION

- A. Section 013000 Administrative Requirements: Requirements for coordination.
- B. Coordinate Work of this Section with installation of fittings and joints that require restraint.

### 1.5 PREINSTALLATION MEETINGS

- A. Section 013000 Administrative Requirements: Requirements for preinstallation meeting.
- B. Convene minimum one week prior to commencing Work of this Section.

### 1.6 SUBMITTALS

- A. Section 013300 Submittal Procedures: Requirements for submittals.
- B. Product Data: Submit manufacturer catalog information for restrained joint details and installation instructions.
- C. Shop Drawings:
  - 1. Indicate restrained joint details and materials being used.
  - 2. Submit layout drawings showing piece numbers and locations.
  - 3. Indicate restrained joint locations.
- D. Samples: Submit two samples of joint restraint parts.
- E. Manufacturer's Certificate: Certify that products meet or exceed specified requirements.
- F. Delegated Design Submittals:
  - 1. Submit signed and sealed Shop Drawings with design calculations and assumptions for restrained lengths.
  - 2. Submit joint restraint details.
  - 3. Use joint restraint devices specifically designed for applications described in manufacturer information.
- G. Manufacturer Instructions: Submit detailed instructions on installation requirements, including storage and handling procedures.
- H. Qualifications Statement:
  - 1. Submit qualifications for manufacturer, fabricator, and licensed professional.

# 1.7 SUSTAINABLE DESIGN SUBMITTALS – (NOT USED)

A. Section 018113 - Sustainable Design Requirements: Requirements for sustainable design submittals.

#### B. Manufacturer's Certificate:

- 1. Certify that products meet or exceed specified sustainable design requirements.
- 2. Materials Resources Certificates:
  - a. Certify source and origin for salvaged and reused products.
  - b. Certify recycled material content for recycled content products.
  - c. Certify source for regional materials and distance from Project Site.

### C. Product Cost Data:

- 1. Submit cost of products to verify compliance with Project sustainable design requirements.
- 2. Exclude cost of labor and equipment to install products.
- 3. Provide cost data for following products:
  - a. Salvaged, refurbished, and reused products.
  - b. Products with recycled material content.
  - c. Regional products.

### 1.8 CLOSEOUT SUBMITTALS

- A. Section 017000 Execution and Closeout Requirements: Requirements for submittals.
- B. Project Record Documents: Record actual locations of joint restraints.

# 1.9 QUALITY ASSURANCE

- A. Perform Work according to Owner and current industry standards.
- B. Maintain one copy of each standard affecting Work of this Section on Site.

# 1.10 QUALIFICATIONS

- A. Manufacturer: Company specializing in manufacturing products specified in this Section with minimum three years' documented experience.
- B. Fabricator: Company specializing in fabricating products specified in this Section with minimum three years' documented experience and approved by manufacturer.
- C. Licensed Professional: Professional engineer experienced in design of specified Work and licensed in State of Georgia.

# 1.11 DELIVERY, STORAGE, AND HANDLING

- A. Section 016000 Product Requirements: Requirements for transporting, handling, storing, and protecting products.
- B. Inspection: Accept materials on Site in manufacturer's original packaging and inspect for damage.
- C. Store materials according to manufacturer instructions.

#### D. Protection:

- 1. Protect materials from moisture and dust by storing in clean, dry location remote from construction operations areas.
- 2. Provide additional protection according to manufacturer instructions.

# 1.12 EXISTING CONDITIONS

#### A. Field Measurements:

- 1. Verify field measurements prior to fabrication.
- 2. Indicate field measurements on Shop Drawings.

### **PART 2 - PRODUCTS**

### 2.1 PERFORMANCE AND DESIGN CRITERIA

- A. Provide pressure pipeline with restrained joints at each bend, tee, and change in direction.
- B. Provide restraint at all points where hydraulic thrust may develop.

# 2.2 TIED JOINT RESTRAINT SYSTEMS

### A. Manufacturers:

1. Furnish materials according to Owner, current industry standards and per Part 1.10 Qualifications.

### B. Tie Bolts:

- 1. Mechanical Joints on 2 and 3-Inch Pipe:
  - a. Size: 5/8 inch.
  - b. Comply with ASTM A588, Grade B.
  - c. Comply with ASTM A325, Type 3, except increase tensile strength of full-body threaded section to 40,000 lb. minimum for 5/8 inch and 60,000 lb. minimum for 3/4 inch by heat-treating (quenching and tempering) to manufacturer's reheat and hardness specifications.

- 2. Mechanical and Flanged Joints on 4 to 12-Inch Pipe:
  - a. Size: 3/4 inch.
  - b. Comply with ASTM A588, Grade B.
  - c. Comply with ASTM A325, Type 3, except increase tensile strength of full-body threaded section to 40,000 lb. minimum for 5/8 inch and 60,000 lb. minimum for 3/4 inch by heat-treating (quenching and tempering) to manufacturer's reheat and hardness specifications.
- 3. Mechanical Joints on 14 to 24-Inch Pipe:
  - a. Size: 3/4 inch.
  - b. Comply with ASTM A588, Grade B and ASTM A325, Type 3.
- 4. Mechanical and Flanged Joints on 30-Inch and Larger Pipe:
  - a. Size: 1 inch.
  - b. Comply with ASTM A588, Grade B.
  - c. Comply with ASTM A325, Type 3, except increase tensile strength of full-body threaded section to 100,000 lb. minimum by heat-treating (quenching and tempering) to manufacturer's reheat and hardness specifications.

### C. Tie Nut:

- 1. Description: Hex nut for each tie bolt and tie rods.
- 2. Comply with ASTM A563, Grade C3.
- 3. Finish: galvanized.

### D. Tiepin:

- 1. Bends and Hydrants: 3/4-inch round bar stock.
- 2. Size and Shape: 6-inch hairpin.
- 3. Comply with ASME B1.1 and ASTM A588.
- 4. Finish: galvanized.

# E. Tie Coupling:

- 1. Description: Extension of continuous-threaded rods.
- 2. Provide with center stop to aid installation.
- 3. Comply with ASTM A588.
- 4. Finish: galvanized.

# F. Tie Clamp:

- 1. Description: Retainer clamp for ductile iron, asbestos-cement, and PVC push-on pipe.
- 2. Location: In front of bell.
- 3. Comply with ASTM A36, ASTM A307, Grade A, and ASTM A563, Grade A.
- 4. Finish: galvanized.
- G. Tie Rod:

- 1. Description: Continuous-threaded rod for cutting to desired lengths.
- 2. Comply with ASTM A588, Grade B, ASTM A325, Type 3, and ASME B1.1.
- 3. Finish: galvanized.

# H. Tie Bar:

- 1. Description: Steel bar used to restrain push-in plugs.
- 2. Comply with ASTM A36.
- 3. Finish: galvanized.

### I. Tie Washer:

- 1. Description: Round flat washers.
- 2. ASTM A588, ASTM F436, Type 1.
- 3. Finish: galvanized.

### J. Harnessing

- 1. Provide harness rods only where specifically shown on the Drawings or directed by the ENGINEER.
- 2. Where possible, harness rods shall be installed through the mechanical joint bolt holes. Where it is not possible, provide 90 degree bend eye bolts.

### K. Retainer Glands:

1. Provide retainer glands where shown on the Drawings and all associated fittings, valves and related piping. Retainer glands shall be installed in accordance with the manufacturer's recommendations, particularly, the required torque of the set screws. The CONTRACTOR shall furnish a torque wrench to verify the torque on all set screws which do not have inherent torque indicators.

# L. Concrete Thrust Collars

1. Collars shall be constructed as shown on the Drawings. Concrete shall be as shown on the Drawings. Reinforcing steel shall be as shown on the Drawings. The thrust collar shall be a weld on collar as specified in this Section. Weld-on collar shall be continuously welded to the pipe by the pipe manufacturer.

### M. Concrete Blocking

- 1. Provide concrete blocking for all bends, tees, valves, and other points where thrust may develop.
- 2. Concrete shall be Class "B" as specified in GDOT Specification 500 Concrete.

051.18.2.22

3. Form and pour concrete blocking at fittings as shown on the Drawings and as directed by the ENGINEER. Pour blocking against undisturbed earth. Increase dimensions where required excavation.

# 2.3 SUSTAINABILITY CHARACTERISTICS – (NOT USED)

- A. Section 018113 Sustainable Design Requirements: Requirements for sustainable design compliance.
- B. Material and Resource Characteristics:
  - 1. Recycled Content Materials: Furnish materials with maximum available recycled content.
  - 2. Regional Materials: Furnish materials extracted, processed, and manufactured within 500 miles of Project Site.

### 2.4 MATERIALS

- A. Steel:
  - 1. High-Strength Low-Alloy Steel: Comply with ASTM A588, heat treated.
  - 2. High-Strength Low-Alloy Steel: Comply with ASTM A588.
  - 3. Carbon Steel: Comply with ASTM A36.

### 2.5 FINISHES

- A. Zinc Plating:
  - 1. Factory applied.
  - 2. Comply with ASTM B633.
- B. Galvanizing:
  - 1. Factory applied.
  - 2. Comply with ASTM A153.
- C. Galvanizing:
  - 1. Comply with ASTM A123.
  - 2. Hot dip galvanized after fabrication.

### **PART 3 - EXECUTION**

### 3.1 EXAMINATION

- A. Section 017000 Execution and Closeout Requirements: Requirements for installation examination.
- B. Verify that pipe and fittings are ready to receive Work.
- C. Field measure and verify conditions for installation of Work.

# 3.2 PREPARATION

- A. Section 017000 Execution and Closeout Requirements: Requirements for installation preparation.
- B. Clean surfaces of pipe and fittings that are to receive tied joint restraint systems.

# 3.3 INSTALLATION

- A. According to AWWA C600.
- B. Install joint restraint system such that joints are mechanically locked together to prevent joint separation.
- C. Installation Standards: Install Work according to Owner and current industry standards.

### 3.4 TOLERANCES

- A. Section 014000 Quality Requirements: Requirements for tolerances.
- B. Torque 5/8-inch nuts on mating threaded fasteners from 45 to 60 ft.-lbf.
- C. Torque 3/4-inch nuts on mating threaded fasteners from 75 to 90 ft.-lbf..
- D. Torque 1-inch nuts from 100 to 120 ft.-lbf.

**END OF SECTION 330509** 

# SECTION 330561 CONCRETE MANHOLES

### **PART 1 - GENERAL**

### 1.1 SUMMARY

#### A. Section Includes:

- 1. Modular precast concrete manholes and structures with tongue-and-groove joints and masonry transition to cover frame, covers, anchorage, and accessories.
- 2. Masonry manhole and structure sections with masonry transition to cover frame, covers, anchorage, and accessories.
- 3. Cast-in-place concrete manholes and structures with masonry transition to cover frame, covers, anchorage, and accessories.
- 4. Doghouse manhole connections to existing sanitary sewer lines.
- 5. Bedding and cover materials.
- 6. Pile support systems.
- 7. Vertical adjustment of existing manholes and structures.

# B. Related Requirements:

1. Section 31 23 33 – Trenching and Backfilling: Excavating for manholes, structures, and foundation slabs.

### 1.2 **DEFINITIONS**

A. Bedding: Specialized material placed under manhole prior to installation and subsequent backfill operations.

### 1.3 UNIT PRICE - MEASUREMENT AND PAYMENT

A. Section 012000 - Price and Payment Procedures: Contract Sum/Price modification procedures.

# 1.4 REFERENCE STANDARDS

- A. American Association of State Highway Transportation Officials:
  - 1. AASHTO M91 Standard Specification for Sewer and Manhole Brick (Made from Clay or Shale).
  - 2. AASHTO M288 Standard Specification for Geotextile Specification for Highway Applications.
  - 3. AASHTO M306 Standard Specification for Drainage, Sewer, Utility, and Related Castings.

### B. American Concrete Institute:

- 1. ACI 318 Building Code Requirements for Structural Concrete.
- 2. ACI 350 Code Requirements for Environmental Engineering Concrete Structures.

3. ACI 530/530.1 - Building Code Requirements and Specification for Masonry Structures.

### C. ASTM International:

- 1. ASTM A48 Standard Specification for Gray Iron Castings.
- 2. ASTM A123 Standard Specification for Zinc (Hot-Dip Galvanized) Coatings on Iron and Steel Products.
- 3. ASTM C32 Standard Specification for Sewer and Manhole Brick (Made from Clay or Shale).
- 4. ASTM C55 Standard Specification for Concrete Building Brick.
- 5. ASTM C361 Standard Specification for Reinforced Concrete Low-Head Pressure Pipe.
- 6. ASTM C478 Standard Specification for Circular Precast Reinforced Concrete Manhole Sections.
- 7. ASTM C497 Standard Test Methods for Concrete Pipe, Manhole Sections, or Tile.
- 8. ASTM C877 Standard Specification for External Sealing Bands for Concrete Pipe, Manholes, and Precast Box Sections.
- 9. ASTM C913 Standard Specification for Precast Concrete Water and Wastewater Structures.
- 10. ASTM C923 Standard Specification for Resilient Connectors between Reinforced Concrete Manhole Structures, Pipes, and Laterals.
- 11. ASTM C990 Standard Specification for Joints for Concrete Pipe, Manholes, and Precast Box Sections Using Preformed Flexible Joint Sealants.
- 12. ASTM F593 Standard Specification for Stainless Steel Bolts, Hex Cap Screws, and Studs.
- 13. ASTM F1554 Standard Specification for Anchor Bolts, Steel, 36, 55, and 105-ksi Yield Strength.

# 1.5 COORDINATION

- A. Section 013000 Administrative Requirements: Requirements for coordination.
- B. Coordinate Work of this Section with connection to municipal sewer utility service and trenching.

### 1.6 PREINSTALLATION MEETINGS

- A. Section 013000 Administrative Requirements: Requirements for preinstallation meeting.
- B. Convene minimum one week prior to commencing Work of this Section.

### 1.7 SUBMITTALS

- A. Section 013300 Submittal Procedures: Requirements for submittals.
- B. Product Data: Submit manufacturer information for manhole covers, component construction, features, configuration, and dimensions.
- C. Shop Drawings:

- 1. Indicate structure locations and elevations.
- 2. Indicate sizes and elevations of piping, conduit, and penetrations.
- D. Manufacturer's Certificate: Certify that products meet or exceed specified requirements.
- E. Manufacturer Instructions: Submit detailed instructions on installation requirements, including storage and handling procedures.
- F. Source Quality-Control Submittals: Indicate results of factory tests and inspections.
- G. Field Quality-Control Submittals: Indicate results of Contractor-furnished tests and inspections.
- H. Qualifications Statement:
  - 1. Submit qualifications for manufacturer.

# 1.8 SUSTAINABLE DESIGN SUBMITTALS – (NOT USED)

### 1.9 CLOSEOUT SUBMITTALS

- A. Section 017000 Execution and Closeout Requirements: Requirements for submittals.
- B. Project Record Documents: Record actual locations of manholes and connections, and record invert elevations.

### 1.10 QUALITY ASSURANCE

A. Perform Work according to municipal and current industry standards.

# 1.11 QUALIFICATIONS

A. Manufacturer: Company specializing in manufacturing products specified in this Section with minimum three years' documented experience.

# 1.12 DELIVERY, STORAGE, AND HANDLING

- A. Section 01 60 00 Product Requirements: Requirements for transporting, handling, storing, and protecting products.
- B. Inspection: Accept materials on Site in manufacturer's original packaging and inspect for damage.
- C. Handling: Comply with precast concrete manufacturer instructions and ASTM C913 for unloading and moving precast manholes and drainage structures.
- D. Storage:

- 1. Store materials according to manufacturer instructions.
- 2. Store precast concrete manholes and drainage structures to prevent damage to Owner's property or other public or private property.
- 3. Repair property damaged from materials storage.

#### E. Protection:

- 1. Protect materials from moisture and dust by storing in clean, dry location remote from construction operations areas.
- 2. Provide additional protection according to manufacturer instructions.

### 1.13 AMBIENT CONDITIONS

- A. Section 015000 Temporary Facilities and Controls: Requirements for ambient condition control facilities for product storage and installation.
- B. Cold Weather Requirements: Comply with ACI 530/530.1.

### 1.14 EXISTING CONDITIONS

- A. Field Measurements:
  - 1. Verify field measurements prior to fabrication.
  - 2. Indicate field measurements on Shop Drawings.

### 1.15 WARRANTY

- A. Section 017000 Execution and Closeout Requirements: Requirements for warranties.
- B. Furnish five-year manufacturer's warranty for concrete manholes.

# **PART 2 - PRODUCTS**

### 2.1 CONCRETE AND MASONRY MANHOLES

- A. Manufacturers:
  - 1. Furnish materials according to municipal and current industry standards.
- B. Manhole Sections:
  - 1. Materials:
    - a. Reinforced Precast Concrete: Comply with ASTM C478.
    - b. Gaskets: Comply with ASTM C923.
  - 2. Joints:

- a. Comply with ASTM C913.
- b. Maximum Leakage: 0.025 gal. per hour per foot of joint at 3 feet of head.
- c. The reinforced concrete manhole in riser sections, excepting grade rings, shall be formed with male and female ends with rubber gasket rings between the male and female end to reduce infiltration and leakage.

# 3. Manhole Steps:

a. Manhole steps that are cast or mortared into the walls of risers or conical top sections shall be aligned in each section so as to form a continuous ladder with rungs equally spaced vertically not more than twelve-inch (12") centers. Steps shall be installed in manholes to within two feet (2') of crown of lowest pipe.

### 4. Finished Grade:

a. Precast manholes shall be furnished and set with proper allowance for the cast iron frame and cover to be set at the correct finished grade as shown on the plans.

### 5. Manhole Boots:

a. Provide neoprene boot seals where sewer enters the manhole and wet well. Holes for pipe entering or leaving shall be core-drilled at the factory.

### C. Clay Brick Units:

- 1. Comply with ASTM C32.
- 2. Grade: MS.
- 3. Configuration: Quality, sound, hard-burned, perfectly shaped with smooth regular shape.
- 4. Shall not absorb more than 16 percent of water by weight when submerged in water for 24 hrs, having been in a thoroughly dry state prior to placing in water.
- 5. Nominal Modular Size: Match existing or as directed in Drawings.
- 6. Bed bottom and sides of any bricks used with mortar. Apply a smooth coat of mortar three-quarter-inch (3/4") thick on the inside and outside.

# D. Mortar and Grout:

- 1. Mortar:
  - a. All cement used in mortar shall conform to ASTM Designation C-150 and the latest revision thereof. All mortar used shall be comprised of one (1) part Portland Cement and two (2) parts fine sand.
- 2. Grout:
  - a. As specified in Section 03 60 00 Grouting.

# E. Shaft and Eccentric Cone Top Sections:

- 1. Pipe Sections: Reinforced precast concrete.
- 2. Sleeved to receive pipe and conduit sections.
- F. Shape: Cylindrical.

### G. Clear Inside Dimensions:

1. As indicated on Drawings.

### H. Design Depth:

- 1. As indicated on Drawings.
- I. Clear Cover Opening:
  - 1. As indicated on Drawings.
- J. Pipe Entry: Furnish openings as indicated on Drawings.
- K. Structure Joint Gaskets:
  - 1. Comply with ASTM C361.
  - 2. Material: Rubber.

# 2.2 MANHOLE FRAMES, COVERS, AND STEPS

### A. Manufacturers:

1. Furnish materials according to municipal and industry standards.

### B. Description:

- 1. Material: Cast iron.
- 2. Comply with ASTM A48, Class 20.
- 3. Casting must be 95% of design weight.
- 4. Surface: Machined horizontal bearing surfaces.
- 5. Cover: Rotatable and interchangeable
  - a. Bolt down covers shall be equipped with four (4) each one-half-inch (1/2") diameter stainless steel bolts and a one-eighth-inch (1/8") neoprene O-ring gasket.
  - b. Bolt holes shall be bored through so that debris entering the bolt hole will fall into the manhole.
- 6. Cover: Molded with identifying name "Sanitary Sewer".
- 7. Cast iron manhole steps shall be coated with a non-corrosive material such as plastic, or other material approved by the engineer.
- 8. All castings shall be thoroughly cleaned in the shop and given two (2) coats of approved bituminous paint before rusting occurs.

Type	Design Weight	Manufacturer's Reference
Traffic	450lbs	Neenah R-1712B
Watertight Traffic	400lbs	Neenah R-1916F1
Vented	410lbs	Neenah R-1781

# 2.3 PILE SUPPORT SYSTEMS – (NOT USED)

# 2.4 SUSTAINABILITY CHARACTERISTICS – (NOT USED)

### 2.5 MATERIALS

### A. Bedding:

1. Bedding: All bedding materials shall be crushed stone unless shown or specified otherwise. Crushed stone bedding material shall meet the requirements of Georgia Department of Transportation Specification 800.01 for No. 57 stone and be eight inches (8") thick.

### 2.6 ACCESSORIES

# A. Steps:

- 1. Rungs: Non-corrosive material such as plastic, galvanized metal, aluminum, or other material approved by the Engineer.
- 2. Fabrication: Formed integral with manhole sections.
- 3. Diameter, width and spacing:
  - a. Per manufacturer recommendations for meeting current industry safety standards.
- B. Joint Sealant: Comply with ASTM C990.
- C. Fasteners: Stainless steel; ASTM F593.
- D. Geotextile Filter Fabric: As indicated on Drawings.
- E. Soil Backfill from Above Pipe to Finish Grade:
  - 1. As specified in Section –312333 Trenching and Backfilling.
  - 2. Subsoil: No frozen earth, or foreign matter, or rocks more than 6 inches in diameter.

### 2.7 FINISHES

- A. Bituminous Interior Manhole Coating:
  - 1. Manufacturers:
    - a. Furnish materials according to municipal and current industry standards.
- B. Steel Galvanizing:
  - 1. Hot-dip galvanize after fabrication.
  - 2. Comply with ASTM A123.

### 2.8 SOURCE QUALITY CONTROL

- A. Section 014000 Quality Requirements: Requirements for testing, inspection, and analysis.
- B. Provide shop inspection and testing of completed assembly.

# C. Owner Inspection:

- 1. If requested by Owner, make completed material available for inspection at manufacturer's factory prior to packaging for shipment.
- 2. Notify Owner at least seven days before inspection is allowed.

# D. Owner Witnessing:

- 1. If requested by Owner, allow witnessing of factory inspections and tests at manufacturer's test facility.
- 2. Notify Owner at least seven days before inspections and tests are scheduled.

### E. Certificate of Compliance:

- 1. If manufacturer is approved by authorities having jurisdiction, submit certificate of compliance indicating Work performed at manufacturer's facility conforms to Contract Documents.
- 2. Specified shop tests are not required for Work performed by approved manufacturer.

### **PART 3 - EXECUTION**

### 3.1 EXAMINATION

- A. Section 017000 Execution and Closeout Requirements: Requirements for installation examination.
- B. Verify that items provided by other Sections of Work are properly sized and located.
- C. Verify that built-in items are in proper location and are ready for roughing into Work.
- D. Verify that excavation base is ready to receive Work and excavations and that dimensions and elevations are as indicated on Drawings.

### 3.2 PREPARATION

- A. Section 017000 Execution and Closeout Requirements: Requirements for installation preparation.
- B. Mark each precast structure by indentation or waterproof paint showing date of manufacture, manufacturer, and identifying symbols and numbers as indicated on Drawings to indicate its intended use.
- C. Coordinate placement of inlet and outlet pipe or duct sleeves as required by other Sections.

- D. Do not install manholes and structures where Site conditions induce loads exceeding structural capacity of manholes or structures.
- E. Inspect precast concrete manholes and structures immediately prior to placement in excavation to verify that they are internally clean and free from damage; remove and replace damaged units.

### 3.3 INSTALLATION

- A. Conduct operations not to interfere with, interrupt, damage, destroy, or endanger integrity of surface structures or utilities in immediate or adjacent areas.
- B. Manhole and manhole drop connections shall be constructed at the locations shown on the plans. Manholes shall be to the depth shown on the plans or as directed by the engineer. Details of manholes and drop connections are shown on the plans and construction shall be in accordance with these details.
- C. Where inlet leads and mains or lateral pipe sewers enter a manhole, such pipes shall be cut off flush with the inside of the manhole and any irregularities shall be pointed up with mortar. All pre-cast manhole joints, both inside and outside, shall be wiped with mortar for a smooth surface.
- D. The inverts of the sewer lines entering manholes at or near the flow line elevation of the manhole shall be shaped and routed across the floor of the manhole using concrete and mortar to obtain a minimum of 1" drop across the manhole.
- E. After the manhole construction has been completed to the proper elevation, the cast iron manhole frame shall be set in a full bed of mortar and adjusted to the elevation established by the engineer.
- F. The two sides of the concrete encasement on standard drop connections which are adjacent to the manhole shall extend four inches (4") outside the bells of the pipe in the drop connection. The side of the encasement opposite the manhole shall be flush with the pipe bells. The entire drop connection shall be joined with the wall of the manhole in a solid mass of concrete. Drop connections shall be constructed at the time the manhole is constructed. Unless a lateral is to be immediately connected into the drop connection, it shall be plugged at the outer end of the wye or tee with a standard concrete pipe plug. A drop manhole shall be required if the invert of sewer entering manhole is 24" or more than the manhole invert.
- G. Remove large stones or other hard matter impeding consistent backfilling or compaction.
- H. Protect manhole from damage or displacement while backfilling operation is in progress.
- I. Excavating:
  - 1. As specified in Section 312333 Trenching and Backfilling, and in indicated locations and depths.
  - 2. Provide clearance around sidewalls of manhole or structure for construction operations, granular backfill, and placement of geotextile filter fabric.

- 3. If ground water is encountered, prevent accumulation of water in excavations; place manhole or structure in dry trench.
- 4. Where possibility exists of watertight manhole or structure becoming buoyant in flooded excavation, anchor manhole or structure to avoid flotation as approved by Engineer.

# J. Base and Alignment:

- 1. Install manholes supported at proper grade and alignment support system as indicated on Drawings.
- 2. Grout base of shaft sections to achieve slope to exit piping, trowel smooth, and contour as indicated on Drawings.
- 3. Form and place manhole or structure cylinders plumb and level, to correct dimensions and elevations.

### K. Attachments:

- 1. As Work progresses, build fabricated metal items.
- 2. Cut and fit for pipe, conduit, and sleeves.
- 3. Set cover frames and covers level to correct elevations without tipping.
- L. Backfilling: As specified in Section 312333 Trenching and Backfilling.
- M. Coating: Paint interior with two coats of bituminous interior coating at rate of 120 sq. ft./gal. for each coat.

### N. Precast Concrete Manholes:

- 1. Lift precast components at lifting points designated by manufacturer.
- 2. When lowering manholes and structures into excavations and joining pipe to units, take precautions to ensure that interior of pipeline and structure remains clean.
- 3. Assembly:
  - a. Assemble multisection manholes and structures by lowering each section into excavation.
  - b. Install rubber gasket joints between precast sections according to manufacturer recommendations.
  - c. Lower, set level, and firmly position base section before placing additional sections.
- 4. Remove foreign materials from joint surfaces and verify that sealing materials are placed properly.
- 5. Maintain alignment between sections by using guide devices affixed to lower section.
- 6. Joint sealing materials may be installed on Site or at manufacturer's plant.
- 7. Verify that installed manholes and structures meet required alignment and grade.
- 8. Remove knockouts or cut structure to receive piping without creating openings larger than required to receive pipe; fill annular spaces with mortar.
- 9. Cut pipe flush with interior of structure.
- 10. Shape inverts through manhole and structures as indicated on Drawings.
- O. Masonry Manholes:

- 1. Maintain masonry courses to uniform dimension.
- 2. Form vertical and horizontal joints of uniform thickness.
- 3. Lay masonry units in running bond.
- 4. Course one unit and one mortar joint to equal 8 inches.
- 5. Form flush mortar joints.
- 6. Lay masonry units in full bed of mortar with full head joints, uniformly jointed with other Work.
- 7. Install joint reinforcement 16 inches o.c.
- 8. Place joint reinforcement in first and second horizontal joints above base pad and below cover frame opening.

# P. Doghouse Manholes and Structures:

- 1. Stake out location and burial depth of existing sewer line in area of proposed manhole or structure.
- 2. Carefully excavate around existing sewer line to adequate depth for foundation slab installation.
- 3. Protect existing pipe from damage.
- 4. Cut out soft spots and replace with granular fill compacted to 95 percent maximum density.
- 5. Bear firmly and fully on support system as indicated on Drawings.
- 6. Install precast concrete manhole or structure around existing pipe according to applicable Paragraphs in this Section.
- 7. Grout pipe entrances.
- 8. Perform connection to existing pipe.
- 9. Block upstream flow at existing manhole or structure with expandable plug.
- 10. Use hydraulic saw to cut existing pipe at manhole or structure entrance and exit and along pipe length at a point halfway up OD on each side of pipe.
- 11. Bottom half of pipe is to remain as manhole flow channel.
- 12. Saw cut to smooth finish with top half of pipe flush with interior of manhole or structure.
- 13. Grout base of manhole or structure to achieve slope to manhole or structure channel and trowel smooth.
- Q. Sanitary Manhole Drop Connections: As indicated on Drawings.

# R. Castings:

- 1. Set frames using mortar and masonry as indicated on Drawings.
- 2. Install radially laid concrete brick with 1/4-inch-thick, vertical joints at inside perimeter.
- 3. Lay concrete brick in full bed of mortar and completely fill joints.
- 4. If more than one course of concrete brick is required, stagger vertical joints.
- 5. Set frame and cover as shown on the plans and grade away from manhole.

# 3.4 ADJUSTMENT OR REPLACEMENT OF FRAMES AND COVERS

- A. Raising Manhole Frames and Covers:
  - 1. Locate manholes on Drawings that are designated to be raised to grade.
  - 2. Do not adjust elevation with rubber manhole rings.

- 3. Concrete brick for manholes will be accepted only for adjustment of frame to grade (eight inches (8") maximum) and filler for manhole table contouring.
- 4. Use sealant to seal joints between manhole top, rubber rings, and frame.
- 5. Reinstall removed manhole frame and cover.
- B. Replacing Manhole Frames and Covers:
  - 1. Locate manholes on Drawings that are designated for replacement of frames and covers.
  - 2. Remove existing manhole frames and covers to enable reuse.
  - 3. Deliver removed manhole frames and covers to Owner as maintenance materials as specified in Section 017000 Execution and Closeout Requirements.
  - 4. Install new frames and covers for manholes as indicated on Drawings.
  - 5. Adjust new frames and covers to match finished grade.
  - 6. Seal joints between manholes and manhole frames.
- C. Installation Standards: Install Work according to municipal and current industry standards.

# 3.5 FIELD QUALITY CONTROL

- A. Section 014000 Quality Requirements: Requirements for inspecting and testing.
- B. Section 017000 Execution and Closeout Requirements: Requirements for testing, adjusting, and balancing.
- C. Testing:
  - 1. Concrete Manhole Sections:
    - a. Comply with ASTM C497.
- D. Equipment Acceptance: Adjust, repair, modify, or replace components failing to perform as specified and rerun tests.

#### 3.6 ADJUSTING

- A. Section 017000 Execution and Closeout Requirements: Requirements for starting and adjusting.
- B. Vertical Adjustment of Existing Manholes and Structures:
  - 1. Method shall be approved by Engineer.
  - 2. If required, adjust top elevation of existing manholes and structures to finished grades.
  - 3. Frames, Grates, and Covers:
    - a. Remove frames, grates, and covers cleaned of mortar fragments.
    - b. Reset to required elevation according to requirements specified for installation of castings.
  - 4. Reinforcing Bars:

- a. Remove concrete without damaging existing vertical reinforcing bars if removal of existing concrete wall is required.
- b. Clean vertical bars of concrete and bend into new concrete top slab or splice to required vertical reinforcement as indicated on Drawings.

# **END OF SECTION 330561**

# SECTION 331413 PUBLIC WATER UTILITY DISTRIBUTION PIPING

#### PART 1 - GENERAL

#### 1.1 SUMMARY

#### A. Section Includes:

- 1. Pipe and fittings for public line, including potable water line and fire water line.
- 2. Tapping sleeves and valves.
- 3. Positive-displacement meters.
- 4. Pipe support systems.
- 5. Bedding and cover materials.

#### B. Related Requirements:

- 1. Section 330110.58 Disinfection of Water Utility Piping Systems: Disinfection of water mains and appurtenances.
- 2. Section 331419 Valves and Hydrants for Water Utility Service: Fire hydrants, valves, and valve boxes for fire hydrant and water main installations.

## 1.2 UNIT PRICE - MEASUREMENT AND PAYMENT

A. Section 012000 - Price and Payment Procedures: Contract Sum/Price modification procedures.

# B. Pipe and Fittings:

- 1. Basis of Measurement: By linear foot.
- 2. Basis of Payment: Includes excavation and backfill; pipe, fittings, and appurtenances; bedding; thrust restraints; connection and tap to Site service piping; connection and tap to municipal utility water source.

#### C. Valves:

- 1. Basis of Measurement: By each.
- 2. Basis of Payment: Includes excavation, bedding, backfill, valve, fittings, and accessories.

# D. Fire Hydrants:

- 1. Basis of Measurement: By each.
- 2. Basis of Payment: Includes excavation, gravel sump, bedding, backfill, hydrant, valve, connection, and accessories.

#### E. Meters:

- 1. Basis of Measurement: By each.
- 2. Basis of Payment: Includes meter, fittings, and accessories.

# F. Taps:

- 1. Basis of Measurement: By each.
- 2. Basis of Payment: Includes tapping sleeve, tapping valves, and accessories.

#### 1.3 REFERENCE STANDARDS

- A. American Association of State Highway and Transportation Officials:
  - 1. AASHTO T 180 Standard Method of Test for Moisture-Density Relations of Soils Using a 10-lb Rammer and a 18-in. Drop.
- B. American Society of Mechanical Engineers:
  - 1. ASME B16.1 Gray Iron Pipe Flanges and Flanged Fittings: Classes 25, 125, and 250.

#### C. ASTM International:

- 1. ASTM A36 Standard Specification for Carbon Structural Steel.
- 2. ASTM A123 Standard Specification for Zinc (Hot-Dip Galvanized) Coatings on Iron and Steel Products).
- 3. ASTM A307 Standard Specification for Carbon Steel Bolts, Studs, and Threaded Rod 60 000 PSI Tensile Strength.
- 4. ASTM D698 Standard Test Methods for Laboratory Compaction Characteristics of Soil Using Standard Effort 12,400 ft-lbf/ft3.
- 5. ASTM D1557 Standard Test Methods for Laboratory Compaction Characteristics of Soil Using Modified Effort 56,000 ft-lbf/ft3.
- 6. ASTM D1785 Standard Specification for Poly(Vinyl Chloride) (PVC) Plastic Pipe, Schedules 40, 80, and 120.
- 7. ASTM D2241 Standard Specification for Poly(Vinyl Chloride) (PVC) Pressure-Rated Pipe (SDR Series).
- 8. ASTM D3035 Standard Specification for Polyethylene (PE) Plastic Pipe (DR-PR) Based on Controlled Outside Diameter.
- 9. ASTM D3139 Standard Specification for Joints for Plastic Pressure Pipes Using Flexible Elastomeric Seals.
- 10. ASTM D6938 Standard Test Method for In-Place Density and Water Content of Soil and Soil-Aggregate by Nuclear Methods (Shallow Depth).
- 11. ASTM F477 Standard Specification for Elastomeric Seals (Gaskets) for Joining Plastic Pipe.

# D. American Water Works Association:

- 1. AWWA C104 Cement-Mortar Lining for Ductile-Iron Pipe and Fittings.
- 2. AWWA C105 Polyethylene Encasement for Ductile-Iron Pipe Systems.
- 3. AWWA C110 Ductile-Iron and Gray-Iron Fittings.
- 4. AWWA C111 Rubber-Gasket Joints for Ductile-Iron Pressure Pipe and Fittings.
- 5. AWWA C115 Flanged Ductile-Iron Pipe with Ductile-Iron or Gray-Iron Threaded Flanges.
- 6. AWWA C151 Ductile-Iron Pipe, Centrifugally Cast.
- 7. AWWA C153 Ductile-Iron Compact Fittings.

- 8. AWWA C200 Steel Water Pipe, 6 In. (150 mm) and Larger.
- 9. AWWA C203 Coal-Tar Protective Coatings and Linings for Steel Water Pipe.
- AWWA C205 Cement-Mortar Protective Lining and Coating for Steel Water Pipe 4
   In. (100 mm) and Larger Shop Applied.
- 11. AWWA C206 Field Welding of Steel Water Pipe.
- 12. AWWA C207 Steel Pipe Flanges for Waterworks Service, Sizes 4 In. Through 144 In. (100 mm Through 3,600 mm).
- 13. AWWA C208 Dimensions for Fabricated Steel Water Pipe Fittings.
- 14. AWWA C213 Fusion-Bonded Epoxy Coatings and Linings for Steel Water Pipe and Fittings.
- 15. AWWA C300 Reinforced Concrete Pressure Pipe, Steel-Cylinder Type.
- 16. AWWA C301 Prestressed Concrete Pressure Pipe, Steel-Cylinder Type.
- 17. AWWA C500 Metal-Seated Gate Valves for Water Supply Service.
- 18. AWWA C600 Installation of Ductile-Iron Mains and Their Appurtenances.
- 19. AWWA C605 Underground Installation of Polyvinyl Chloride (PVC) and Molecularly Oriented Polyvinyl Chloride (PVCO) Pressure Pipe and Fittings.
- 20. AWWA C606 Grooved and Shouldered Joints.
- 21. AWWA C700 Cold-Water Meters Displacement Type, Metal Alloy Main Case.
- 22. AWWA C701 Cold-Water Meters Turbine Type, for Customer Service.
- 23. AWWA C702 Cold-Water Meters Compound Type.
- 24. AWWA C707 Encoder-Type Remote-Registration Systems for Cold-Water Meters.
- 25. AWWA C900 Polyvinyl Chloride (PVC) Pressure Pipe and Fabricated Fittings, 4 In. Through 12 In. (100 mm Through 300 mm), for Water Transmission and Distribution.
- 26. AWWA C901 Polyethylene (PE) Pressure Pipe and Tubing, 1/2 In. (13 mm) Through 3 In. (76 mm), for Water Service.
- 27. AWWA C905 Polyvinyl Chloride (PVC) Pressure Pipe and Fabricated Fittings, 14 In. Through 48 In. (350 mm Through 1,200 mm), for Water Transmission and Distribution.
- 28. AWWA M6 Water Meters Selection, Installation, Testing, and Maintenance.
- E. Manufacturers Standardization Society of the Valve and Fittings Industry:
  - 1. MSS SP-60 Connecting Flange Joints between Tapping Sleeves and Tapping Valves.
- F. National Fire Protection Association:
  - 1. NFPA 24 Standard for the Installation of Private Fire Service Mains and Their Appurtenances.
- G. NSF International:
  - 1. NSF 61 Drinking Water System Components Health Effects.
  - 2. NSF 372 Drinking Water System Components Lead Content.

# 1.4 COORDINATION

- A. Section 013000 Administrative Requirements: Requirements for coordination.
- B. Coordinate Work of this Section with termination of water main connection at Site boundary, connection to municipal water utility service, and trenching.

#### 1.5 PREINSTALLATION MEETINGS

- A. Section 013000 Administrative Requirements: Requirements for preinstallation meeting.
- B. Convene minimum one week prior to commencing Work of this Section.

#### **1.6** SUBMITTALS

- A. Section 013300 Submittal Procedures: Requirements for submittals.
- B. Product Data: Submit manufacturer information regarding pipe materials, pipe fittings, valves, and hydrants.
- C. Manufacturer's Certificate: Certify that products meet or exceed specified requirements.
- D. Field Quality-Control Submittals: Indicate results of Contractor-furnished tests and inspections.
- E. Preconstruction Photographs: Submit digital files of color photographs of Work areas and material storage areas, as specified in Section 017000 Execution and Closeout Requirements.
- F. Qualifications Statements:
  - 1. Submit qualifications for manufacturer and installer.

#### 1.7 SUSTAINABLE DESIGN SUBMITTALS (NOT USED)

- A. Section 018113 Sustainable Design Requirements: Requirements for sustainable design submittals.
- B. Manufacturer's Certificate:
  - 1. Certify that products meet or exceed specified sustainable design requirements.
  - 2. Materials Resources Certificates:
    - a. Certify source and origin for salvaged and reused products.
    - b. Certify recycled material content for recycled content products.
    - c. Certify source for regional materials and distance from Project Site.

051.18.2.22

#### C. Product Cost Data:

- 1. Submit cost of products to verify compliance with Project sustainable design requirements.
- 2. Exclude cost of labor and equipment to install products.
- 3. Provide cost data for following products:
  - a. Salvaged, refurbished, and reused products.
  - b. Products with recycled material content.
  - c. Regional products.

#### 1.8 CLOSEOUT SUBMITTALS

- A. Section 017000 Execution and Closeout Requirements: Requirements for submittals.
- B. Project Record Documents: Record actual locations of piping mains, valves, connections, thrust restraints, and **invert** elevations.
- C. Identify and describe unexpected variations to subsoil conditions or discovery of uncharted utilities.

# **1.9** QUALITY ASSURANCE

- A. Valves: Mark valve body with manufacturer's name and pressure rating.
- B. Materials in Contact with Potable Water: Certified according to NSF 61 and NSF 372.
- C. Perform Work according to City of Byron standards.
- D. Maintain one copy of each standard affecting Work of this Section on Site.

## 1.10 QUALIFICATIONS

- A. Manufacturer: Company specializing in manufacturing products specified in this Section with minimum three years' documented experience.
- B. Installer: Company specializing in performing Work of this Section with minimum three years' documented experience in installation of liner materials.

#### 1.11 DELIVERY, STORAGE, AND HANDLING

- A. Section 016000 Product Requirements: Requirements for transporting, handling, storing, and protecting products.
- B. Inspection: Accept materials on Site in manufacturer's original packaging and inspect for damage.

#### C. Storage:

- 1. Store materials according to manufacturer instructions.
- 2. Block individual and stockpiled pipe lengths to prevent moving.
- 3. Do not place pipe or pipe materials on private property or in areas obstructing pedestrian or vehicle traffic.
- 4. Store PE and PVC materials out of sunlight.

#### D. Protection:

- 1. Protect materials from moisture and dust by storing in clean, dry location remote from construction operations areas.
- 2. Provide additional protection according to manufacturer instructions.

#### 1.12 EXISTING CONDITIONS

#### A. Field Measurements:

- 1. Verify field measurements prior to fabrication.
- 2. Indicate field measurements on Shop Drawings.

#### 1.13 WARRANTY

A. Section 017000 - Execution and Closeout Requirements: Requirements for warranties

#### PART 2 - PRODUCTS

# **2.1** WATER PIPING

# A. Ductile-Iron Pipe:

- 1. Comply with AWWA C151.
- 2. Bituminous Outside Coating: Comply with AWWA C151.
- 3. Pipe Mortar Lining:
  - a. Comply with AWWA C104.
  - b. Thickness: Double.
- 4. PE Encasement: Comply with AWWA C105.
- 5. Pipe Class:
  - a. Comply with AWWA C151.
  - b. Class 56.

#### 6. Fittings:

- a. Material: Ductile iron; comply with AWWA C110.
- b. Compact Fittings: Comply with AWWA C153.
- c. Coating and Lining:
  - 1) Bituminous Coating: Comply with AWWA C110.
  - 2) Cement-Mortar Lining: Comply with AWWA C104; double thickness.

#### 7. Joints:

- a. Mechanical and Push-on Joints: Comply with AWWA C111.
- b. Flanged Joints: Comply with AWWA C115

#### B. PVC:

- 1. Comply with AWWA C900, Class 235.
- 2. Fittings: Comply with AWWA **C900**.
- 3. Joints:

- a. Comply with **ASTM D3139**.
- b. Seals: PVC flexible elastomeric.
- c. Solvent-cement couplings are not permitted.

#### C. Steel Pipe:

- 1. Comply with AWWA C200.
- 2. Type: **Mill** pipe.
- 3. Minimum Wall Thickness:
  - a. Pipe Diameters 8 Inches and Smaller: 0.375 inches.
  - b. Pipe Diameters Greater than 8 Inches: 0.50 inches.
- 4. Fittings and Special Sections: Comply with AWWA C208.
- 5. Flanges:
  - a. Comply with AWWA C207, Class D.
  - b. Type: Slip-on.
- 6. Field Welding Materials: Comply with AWWA C206.
- 7. Interior Cement Mortar Lining: Comply with AWWA C205.
- 8. Buried Steel Pipe Exterior Lining:
  - a. Description: Shop-applied prime coat and coal tar enamel protective coating.
  - b. Comply with AWWA C203.

## 2.2 TAPPING SLEEVES AND VALVES

#### A. Tapping Sleeves:

- 1. Description:
  - a. Material: **Ductile** iron.
  - b. Type: Dual compression.
  - c. Outlet Flange Dimensions and Drilling: Comply with ASME B16.1, Class 125, and MSS SP-60.

# B. Tapping Valves:

- 1. Description:
  - a. Comply with AWWA C500.
  - b. Type: Double disc with non-rising stem.
  - c. Inlet Flanges: Comply with ASME B16.1, Class 125, and MSS SP-60.

051.18.2.22

d. Mechanical Joint Outlets: Comply with AWWA C111.

# 2.3 VALVES AND FIRE HYDRANTS

A. As specified in Section 331419 - Valves and Hydrants for Water Utility Service.

#### **2.4** PIPE SUPPORTS AND ANCHORING

- A. Metal for Pipe Support Brackets:
  - 1. Material: Structural steel.
  - 2. Finish: Galvanized.
  - 3. Coating: Bituminous paint.
- B. Metal Tie Rods and Clamps or Lugs:
  - 1. Material: Galvanized steel.
  - 2. Size: Comply with NFPA 24.
  - 3. Coating: Bituminous paint.

# 2.5 CONCRETE ENCASEMENT AND CRADLES

- A. Concrete:
  - 1. As specified in Section 033000 Cast-in-Place Concrete.
  - 2. Type: Reinforced, air entrained.
  - 3. Compressive Strength: 4,000 psi at 28 days.
  - 4. Finish: Rough troweled.
- B. Concrete Reinforcement: As specified in Section 032000 Concrete Reinforcing.

#### 2.6 LOCATING AND DETECTION

- A. Locate Wire
  - 1. 12 AWG Solid Strand Copper Conductor
  - 2. PVC Insulation Thickness 15 mil
  - 3. Nylon Jacket Thickness 4 mil
  - 4. Nominal OD 0.121"
  - 5. Color Blue
- B. Detection Tape
  - 1. 2" Wide Detectable Marking Tape
  - 2. Overall Thickness 5 mil
  - 3. Aluminum Foil Thickness 0.35 mil
  - 4. Polyethylene Thickness 3.75 mil
  - 5. Polypropylene Thickness 0.8 mil
  - 6. Color Blue with Diagonal Stripes
  - 7. Printed Message "Caution Buried Potable Water Line Below"

# 2.7 SUSTAINABILITY CHARACTERISTICS (NOT USED)

- A. Section 018113 Sustainable Design Requirements: Requirements for sustainable design compliance.
- B. Material and Resource Characteristics:

- 1. Recycled Content Materials: Furnish materials with maximum available recycled content.
- 2. Regional Materials: Furnish materials extracted, processed, and manufactured within 500 miles of Project Site.

#### 2.8 FINISHES

- A. Steel: Hot-dip galvanized after fabrication, according to ASTM A123/A123M.
- B. Protective Coating: Bituminous paint.

## **2.9** ACCESSORIES

- A. Thrust Restraints: As specified in Section 330509.33 Thrust Restraint for Utility Piping.
- B. Steel Rods, Bolt, Lugs, and Brackets:
  - 1. Comply with ASTM **A36.**
  - 2. Grade A carbon steel.

#### PART 3 - EXECUTION

# 3.1 EXAMINATION

- A. Section 017000 Execution and Closeout Requirements: Requirements for installation examination.
- B. Verify that existing utility water main size, location, and invert are as indicated on Drawings.

# **3.2** PREPARATION

- A. Section 017000 Execution and Closeout Requirements: Requirements for installation preparation.
- B. Preconstruction Site Photos:
  - 1. As specified in Section 017000 Execution and Closeout Requirements.
  - 2. Take photographs along centerline of proposed pipe trench; minimum one photograph for each 50 feet of pipe trench.
  - 3. Show mailboxes, curbing, lawns, driveways, signs, culverts, and other existing Site features.
  - 4. Include Project description, date taken, and sequential number on back of each photograph.

#### C. Pipe Cutting:

1. Cut pipe ends square, ream pipe and tube ends to full pipe diameter, and remove burrs.

- 2. Use only equipment specifically designed for pipe cutting; use of chisels or hand saws is not permitted.
- 3. Grind edges smooth with beveled end for push-on connections.
- D. Remove scale and dirt on inside and outside before assembly.
- E. Prepare pipe connections to equipment with flanges or unions.

#### 3.3 INSTALLATION

# A. Bedding:

- 1. Excavation:
  - a. As specified in Section 312333 Trenching and Backfilling.
  - b. Hand trim for accurate placement of pipe to elevations as indicated on Drawings.
- 2. Dewater excavations to maintain dry conditions and to preserve final grades at bottom of excavation.
- 3. Place bedding material at trench bottom, level fill materials in one continuous layer not exceeding 6 of compacted depth, and compact to 95 percent of maximum density.

## B. Piping:

- 1. Comply with AWWA C600 for Ductile Iron and AWWA C900 for PVC.
- 2. Handle and assemble pipe according to manufacturer instructions.
- 3. Steel Rods, Bolts, Lugs, and Brackets: Coat buried steel before backfilling.
- 4. Maintain 10 feet of horizontal separation between water main and sewer piping.
- 5. Ductile-Iron Piping and Fittings: Comply with AWWA C600.
- 6. Field Welding Materials: Comply with AWWA C206.
- 7. Flanged Joints: Do not use in underground installations except within structures.
- 8. Route pipe in straight line, and re-lay pipe that is out of alignment or grade.
- 9. Bearing:
  - a. Maintain bearing along entire length of pipe.
  - b. Excavate bell holes to permit proper joint installation.
  - c. Do not lay pipe in wet or frozen trench.
- 10. Prevent foreign material from entering pipe during placement.
- 11. Allow for expansion and contraction without stressing pipe or joints.
- 12. Close pipe openings with watertight plugs during Work stoppages.
- 13. Install access fittings to permit disinfection of water system performed under Section 330110.58 Disinfection of Water Utility Piping Systems.

#### C. Locating and Detection

1. Install Tracer Wire over crown of water main pipe, wrapped around pipe to secure at each joint. Make connection to existing tracer wire at either end of relocation work. Pull wire into valve boxes for access by locators.

- 2. Install Detection Tape over centerline of water main at approximate depth of 6" below finished grade.
- D. Valves and Hydrants: As specified in Section 331419 Valves and Hydrants for Water Utility Service.
- E. Tapping Sleeves and Valves: As indicated on Shop Drawings and according to manufacturer instructions.
- F. Thrust Restraints: As specified in Section 330509.33 Thrust Restraint for Utility Piping.
- G. Backfilling:
  - 1. Backfilling: Backfill around sides and to top of pipe as specified in Section 312333 Trenching and Backfilling.
- H. Disinfection of Potable Water Piping Systems: As specified in Section 330110.58 Disinfection of Water Utility Piping Systems.

#### **3.4** TOLERANCES

- A. Section 014000 Quality Requirements: Requirements for tolerances.
- B. Install pipe to indicated elevation within tolerance of 5/8 inch.

#### 3.5 FIELD QUALITY CONTROL

- A. Section 017000 Execution and Closeout Requirements: Requirements for testing, adjusting, and balancing.
- B. Testing:
  - 1. Pressure test piping system according to AWWA C600 and following:
    - a. Test Pressure: Not less than 200 psig or 50 psi in excess of maximum static pressure, whichever is greater.
    - b. Conduct hydrostatic test for a minimum of two hours.
    - c. Slowly fill section to be tested with water; expel air from piping at high points.
    - d. Install corporation cocks at high points.
    - e. Close air vents and corporation cocks after air is expelled.
    - f. Raise pressure to specified test pressure.
    - g. Observe joints, fittings, and valves under test.
    - h. Remove and renew cracked pipes, joints, fittings, and valves showing visible leakage, and retest.
    - i. Correct visible deficiencies and continue testing at same test pressure for additional two hours to determine leakage rate.
    - j. Maintain pressure within plus or minus 5 psi of test pressure.
    - k. Leakage is defined as quantity of water supplied to piping necessary to maintain test pressure during period of test.
    - 1. Compute maximum allowable leakage using following formula:

- 1)  $L = SD \times sqrt(P)/C$ .
- 2) L = testing allowance, gph.
- S = length of pipe tested, feet.
- 4) D = nominal diameter of pipe, inches.
- 5) P = average test pressure during hydrostatic test, psig.
- 6) C = 148,000.
- m. If pipe under test contains sections of various diameters, calculate allowable leakage from sum of computed leakage for each size.
- n. Leakage:
  - 1) If test of pipe indicates leakage greater than allowed, locate source of leakage, make corrections, and retest until leakage is within allowable limits.
  - 2) Correct visible leaks regardless of quantity of leakage.
- 2. Compaction Testing:
  - a. Comply with **ASTM D1557**.
  - b. If tests indicate Work does not meet specified requirements, remove Work, replace, and retest.

#### **END OF SECTION 331413**

# SECTION 331419 VALVES AND HYDRANTS FOR WATER UTILITY SERVICE

#### PART 1 - GENERAL

# 1.1 SUMMARY

#### A. Section Includes:

- 1. Fire hydrants.
- 2. Valves
- 3. Valve Boxes

## B. Related Requirements:

- 1. Section 033000 Cast-in-Place Concrete: Concrete for thrust restraints.
- 2. Section 330110 Disinfection of Water Utility Piping Systems: Requirements for flushing and disinfecting.
- 3. Section 330509 Thrust Restraint for Utility Piping: Thrust restraints as required by this Section.
- 4. Section 331413 Public Water Utility Distribution Piping: Pressure testing of valves and hydrants.

# 1.2 UNIT PRICE - MEASUREMENT AND PAYMENT

A. Section 012000 - Price and Payment Procedures: Contract Sum/Price modification procedures.

#### B. Valves:

- 1. Basis of Measurement: By each.
- 2. Basis of Payment: Includes excavation, valve, valve box, accessories, bedding, and backfill.

## C. Fire Hydrants:

- 1. Basis of Measurement: By each.
- 2. Basis of Payment: Includes excavation, hydrant, isolation valve and box, accessories, foundation bedding, and backfill.

# 1.3 REFERENCE STANDARDS

# A. American Water Works Association:

- 1. AWWA C500 Metal-Seated Gate Valves for Water Supply Service.
- 2. AWWA C502 Dry-Barrel Fire Hydrants.
- 3. AWWA C503 Wet-Barrel Fire Hydrants.
- 4. AWWA C509 Resilient-Seated Gate Valves for Water Supply Service.

- 5. AWWA C550 Protective Interior Coatings for Valves and Hydrants.
- B. National Fire Protection Association:
  - 1. NFPA 291 Recommended Practice for Fire Flow Testing and Marking of Hydrants.
- C. NSF International:
  - 1. NSF 61 Drinking Water System Components Health Effects.
  - 2. NSF 372 Drinking Water System Components Lead Content.

# 1.4 COORDINATION

- A. Section 013000 Administrative Requirements: Requirements for coordination.
- B. Coordinate Work of this Section with installation of water mains.

#### 1.5 PREINSTALLATION MEETINGS

- A. Section 013000 Administrative Requirements: Requirements for preinstallation meeting.
- B. Convene minimum one week prior to commencing Work of this Section.

#### 1.6 SUBMITTALS

- A. Section 013300 Submittal Procedures: Requirements for submittals.
- B. Product Data: Submit manufacturer information regarding component materials, fittings, assembly and parts diagram, and accessories.
- C. Manufacturer's Certificate: Certify that products meet or exceed specified requirements.
- D. Manufacturer Instructions: Submit detailed instructions on installation requirements, including storage and handling procedures.
- E. Source Quality-Control Submittals: Indicate results of factory tests and inspections.
- F. Field Quality-Control Submittals: Indicate results of Contractor-furnished tests and inspections.
- G. Qualifications Statements:
  - 1. Submit qualifications for manufacturer and installer.
  - 2. Submit manufacturer's approval of installer.

# 1.7 SUSTAINABLE DESIGN SUBMITTALS – (NOT USED)

A. Section 018113 - Sustainable Design Requirements: Requirements for sustainable design submittals.

#### B. Manufacturer's Certificate:

- 1. Certify that products meet or exceed specified sustainable design requirements.
- 2. Materials Resources Certificates:
  - a. Certify source and origin for salvaged and reused products.
  - b. Certify recycled material content for recycled content products.
  - c. Certify source for regional materials and distance from Project Site.

#### C. Product Cost Data:

- 1. Submit cost of products to verify compliance with Project sustainable design requirements.
- 2. Exclude cost of labor and equipment to install products.
- 3. Provide cost data for following products:
  - a. Salvaged, refurbished, and reused products.
  - b. Products with recycled material content.
  - c. Regional products.

#### 1.8 CLOSEOUT SUBMITTALS

- A. Section 017000 Execution and Closeout Requirements: Requirements for submittals.
- B. Project Record Documents: Record actual locations of valves and hydrants.

#### 1.9 MAINTENANCE MATERIAL SUBMITTALS

- A. Section 017000 Execution and Closeout Requirements: Requirements for maintenance materials.
- B. Tools: Furnish one tee wrench of required length to Owner.

# 1.10 QUALITY ASSURANCE

- A. Materials in Contact with Potable Water: Certified according to NSF 61 and NSF 372.
- B. Cast manufacturer's name, pressure rating, and year of fabrication into valve body.
- C. Perform Work according to Owner and current industry standards.
- D. Maintain one copy of each standard affecting Work of this Section on Site.

#### 1.11 QUALIFICATIONS

A. Manufacturer: Company specializing in manufacturing products specified in this Section with minimum three years' documented experience.

B. Installer: Company specializing in performing Work of this Section with minimum three years' documented experience and approved by manufacturer.

#### 1.12 DELIVERY, STORAGE, AND HANDLING

- A. Section 016000 Product Requirements: Requirements for transporting, handling, storing, and protecting products.
- B. Delivery:
  - 1. Seal valve and hydrant ends to prevent entry of foreign matter.
  - 2. Inspection: Accept materials on Site in manufacturer's original packaging and inspect for damage.
- C. Store materials according to manufacturer instructions.
- D. Protection:
  - 1. Protect materials from moisture and dust by storing in clean, dry location remote from construction operations areas.
  - 2. Provide additional protection according to manufacturer instructions.

#### **PART 2 - PRODUCTS**

#### 2.1 FIRE HYDRANTS

- A. Manufacturers:
  - 1. Hydrants shall be traffic model and shall be American DarlingB-84-B-5.
- B. Dry-Barrel, Breakaway Type:
  - 1. Comply with AWWA C502.
  - 2. Pressure: 200 psi working pressure.
  - 3. Body: Cast iron.
  - 4. Valve: Compression type.
  - 5. Burial Depth: Minimum of 4-1/2 feet.
  - 6. Provide extension section where necessary for vertical installation and in accordance with manufacturer recommendations.
  - 7. Inlet Connection Size: 6 inches.
  - 8. Valve Opening: Not less than 5-1/4 inches in diameter.
  - 9. End Connections: Mechanical joint.
  - 10. Operating Nut: National Standard
  - 11. Interior Coating: Comply with AWWA C550.
  - 12. Shall be able to face a minimum of eight directions.
  - 13. Hydrant shall be fully bronze with all working parts of bronze.
  - 14. Valve seat ring shall be bronze and screw into a bronze retainer.

- 15. All working parts shall be removable through the top of the hydrant without disturbing the barrel of the hydrant.
- 16. The operating threads shall be totally enclosed in an operating chamber separated from the hydrant barrel by a rubber O-ring stem seal and lubricated by a grease or an oil reservoir. A stop nut shall be positioned in the top operating mechanism of the hydrant so that the valve stem cannot contact the bottom of the shoe when the hydrant is fully open.

#### C. Hose Connections:

- 1. One pumper 4-1/2 inches, two hose nozzles 2-1/2 inches.
- 2. National Standard thread type.
- 3. Attach nozzle caps by separate non-kinking chains.

#### D. Finishes:

- 1. All outside surfaces of the barrel above grade shall be painted with enamel equal to Koppers Glamortex 501.
- 2. Color: Yellow.

#### 2.2 VALVES

- A. Manufacturers
  - 1. Gate Valves shall be manufactured by M&H or Mueller
- B. Performance and Design Criteria:
  - 1. Pressure Rating:
    - a. 12-inch Diameter and Smaller: 200 psig.
    - b. 14-inch Diameter and Larger: 150 psig.
  - 2. End Connections: Mechanical joint
  - 3. Furnish valves of diameters 16 inches and larger with bypass valves and gear operators.

331419 - 5

- 4. Coatings:
  - a. Comply with AWWA C550.
  - b. Application: Interior and exterior.

#### C. Double-Disc Gate Valves:

- 1. Description:
  - a. Comply with AWWA C500.
  - b. Materials:
    - 1) Body: Iron.
    - 2) Trim: Bronze.
  - c. Seat Type: Double disc; parallel.
  - d. Stem:

- 1) Type:Non-rising.
- 2) Seals: O-ring.
- e. Operation:
  - 1) Square operating nut.
  - 2) Opening Direction: Counterclockwise.
- D. Resilient-Wedge Gate Valves:
  - 1. Description:
    - a. Comply with AWWA C509.
    - b. Body: Ductile iron.
    - c. Seats: Resilient.
    - d. Stem:
      - 1) Type: Non-rising.
      - 2) Material: Bronze.
    - e. Operation:
      - 1) Square operating nut.
      - 2) Opening Direction: Counterclockwise.

#### 2.3 VALVE BOXES

- A. Description:
  - 1. 12-inch Diameter Valves and Smaller:
    - a. Material: Cast iron.
    - b. Type: Two piece; screw.
  - 2. Valves Larger than 12-inch Diameter:
    - a. Material: Cast iron.
    - b. Type: Three piece; screw.
    - c. Base: Round.
  - 3. Lid Inscription: WATER.

# 2.4 SUSTAINABILITY CHARACTERISTICS – (NOT USED)

A. Section 018113 - Sustainable Design Requirements: Requirements for sustainable design compliance.

051.18.2.22

B. Material and Resource Characteristics:

- 1. Recycled Content Materials: Furnish materials with maximum available recycled content.
- 2. Regional Materials: Furnish materials extracted, processed, and manufactured within 500 miles of Project Site.

#### 2.5 ACCESSORIES

- A. Thrust Restraints: As specified in Section 330509 Thrust Restraint for Utility Piping.
- B. Valve Box Aligner: High-strength plastic device designed to automatically center valve box base and to prevent it from shifting off center during backfilling.
- C. Fire Hydrant Drainage Gravel: As specified in Section 310516 Aggregates for Earthwork.

# 2.6 SOURCE QUALITY CONTROL

- A. Provide shop inspection and testing of completed assembly.
- B. Owner Inspection:
  - 1. Make completed valves and hydrants available for inspection at manufacturer's factory prior to packaging for shipment.
  - 2. Notify Owner at least seven days before inspection is allowed.

#### C. Owner Witnessing:

- 1. Allow witnessing of factory inspections and tests at manufacturer's test facility.
- 2. Notify Owner at least seven days before inspections and tests are scheduled.

#### D. Certificate of Compliance:

- 1. If manufacturer is approved by authorities having jurisdiction, submit certificate of compliance indicating Work performed at manufacturer's facility conforms to Contract Documents.
- 2. Specified shop tests are not required for Work performed by approved manufacturer.

#### **PART 3 - EXECUTION**

#### 3.1 EXAMINATION

A. Section 017000 - Execution and Closeout Requirements: Requirements for installation examination.

- B. Determine exact location and size of valves from Drawings.
- C. Identify required lines, levels, contours, and datum locations.

D. Verify that elevations of existing facilities prior to excavation and installation of valves and hydrants are as indicated on Drawings.

#### 3.2 PREPARATION

- A. Section 017000 Execution and Closeout Requirements: Requirements for installation preparation.
- B. Locate, identify, and protect from damage utilities to remain.
- C. Do not interrupt existing utilities without permission and without making arrangements to provide temporary utility services.
  - 1. Notify Engineer not less than 7 days in advance of proposed utility interruption.
  - 2. Do not proceed without written permission from Engineer.

#### 3.3 INSTALLATION

- A. Perform trench excavation, backfilling, and compaction as specified in Section 312333 Trenching and Backfilling.
- B. Install valves and hydrants in conjunction with pipe laying.
- C. Provide buried valves with valve boxes installed flush with finished grade.
- D. Provide support blocking and drainage gravel while installing fire hydrants; do not block drain hole.
- E. Orientation:
  - 1. Set valves and hydrants plumb.
  - 2. Set fire hydrants with pumper nozzle facing roadway.
  - 3. Set fire hydrants with centerline of pumper nozzle 18 inches above finished grade and with safety flange not more than 6 inches nor less than 2 inches above grade.
- F. After main-line pressure testing, flush fire hydrants and check for proper drainage.
- G. Installation Standards: Install Work according to Newton County Water and Sewerage Authority standards.
- H. Disinfection of Water Piping System: Flush and disinfect valves and hydrants with water mains as specified in Section 330110 Disinfection of Water Utility Piping Systems.

# 3.4 FIELD QUALITY CONTROL

- A. Section 014000 Quality Requirements: Requirements for inspecting and testing.
- B. Section 017000 Execution and Closeout Requirements: Requirements for testing, adjusting, and balancing.

<b>a</b>	
C.	Testing: Pressure test hydrants with water mains as specified in Section 331413 – Public Water Utility Distribution Piping.
	END OF SECTION 331419
ITY OF	BYRON